INFLUENCE OF TECHNOSTRESS AND TEACHER AUTONOMY ON BURNOUT AMONG UNAIDED SECONDARY SCHOOL TEACHERS

Thesis Submitted for the Degree of DOCTOR OF PHILOSOPHY IN EDUCATION

by

SHAFEEK P

Supervised by

Dr. HASSAN KOYA M.P

Associate Professor



FAROOK TRAINING COLLEGE RESEARCH CENTRE IN EDUCATION KOZHIKODE

2023

DECLARATION

I SHAFEEK P., do hereby declare that this thesis entitled as INFLUENCE OF TECHNOSTRESS AND TEACHER AUTONOMY ON BURNOUT AMONG UNAIDED SECONDARY SCHOOL TEACHERS is a genuine record of research work done by me under the supervision of Dr. HASSAN KOYA M.P., Associate Professor (Rtd.), Farook Training College, Research Centre in Education, University of Calicut, and that no part of the thesis has been presented earlier for the award of any Degree, Diploma and Associateship in any University.

SHAFFEK P

Place: Farook College Date: **27**/01/2023 Dr. HASSAN KOYA M.P Associate Professor (Rtd.) Farook Training College Research Centre in Education

Certificate

I Dr. HASSAN KOYA M.P., do hereby certify that this report "INFLUENCE OF TECHNOSTRESS AND TEACHER AUTONOMY ON BURNOUT AMONG UNAIDED SECONDARY SCHOOL TEACHERS" submitted for the degree of Doctor of Philosophy in Education of the Farook Training College, is a record of bonafide study and research carried out by Mr. SHAFEEK P., under my supervision and guidance and that no part thereof has been presented before for any other Degree, Diploma or Associateship in any other university.

Place: Farook College

Date: 27/01/2023



Dr. HASSAN KOYA M.P. (Supervising Teacher)

Associate Professor (Reta) Farock Training College

Acknowledgement

First and foremost, I would like to express my gratitude to Almighty God for his continued kindness and support throughout the life. Without his blessings, I would not have been able to finish my research on time.

J am grateful to my supervisor, **Dr. Hassan Koya M P**, Associate Professor (Rtd.), Farook Training College, for his guidance and support throughout my study. His knowledge, motivation, and expert supervision were instrumental in the successful completion of my research report. J appreciate his dedication and scholarly guidance throughout the tenure of my study.

J would like to express my gratitude to **Dr. T. Mohamed Saleem**, Principal, Farook Training College and **Dr. C. A. Jawahar**, Former Principal, Farook Training College, for providing all necessary facilities and encouragement to conduct for the study.

I am grateful to Dr. Umer Farooque TK, Assistant Professor and Research Co-ordinator, Farook Training College for all the support provided.

J also extend my sincere gratitude to all teaching staffs and office staffs of Farook Training College for their support and encouragement during the study.

J would like to thank the Librarian and other library staffs of Farook Training College Library and CHMK Library, Calicut University.

The investigator also thankful to Dr. Muneer V, Assistant Professor, Farook Training College, and Miss. Saritha A.S, Research Scholar, Farook Training College and other respected research scholars for their continuous support,

J extend my sincere gratitude to Head master and teachers of various unaided secondary schools for valuable assistance in arranging adequate facilities during the data collection and their co-operation.

Profoundly thanking Mr. Ramprakash, Infratec-Chenakkal, for alignment and binding works.

The investigator would also like to express his gratitude to other various experts, friends & relatives, and others who helped him to conduct this study and prepare this research report.

I thank my parents, in-laws, siblings, wife and kids for their love, prayers encouragement and support.

Farook Training College

SHAFEEK. P

CONTENTS

LIST OF TABLES

LIST OF FIGURES

LIST OF APPENDICES

Chapter	Title	Page No.
1	INTRODUCTION	1-22
2	REVIEW OF RELATED LITERATURE	23-80
3	METHODOLOGY	81-116
4	ANALYSIS AND INTERPRETATION	117-240
5	SUMMARY, FINDINGS AND SUGGESTIONS	241-278
6	RECOMMENDATIONS	279-290
	REFERENCES	291-303
	APPENDICES	

LIST OF TABLES

Table No.	Title	Page No.
1	Time Period of Digital Transition	24
2	Definitions of Technostress	26
3	Research Studies on Technostress	47
4	Recent Research Studies on Teacher Autonomy	66
5	Definitions of Burnout	68
6	Details of Negative and Positive Items in Technostress Scale	92
7	Data and Results of Item Analysis for Technostress Scale	94
8	Details of Negative and Positive Items in Teacher Autonomy Scale	100
9	Data and Results of Item Analysis for Teacher Autonomy Scale	102
10	Details of Negative and Positive Items in Teacher Burnout Inventory	106
11	Data and Result of Item Analysis for Teacher Burnout Inventory	108
12	List of Schools and Number of Students for Each Category	110
13	Final Break-up of the Sample	113
14	Statistical Constants for the Distribution of Scores of Technostress and its Components for Total Sample	117
15	Statistical Constants for the Distribution of Scores of Teacher Autonomy and its Components for Total Sample	120
16	Statistical Constants for the Distribution of Scores of Teacher Burnout and its Components for Total Sample	122
17	Data and Results of Percentile Calculation of Technostress and its Components for Total Sample	125
18	One Sample t test for the Variable Technostress and its Components	126
19	Data and Results of Percentile Calculation of Teacher Autonomy and its Components for Total Sample	127
20	One Sample t test for the Variable Teacher Autonomy and its Components	129
21	Data and Results of Percentile Calculation of Teacher Burnout and its Components for Total Sample	130

Table No.	Title	Page No.
22	One Sample t test for the Variable Teacher Burnout and its Components	131
23	Results of Mean Difference Analysis of Technostress and its Components	133
24	Results of Mean Difference Analysis of Teacher Autonomy and its Components	134
25	Results of Mean Difference Analysis of Teacher Burnout and its Components	135
26	Results of Mean Difference Analysis of Technostress and its Components	136
27	Results of Mean Difference Analysis of Teacher Autonomy and its Components	137
28	Results of Mean Difference Analysis of Teacher Burnout and its Components	138
29	Results of Mean Difference Analysis of Technostress and its Components	139
30	Results of Mean Difference Analysis of Teacher Autonomy and its Components	140
31	Results of Mean Difference Analysis of Teacher Burnout and its Components	141
32	Results of Mean Difference Analysis of Technostress and its Components	142
33	Results of Mean Difference Analysis of Teacher Autonomy and its Components	143
34	Results of Mean Difference Analysis of Teacher Burnout and its Components	144
35	Results of the One-way ANOVA of Technostress and its Components	145
36	Results of the One-way ANOVA of Teacher Autonomy and its Components	146
37	Results of the One-way ANOVA of Teacher Burnout and its Components	147
38	Results of the One-way ANOVA of Technostress and its Components	148

Table No.	Title	Page No.
39	Results of the One-way ANOVA of Teacher Autonomy and its Components	149
40	Summary of Post Hoc test for Teacher Autonomy and its Components viz., Teaching and assessment, Professional Development and Extracurricular Subjects by Experience	151
41	Results of the One-way ANOVA of Teacher Burnout and its Components	152
42	Summary of Post Hoc Test for Teacher Burnout and its Components viz., Emotional Exhaustion and Reduced Personal Accomplishment by Experience	153
43	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Unaided School Teachers for Total Sample	155
44	Summary of Post Hoc Test for Teacher Burnout for Total Sample by Technostress Group	156
45	Summary of Post Hoc test for Teacher Burnout for Total Sample by Teacher Autonomy Group	157
46	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Male Unaided School Teachers	160
47	Summary of Post Hoc Test for Teacher Burnout for the Male Subsample by Technostress Group	161
48	Summary of Post Hoc test for Teacher Burnout for Male Subsample by Teacher Autonomy Group	162
49	Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Female Unaided School Teachers	165
50	Summary of Post Hoc Test for Teacher Burnout for the Female Subsample by Technostress Group	166
51	Summary of Post Hoc Test for Teacher Burnout the Female Su sample by Teacher Autonomy Group	167
52	Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Rural Unaided School Teachers	170

Table No.	Title	Page No.
53	Summary of Post Hoc test for Teacher Burnout for the Rural Teacher Subsample by Technostress Group	170
54	Summary of Post Hoc Test for Teacher Burnout the Rural Teacher Subsample by Teacher Autonomy Group	172
55	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Urban Unaided School Teachers	174
56	Summary of Post Hoc Test for Teacher Burnout for the Urban Teacher Subsample by Technostress Group	175
57	Summary of Post Hoc Test for Teacher Burnout the Urban Teacher Subsample by Teacher Autonomy Group	176
58	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of CBSE Unaided School Teachers	179
59	Summary of Post Hoc Test for Teacher Burnout for the CBSE Subsample by Technostress Group	179
60	Summary of Post Hoc Test for Teacher Burnout the CBSE Subsample by Teacher Autonomy Group	181
61	Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of State Syllabus Unaided School Teachers	183
62	Summary of Post Hoc test for Teacher Burnout for the State Syllabus Subsample by Technostress Group	184
63	Summary of Post Hoc Test for Teacher Burnout the State Syllabus Subsample by Teacher Autonomy Group	185
64	Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of basic Qualification Unaided School Teachers	188
65	Summary of Post Hoc test for Teacher Burnout for the basic Qualification Teacher Subsample by Technostress Group	189
66	Summary of Post Hoc Test for Teacher Burnout the basic Qualification Teacher Subsample by Teacher Autonomy Group	190
67	Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Highly Qualified Unaided School Teachers	193

Table No.	Title	Page No.
68	Summary of Post Hoc Test for Teacher Burnout for the Highly Qualified Teacher Subsample by Technostress Group	194
69	Summary of Post Hoc Test for Teacher Burnout the Highly Qualified Teacher Subsample by Teacher Autonomy Group	195
70	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Language Unaided School Teachers	198
71	Summary of Post Hoc test for Teacher Burnout for the Language Teacher Subsample by Technostress Group	198
72	Summary of Post Hoc Test for Teacher Burnout the Language Teacher Subsample by Teacher Autonomy Group	200
73	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Social Science Unaided School Teachers	202
74	Summary of Post Hoc Test for Teacher Burnout for the Social Science Teacher Subsample by Technostress Group	203
75	Summary of Post Hoc Test for Teacher Burnout the Social Science Teacher Subsample by Teacher Autonomy Group	204
76	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Science Unaided School Teachers	207
77	Summary of Post Hoc Test for Teacher Burnout for the Science Teacher Subsample by Technostress Group	208
78	Summary of Post Hoc Test for Teacher Burnout the Science Teacher Subsample by Teacher Autonomy Group	209
79	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Mathematics Unaided School Teachers	212
80	Summary of Post Hoc Test for Teacher Burnout for the Mathematics Teacher Subsample by Technostress Group	212
81	Summary of Post Hoc Test for Teacher Burnout the Mathematics Teacher Subsample by Teacher Autonomy Group	214

Table No.	Title	Page No.
82	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Novice Unaided School Teachers	216
83	Summary of Post Hoc test for Teacher Burnout for the Novice Teacher Subsample by Technostress Group	217
84	Summary of Post Hoc Test for Teacher Burnout the Novice Teacher Subsample by Teacher Autonomy Group	218
85	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Less Experienced Unaided School Teachers	221
86	Summary of Post Hoc Test for Teacher Burnout for the Less Experienced Teacher Subsample by Technostress Group	222
87	Summary of Post Hoc test for Teacher Burnout the Less Experienced Teacher Subsample by Teacher Autonomy Group	223
88	Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Experienced Unaided School Teachers	226
89	Summary of Post Hoc Test for Teacher Burnout for the Experienced Teacher Subsample by Technostress Group	226
90	Summary of Post Hoc Test for Teacher Burnout the Experienced Teacher Subsample by Teacher Autonomy Group	228
91	Pearson's Product Moment Coefficient of Correlation for Technostress and its Components and Teacher Burnout	230
92	Pearson's Product Moment Coefficient of Correlation for Teacher Autonomy and its Components and Teacher Burnout	231
93	R, R2 and Adjusted R2 for Model	232
94	Statistical Significance of the Overall Model	233
95	Details of Regression Coefficients and Individual Contribution	233
96	R, R2 and Adjusted R2 for Model 1	235
97	Values of R, R2 and Adjusted R2 of Model 2	236
98	Statistical Significance of the Overall Model	236
99	Details of Regression Coefficients and Individual Contribution	237

Figure No.	Title	Page No.
1	Technology Acceptance Model	38
2	The Autonomy Triangle	62
3	Transactional Model of Burnout	71
4	Variables Selected for the Study	82
5	Diagrammatic Representation of Variables	82
6	Summary of the Methodology	116
7	Histogram with Normal Plot of Variable Technostress for Total Sample	119
8	P-P Plot of Variable Technostress for Total Sample	119
9	Histogram with Normal Plot of Variable Teacher Autonomy for Total Sample	121
10	P-P plot of variable Teacher Autonomy for Total Sample	122
11	Histogram with Normal Plot of Variable Teacher Burnout for Total Sample	123
12	P-P Plot of Variable Teacher Burnout for Total Sample	123
13	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Total Sample	159
14	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Male Subsample	164
15	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Female Subsample	169
16	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Rural Teacher Subsample	173
17	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Urban Teacher Subsample	178
18	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for CBSE Subsample	182
19	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for State Syllabus Subsample	187

LIST OF FIGURES

Figure No.	Title	Page No.
20	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for basic Qualified Teacher Subsample	192
21	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Highly Qualified Teacher Subsample	197
22	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Language Teacher Subsample	201
23	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Social Science Teacher Subsample	206
24	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Science Teacher Subsample	211
25	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Mathematics Teacher Subsample	215
26	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Novice Teacher Subsample	220
27	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Less Experienced Teacher Subsample	225
28	Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Experienced Teacher Subsample	229

Appendix No.	Title
Ι	Technostress Scale - Malayalam (Draft)
Π	Technostress Scale - Malayalam (Final)
III	Technostress Scale - English (Final)
IV	Teacher Autonomy Scale - Malayalam (Draft)
V	Teacher Autonomy Scale - Malayalam (Final)
VI	Teacher Autonomy Scale - English (Final)
VII	Teacher Burnout Inventory - Malayalam
VIII	Teacher Burnout Inventory - English

LIST OF APPENDICES

Chapter 1

INTRODUCTION

- \Leftrightarrow Introduction
- ✤ Need and Significance
- ✤ Statement of the Problem
- ✤ Definition of key terms
- ✤ Objectives
- ✤ Hypotheses
- ✤ Variables

A

- Methodology
- ✤ Scope and Limitation
- ✤ Organization of the report

"Teachers are the backbone of any country, the pillar upon which all aspirations are converted into realities"

Dr. A P J Abdul Kalam

All education spectrums have undergone drastic changes irrespective of role of the teacher, place of the pupil, subject matter and methods of instruction. No matter how many educational system changes, the role of the teacher is paramount. The important and crucial role of the teacher in the learning process is generally recognized in all pedagogical sciences. There is a prevalent perception that teachers play a more important role in educational advancement than changing the physical facilities of the school or curriculum development. Teaching is such a profession that includes interaction with the burnt mind of the young, which is the most power full resource above the earth, beneath the earth and on the earth. So, it is essential for the healthy mind of the future generation that their teachers should be psychologically, physically, emotionally and socially healthy people. But in the student-centred system there was a situation where teachers' problems knowingly or unknowingly were ignored. In a way, it is the negligence of the pupil's issue itself.

There is a notion that teachers are not getting due consideration in the new education system. Above that, unaided teachers are the most marginalised and overlooked among teachers. There is found less serious discussion on the various issues of unaided teachers. As per the report of Unified District Information on School Education (UDISE +) 2017-18 the numbers of unaided teachers in Kerala is 71202. This is almost the same as the number of government teachers in Kerala. In Kerala, 2.33% of teachers are working in aided sectors. Only 26.26 % teachers work in government sector and 26.20 % work in unaided sectors. So, the issues of unaided teachers are a vital matter of our education system.

Kumar (2020) invited the attention of stakeholders and public to the cruel reality that unaided teachers are working like slaves. The problems of private

education sector and the teachers who work there are concealed in the enthusiasm shown by the government and the intellectual community in highlighting the public education sector in the mainstream media. There has been less discussion so far about the problems faced by teachers in the unaided sector. More than seventy percent of teachers in the unaided sector are women. Therefore, the problems in this area are also becoming the problems of women.

Peters (2001) indicates responsibilisation as one of the most powerful strategies of neoliberal economy. Unaided educational institutions put forward a very effective use of responsibilisation. Its nature is to warn teachers of self-assessment and self –warning through the constant conduct of examination and assessments among students. There are number of schools where teachers are not allowed to sit in classrooms or use toilets except at pre-arranged times and being fired as a matter of convenience to management.

Pithers and Sodden (1998) emphasised that role of overload is one of the main reason of teachers' stress. In the case of unaided sectors these issues of overload became a great mistreatment. The Workload of unaided teachers has increased by leaps and bounds and has turned into a huge exploitation. In addition to being overworked before and after the school hours, they are often twice as hard working compared to teachers in the public sector. Most unaided schools have strange rules that teacher can only teach in the classroom not outside the classroom. Classes in many places do not have chairs. There are some reports of incidents where the principal saw the teacher sitting in the chair in the classroom during the regular observations and severely abused in front of the students. There are lot of such issues that lead teachers to burnout. Here the investigator chooses two causes among them to check how it influences the teachers' burnout, that is technostress and teacher autonomy.

Time has witnessed that technology is causing great changes in all aspects of human life, both personal and social. The development of technology in an unpredictable way is advancing day by day and it is having a huge impact on all interventions. Radical changes in society are taking place in a way that man could not have even imagined a few years ago. Due to the development of technology, the world today is in the palms of everyone. They can know what is happening in any corner of the world in real- time and interact with people in any part of the world. In other words, it can be said that the equality of interventions beyond nationality and status is made possible by new technology. The use of information technology enhances the effectiveness of learning and develops the motivation for learning, which makes the learning process more successful. Information technology not only opens opportunities for the variability of educational activities, their personalization and, diversity but also allows for a new way of organizing the communication of all subjects of learning and building an educational system. The all-education system, its pedagogy, content, transaction, and method of teaching transformed into technology- based one.

Man began to experience new problems due to the proliferation of technology. His career, family life, social life and work life were badly affected. The term technostress began to be used to describe the negative impact of technology on the emotional, and psychological levels. The 21st century is a century of advanced technology. The modern student lives in the world of electronic culture. The role of the teacher in information culture is changing. He should be the coordinator of the flow of information. A teacher who adapts to the times should be ready to use information technology scientifically and effectively. Otherwise, the unscientific use of technology in any profession will lead to severe problems like technostress.

When incorporating technology into education, teachers may encounter several challenges such as lack of training, limited access to technology, difficulty adapting lesson plans, technical difficulties, lack of support, and limited time. These challenges can range from teachers not being familiar with the technology to schools

in low-income areas not having the resources to provide access, to teachers having to spend time adapting their lesson plans and dealing with technical difficulties which can take away from learning time. Additionally, some teachers may not have enough support from their administration, IT department, or other teachers, which can make it difficult to integrate technology effectively into their instruction, and with an increase in technology usage, teachers may have less time to focus on other aspects of teaching and learning.

As a result of this revolution, now teachers began to feel technostress. It may be due to the complexity and uncertainty of technology or the invasion of technology or the lack of technological facility. Technostress which has not been properly addressed may lead to the severe issue of burnout. All are concentrating on the advantages of technological advancements, but it should be studied and noted that the dark side of the technological invasion into the educational system. The word technostress was first used by Craig Brod in 1984. He explained it as "a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner". Technostress is the undesirable psychological link between individuals and the introduction of novel technologies. "Technostress is a process that includes the presence of 'technology environmental conditions; which are appraised as demands or 'technostressors' that are taxing on the individual and require a change; which set into motion coping responses; that lead to psychological, physical, and behavioural outcomes for the individual" (Tarafdar et al., 2019).

Although Brod (1984) has explained technostress as a disease, other scholars like Davis Millis (1998) described it as an incapability to adjust to changes, brought in by technology in his essay on 'Techno-stress and the organization: A Davis Millis Manager's Guide to Survival in the Information Age considers technostress', "as a condition whereby a person has to adapt to new technology especially when there is the inadequacy of the equipment, support, or the technology itself". After few years Weil and Rosen (1997) altered this explanation of Broad since they didn't accept that technostress as a disease. They elaborated the concept of technostress and defined it as "any negative impact on attitudes, thoughts, behaviour or psychology caused directly or indirectly by technology".

Though technostress has been given different labels such as computer anxiety and negative computer attitudes (Wang et al., 2008), it is characterized by the tenseness and anxiety that an individual feel when using technology. This negative emotion should be considered important because it tends to prevent one from further using the technology" (Joo, et al., 2016). There are other terms which have the same meaning with "technostress" used by other scholars contain technophobia, cyber phobia, computer phobia, computer anxiety, and computer stress. Further, the term digital depression has also been employed to recognise the mood of a worker when being dazed by technology (Chua et al 1999; Durndell & Haag, 2002; Mustaffa et al., 2007).

Another issue considered here as a reason for burnout of teacher is teacher autonomy. The very word Autonomy indicates the basic human requirement and capacity to perform independently. Exclusively in educational field teachers need right autonomy to act for the betterment of the teaching -learning process. NCERT (2014), mentioned, "The origin of Autonomy has been one of the central concerns of philosophers since ancient times. The concept first came into prominence in ancient Greece and was derived from the Greek words 'auto' (self) and 'nomos' (rule or law), meaning one who gives oneself their own laws". Immanuel Kant was the one of the chief exponents who expresses his views on the notion of autonomy which is based on his 'Moral theory' has great philosophical importance. To attaining selfsufficiency one can make himself free from external context that affects them. Autonomous individual has their own control and choice.

Autonomy strengthens every profession. Freedom is essential in all occupations that require skill and involve personal interventions. Especially in the teaching process which leads to the radical development of the students, autonomy has a great role. Raya et al. (2017) explain autonomy as "the competence to develop as a self-determined, socially responsible and critically aware participant in (and beyond) educational environments, within a vision of education as (inter)personal empowerment and social transformation".

The term burnout was first introduced in academic scenario by Freudenberger (1974), who defined it as "to fail, to wear out, or become exhausted by making excess demands on energy, strength or resources". Matheny, et al. (2000) noted that earlier research in to the phenomenon described burnout as a loss of idealism and enthusiasm for work. Burnout, as a form of work-related strain, is the result of a significant accumulation of work-related stress. Maslach (1982) defined burnout as "a syndrome of emotional exhaustion, depersonalization, and reduced personal accomplishment that can occur among individuals who do people work of some kind".

According to the well-known definition of burnout given by Maslach and Jackson, (1981), burned out people suffer from emotional exhaustion, depersonalization and a reduced sense of personal accomplishment. Emotional exhaustion denotes to feelings of being emotionally strained and having depleted one's emotional resources. Depersonalization refers to an undesirable, heartless and isolated attitude towards the people one works with, i.e. patients, clients or students. Reduced personal accomplishment refers to someone's negative self-evaluation in relation to their job performance (Schaufeli, et al., 2009).

The influence of technostress and teacher autonomy on burnout among unaided secondary school teachers is a topic of growing concern as technology continues to play a larger role in the field of education. Technostress refers to the negative psychological and physiological effects of prolonged exposure to technology, such as feelings of anxiety, frustration, and exhaustion. Teacher autonomy, on the other hand, refers to the degree of control and decision-making authority teachers have over their work, including use of technology in the classroom. Research suggests that unaided secondary school teachers who are exposed to high levels of technostress may be at a greater risk of burnout, a state of chronic emotional, physical, and mental exhaustion caused by prolonged stress. This is because they may feel overwhelmed and ill-equipped to deal with the constant demands of technology integration and the lack of autonomy in decision- making. Furthermore, a lack of autonomy in the decision-making process can also lead to a feeling of powerlessness and an increased risk of burnout among teachers, who may feel that their professional autonomy is being threatened. Overall, the influence of technostress and teacher autonomy on burnout among unaided secondary school teachers is an important area of research, as it can help to identify the technostress and autonomy, and how it contribute to burnout among teachers and inform interventions to support their well-being.

Need and Significance of the Study

During the student-centred education, we unintentionally ignored the concerns of teachers. Here the investigator tries to identify two major problems of teachers at present and how it become the giant reason for the burnout of teachers. In this techno pedagogical realm teachers suffering lot of glitches. Here the investigator selected two chief problems that is technostress and teacher autonomy. Technostress of teachers is an unexplored area. Technostress is defined as a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner. In all means, a teacher should get real and positive autonomy which makes him perfect in the devotion of guiding generation. Huang (2005) defined teacher autonomy as "teachers' willingness, capacity and freedom to take control of

their own teaching and learning". These two issues become the main cause for burnout of teachers. Freudenberg (1980) identified burnout as a "state of fatigue or frustration brought about by devotion to a cause, way of life, or relationship that failed to produce the expected reward".

In this postmodern world man start to realize the negative side of modern revolution in all areas of development. Like almost all other things under the sky, information technological revolution too has its own positive and negative impact. It helps a lot to contribute great changes in the education filed. Simultaneously, it brings some negative impacts on education system, students and teachers. Technostress is one of the main issue which affects both pupil and teacher. Teachers' technostress a relevant topic due to the intensity of the integration of new technologies into teaching. Furthermore, technostress seems to influence teachers' intentions to use technology (Joo et al., 2016), in general, high levels of technostress are found to be associated with lower job satisfaction and job performance, and even with intentions to quit (Pullins et al., 2020). "Indications about the relevance of better understanding technostress abound and include observations such as Facebook fatigue or employee burnout and studies suggesting that one in ten employee sick days are probably technostress related" (Maier, 2014).

Studies reveals that "technostress symptoms may lead to a poorer quality of social interaction between the teacher and pupils, and consequently to poorer learning. Moreover, there are only few studies focusing on teachers' technostress" (Al-Fudail & Mellar, 2008; Yin et al., 2012; Joo et al., 2016). It lights on the relevance of the study of technostress among teachers.

Latest researches stated that "educators using novel technology in classrooms have felt technostress, which instigated opposing effects in the implementation of technology. For instance, Al-Fudail and Mellar (2008) observed that there exists numerous elements triggering technostress such as technological system failure, inadequate technical and social backing for technology practice, increased time for setting up and preparing class lectures, and insufficient institutional culture for technology implementation. In a similar manner, Joo et al., (2016) stated that teachers' technostress levels increased after implementing digital textbooks in South Korea. He appealed that teachers felt disappointed when technological system fiascos happened, because they felt like they were incapable to control the condition. Also, the study stated that direction for teachers on how to incorporate digital textbooks into their instruction was imperfect, increasing teachers' psychological and physical overload for technology implementation." (Joo et al., 2016).

Findings of the studies highlight the relationship between work autonomy and technostress. Autonomy influence technostress more than demographics (Ragu-Nathan et al., 2008; Tarafdar et al., 2011). Teacher autonomy has elevated as one of those attractive modern terms connected with educational excellence, innovation and devolution of schools across diverse nations. Yet, for a diversity of causes and in spite of its pervasive use, the notion of teacher autonomy and its implications for education and school patrons continue dense.

Now days "Teaching is far too complex to be reduced to a simple recipe (first do this, then do that') Or to be assigned a lock–step calendar ('if it's Tuesday, we must be on page 47'). Teachers need autonomy to respond to the highly individualised dynamics of the classroom, to re-teach using different strategies when student's struggle, and to divert from the lesson plan during those magic moments when student interest takes an idea in a new direction. They also should have the autonomy to pursues some topics and areas of study that are of particular interest to them" (Dufour, 1999).

Positive autonomy helps to teachers act freely and creatively. "They give more importance to update their knowledge and never use the same lecture notes in the class year after which were prepared by them or other teachers. They never spoil the students by spoon-feeding them. They never get satisfied with what they have learnt but find

delight in enhancing their knowledge and techniques of fostering learner autonomy every day. They share what they have learnt with their colleagues. To summarize, autonomous teachers work with learners and not for learners." (Raya & Vieiria, 2015).

Researchers highlighted the work autonomy as a main aspect of a person's sensation of whether their profession is wearing or satisfying (Wharton, 1993; Moller et al., 2006). Rayan and Deci (2017) indicated the advantages of autonomy -supportive leadership in encouraging employee empowerment and self-initiation. There is a report as named A Nation at Risk, which is published in 1983 by United States National commission on excellence in education and it was the one of the first exertions that initiated school reform movement. Based on this report there are explosion of rules and regulations which published to empower the efforts to improve the education system. As part of school reform movement, there arise the need of competent individuals in the field of teaching. To achieve the quality numbers in the field should be upsurge the feeling of professionalism in the field. This could only be accomplished if autonomous of teachers was accepted and executed (Rudolph, 2006). If teachers are to be professionally empowered and uplifted, teachers must have the autonomy to recommend the best dealing for their pupils as doctors and lawyers do for their clients.

Recent studies have depicted teacher autonomy in a very positive way as, for instance, "it is positively correlated with job satisfaction, empowerment, engagement and professionalism and is negatively correlated to emotional exhaustion" (Erss et al., 2016; Wermke et al., 2019; Skaalvik & Skaalvik, 2014; Wilches, 2007). Job autonomy is considered a major factor in an individual's assertion of whether their job is exhausting or satisfying (Wharton, 1993; Moller et al., 2006), and recent research has demonstrated the benefits of autonomy-supportive leadership in promoting worker empowerment and self-initiation (Ryan & Deci, 2017)

In examining factors that may lessen teacher stress and burnout, Collie (2021) collected data between March and May 2020 from 325 Australian teachers and found that autonomy-supportive leadership was associated with greater buoyancy and, in

turn, lower somatic burden, stress related to change, and emotional exhaustion. In contrast, autonomy-thwarting or controlling practices involve pressuring individuals to feel, act, and think in particular ways, and have been positively associated with emotional exhaustion (Ryan & Deci, 2017). Research on P-12 teachers has consistently identified positive outcomes associated with increased teacher autonomy.

A study of 251 South African teachers found increased job autonomy to be related to teachers' increased feelings of meaningfulness and engagement in their work (Peral & Geldenhuys, 2016). Similarly, teacher autonomy may be a viable means of increasing their well-being (Collie et al., 2018), as previous studies have reported decreased feelings of autonomy to be a common experience in the teaching profession (Greenville-Cleave & Boniwell, 2012).

Frostenson (2015) observed that autonomy is deteriorating due to new educational changes, resulting in the deprofessionalisation of teachers and concluded that autonomy is declining due to recent educational changes, resulting in the deprofessionalisation of teachers. While technology has made easy various everyday jobs, it also has affected people's well-being. Really it is a double edged sword. Subramanyam et al. (2000) claimed that technology have displaced activities that can affect individual's physical wellbeing. The authors underline those menaces such as overweightness, seizures, and hand wounds have been connected to the use of technology. As technology has used in a great extent, it is essential to do more studies on their influence on society in general.

Burnout can evident in several professions however is especially extensive among teachers (Iancu et al., 2018). In fact, teaching is regarded one of the most stressful job (Kyriacou, 2001). It maybe predictable that there are lot of demands and stressors that teachers experience on day–to day basis (Mccarthy, et al., 2016), consisting high workload, frequent performance evaluation and student misbehaviour (Kyriacou, 2001). All these elements together provide ample chances for teachers to

be at menace of burnout (Schaufeli & Buunk, 2003). Gorden (2007) indicated that when stress is increasing the employees felt they are burnout.

While time passes the factors that affects the burnout of teachers also changes. The revolution of information technology horribly changed all walks of life in despite of professional and personal life. The need for a study on the influence of technostress and teacher autonomy on burnout among unaided secondary school teachers is important for several reasons. The study will aid in understanding how technology integration in education impacts the mental and emotional well-being of teachers, as technology continues to play a larger role in education, it is necessary to comprehend how it affects teachers and develop strategies to support them. Furthermore, the study will help in realizing the relationship between technostress, teacher autonomy, and burnout can help educators and policymakers develop targeted interventions to support teachers' mental and emotional well-being. Additionally, the study will provide insights into the experience of unaided secondary school teachers which is an important aspect of the education system in Kerala. Comprehending the challenges faced by these teachers and ways to mitigate them can help to improve the quality of education in the country. Lastly, the study will create awareness among teachers, policymakers, and the public about the importance of addressing the well-being of teachers in the context of technology integration in education. In conclusion, researching the influence of technostress and teacher autonomy on burnout among unaided secondary school teachers is vital as it can help to improve the well-being of teachers and the quality of education in the state. Through the literature the researcher learned about teacher autonomy and technostress but hasn't come across much research in attempting to relate technostress and teacher autonomy to teacher burnout. There for the researcher has decided to attempt to conduct a study on influence of technostress and teacher autonomy on teacher burnout.

Statement of the Problem

As technology becomes increasingly integrated into the classroom and workplace, teachers are facing new and unique challenges in the form of technostress. At the same time, many teachers feel that they lack autonomy in their work, which can lead to feelings of dissatisfaction and powerlessness. The combination of these factors can lead to burnout, which can negatively impact the well-being and effectiveness of teachers. Thus the problem for the present study is entitled "INFLUENCE OF TECHNOSTRESS AND TEACHER AUTONOMY ON BURNOUT AMONG UNAIDED SECONDARY SCHOOL TEACHERS".

Definition of Key Terms

The various term used in the title have been operationally defined below

Influence

According to Merriam Webster dictionary 'influence is the power or capacity of causing an effect in indirect or intangible ways.

In the present study, influence stands for the influence of the independent variable on the dependent variable.

Technostress

"Technostress is defined as a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner" (Brod, 1984).

In the present study, Technostress is the negative psychological link between unaided secondary school teachers and the introduction of new technologies.

Teacher Autonomy

"Teacher autonomy is not independence or isolation rather it involves teachers' independence, responsibility, mutual support, professional discretion, and commitment to the educational community" (Wilches, 2007).

In this present study, Teacher autonomy is the freedom of teachers to take control of all activities related to their teaching. It includes freedom in the areas of establishing school identity and praxis, Teaching and assessment, Parental involvement, professional development, extracurricular subjects, and Curriculum development

Burnout

Burnout defined as a "state of fatigue or frustration brought about by devotion to a cause, way of life, or relationship that failed to produce the expected reward" (Freudenberger, 1980).

In this present study, teacher burnout is a psychological syndrome of emotional exhaustion, depersonalization, and reduced personal accomplishment which occur among unaided secondary school teachers.

Objectives of the Study

- 1. To find out the extent of technostress, teacher autonomy and burnout among unaided secondary school teachers of Kerala.
- To find out whether there exists any significant difference in technostress, teacher autonomy and burnout among unaided secondary school teachers based on relevant subgroups viz. gender, locale, type of school, educational qualification, discipline of teaching and teaching experience.
- To find out the main effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant subgroups.
- To find out the first order interaction effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant subgroups.

- To find out the individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.
- 6. To develop regression equation to predict teacher burnout from technostress and teacher autonomy.

Hypotheses of the Study

- There exists significant gender difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- There exists significant locale difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- There exists significant school difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- 4. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on educational qualification.
- 5. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers-based on subject of teaching.
- 6. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on teaching experience.

- 7. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for total sample.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school male teachers.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school female teachers.
- 10. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers in rural area.
- 11. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers in urban area.
- 12. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for CBSE unaided secondary school teachers.
- 13. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the

dependent variable Teacher burnout for state syllabus unaided secondary school teachers.

- 14. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers with basic qualification.
- 15. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers with higher qualification.
- 16. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school language teachers.
- 17. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school social science teachers.
- 18. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school science teachers.
- 19. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the

dependent variable Teacher burnout for unaided secondary school mathematics teachers.

- 20. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school novice teachers.
- 21. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school less experience teachers.
- 22. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school experienced teachers.
- 23. There is significant individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.

Variables Selected for the Study

. Independent Variables

- Technostress
- Teacher Autonomy

Dependent Variable

• Teacher Burnout

Introduction 19

Methodology

The methods proposed for the present study is briefly described below:

Design of the Study

The investigator used survey method to study the influence of independent variables Technostress and Teacher Autonomy on dependent variable Teacher Burnout. The survey method comes under the preview of Descriptive study.

Population and Sample

The present study is conducted on a representative sample of 510 unaided secondary school teachers selected from Kerala. Due weightages were given to the relevant subgroups of the population such as gender, type of school, locale of the institution, educational qualification, subject of teaching and teaching experience. Stratified sampling technique is used for the present study.

Tools to be Used

The investigator proposed to construct a

- 1. Technostress Scale (Shafeek & Koya, 2019)
- 2. Teacher Autonomy Scale (Shafeek & Koya, 2019)
- 3. Teacher Burnout Inventory (Shafeek & Koya, 2019)

Statistical Techniques to be Used

The main statistical techniques employed for the present investigations are given below:

- Basic descriptive statistics.
- Percentile
- One sample t test

- Two tailed test of significance of difference between mean scores of large independent samples.
- One way ANOVA.
- Two way ANOVA with 3x3 factorial design.
- ANOVA followed by Scheffe's test of post hoc comparison
- Pearson's product moment coefficient of correlation.
- Multiple Regression Analysis

Limitations of the Study

Even though lot of endeavours have been made to make the study as accurate and generalizable as possible, some limitations were sneaked into the study as shown below:

- Many schools not give consent for conducting the survey. So only 510 responses from teachers were obtained due to this limitation.
- The sample of the study didn't include all level of teachers. The study was limited only to secondary level teachers due to the practical difficulties.

Scope and Delimitation

The main purpose of present investigation is to explore how technostress and teacher autonomy influence burnout of unaided secondary school teachers. For this study appropriate tools constructed and standardized by the investigator. With the help of suitable tools, the required data were collected from a stratified sampling of 510 unaided secondary school teachers of Kerala state to make the study more unbiased and precise. Analysis of the data was done with maximum care. Since the sample of the study comprises various teachers from different districts. The results

can be generalized. The influence of the study may provide valuable suggestions for teachers, educators and administrators.

Though possible precautions were taken to obtain reliable and generalizable results, the investigator would like to point out certain limitations likely to enter into the study.

The present study is delimited to the following aspects.

- Selection of the independent variables for the study is confined to two major variables viz., technostress and teacher autonomy.
- Selection of a few schools across the state intended for data collection may not be representative of all institutions in the state.
- 3. The generalizability of the study will be limited to the extent of the nature of the tools and sample selected.

Organization of the Research Report

The report of the study is organized in five chapters. The details incorporated in each chapter are as follows:

Chapter 1 includes a brief introduction of the problem, need and significance of the study, statement of the problem, definition of key terms, variables of the study, hypotheses and objectives methodology, scope and limitations of the study.

Chapter II presents a brief theoretical outline of variables and a summary of the review of related studies.

Chapter III includes the methodology of the study in detail. This chapter comprises description of method used, variables, tools used for the data collection, sample for the study, data collection procedure, and statistical techniques used for analysis.
Chapter IV deals with the statistical analysis of data and interpretation of the results.

Chapter V contains a brief view of summary of the study, major findings and conclusions of the study. It also presents detailed report on educational implications of the study and suggestions for further research.

Chapter VI presents the suggestions for further research and recommendations of the study in detail.

Report is followed by the references and a series of appendices pertaining to this study.

Chapter 2

REVIEW OF RELATED LITERATURE

- Theoretical overview of Technostress
 Studies related to Technostress
- Theoretical overview of Teacher Autonomy
- Studies related to Teacher Autonomy
- Theoretical overview of Teacher Burnout
- Studies related to Teacher Burnout

Literature review is a portrayal of the literature pertinent to a particular topic or area. It gives an outline of what has been studied, who are the main scholars in the field, what are the dominant theories, what ways and means are apposite and beneficial. It is indispensable for any research and the researcher should comprehend up-to date data about what has been studied in the area. It is a crucial and important part of any research effort since it evades replication of research. Overall, the review of related literature will provide a comprehensive overview of the current state of knowledge on the topic and will serve as a foundation for the current study. It will also help to identify any gaps in the literature that the current study aims to address and provide a context for the research questions and hypotheses. This chapter deals with the theoretical background and review of related literature of technostress, teacher autonomy and teacher burnout. The literature reviewed in the present study has been classified into the following heads

- Theoretical overview of Technostress
- Studies related to Technostress
- Theoretical overview of Teacher Autonomy
- Studies related to Teacher Autonomy
- Theoretical overview of Teacher Burnout
- Studies related to Teacher Burnout

Theoretical Overview of Technostress

"Technology is just a tool. In terms of getting the kids working together and motivating them, the teacher is the most important" (Bill Gates).

Time has witnessed that technology is instigating prodigious vicissitudes in all facets of human life, both personal and social. The development of technology in an unpredictable way is advancing day by day and it is having an enormous impact on all interventions. Deep-seated changes in society are taking place in a way that man could not have imagined a few years ago. Due to the advance of technology, the world today is in the hands of everyone. They can know what is happening in any nook and corner of the world in real time and communicate with people in any part of the world. In other words, equality of interventions beyond nationality and social status is made possible by

new technology. In the so-called "Information Age" the growing practice of technology has become the pouring force in the way people learn, work and play (Drake, 2000). Information technology is a double-edged sword, producing assistances and issues.

The time period of digital transition is given below

Table 1

Time Period of Digital Transition

Name of the Period	Duration
Age of Mass Communication	1950 - 1990
Personalize Automation	1990 – 1994
Age of Information	1994 – 1996
Instant Interaction	2000 – continue

Table is prepared based on the data given by Al Shami, 2008

Emergence of the Concept of Technostress

Man began to experience new problems due to the proliferation of technology. His career, family life, social life and professional life were badly affected. The term technostress began to be used to describe the negative impact of technology on the emotional and psychological level. The word technostress was first used by Brod in 1984. He explained it as "a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner". Technostress is the undesirable psychological link between individuals and the introduction of novel technologies. 'Technostress is a process that includes the presence of 'technology environmental conditions'; which are appraised as demands or 'techno stressors' that are taxing on the individual and require a change; which set into motion 'coping responses'; that lead to psychological, physical, and behavioural 'outcomes' for the individual" (Tarafdar et al., 2017).

Evolution of Definitions

The issue of technostress has been started to explore since 1980s. Brod (1982) one of the famous American psychotherapist defined as "as a situation resulting in difficulty adapting, stemming from the use of a new technology by an

individual or organization" later he defined it as "a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner". Jay (1981) explains risky forms technostress, i.e.; horrible fear of computers, and coined the term 'computer phobia'.

Although Brod (1984) has explained techno-stress as a disease, other scholars like Davis Millis described it as an incapability to adjust to changes, brought in by technology in his essay on 'Techno-stress and the 33 organization: A Davis Millis Manager's Guide to Survival in the Information Age considers techno-stress', "as a condition whereby a person has to adapt to new technology especially when there is the inadequacy of the equipment, support, or the technology itself". After few years Weil and Rosen (1997) altered this explanation of Broad since they didn't accept that technostress as a disease. They stated it as" any negative impact on attitudes, thoughts, behaviour or psychology caused directly or indirectly by technology".

Weil and Rosen (1997) elaborated the concept of technostress, and defined it as "any negative impact on attitudes, thoughts, behaviours or psychology caused directly or indirectly by technology." Clark and Kalin (1996) pointed out that technostress is a resistance to revolution. Technology should not be alone the key reason in technostress since technologies are mere apparatuses and the stress is a natural response. So, they have recommended that in order to control the technostress, it is the change of the user that has to be controlled not the technology. Champion (1988) expresses same view in his article 'Techno-stress: Technology's Toll'. The information era is all about different transformation like cultural and social, or to be more precise, a reaction to techno change, but not to adaptation of technical components such as gadgets, programs, web, or fibre optics. It has become an issue of adaptation due to individual's incapability to adjust with or to get used to new technologies.

While technostress has been given diverse terms such as computer anxiety and negative computer attitudes (Wang et al., 2008), it is characterized by the tenseness and nervousness that an individual sense when using technology. "This undesirable emotion should be considered vital since it inclines to stop one from further using the technology" (Joo et al., 2016). Famous definitions of technostress are presented below.

Table 2

Definer	Definition
Tarafdar et al. (2019)	"Technostress is the stress that individuals experience due to their use of Information Systems (IS)."
Salanova (2007)	"Techno-stress is anxiety, mental fatigue, scepticism and ineffectiveness resulting from the focusing on ICT use or its future use"
Bondanini et al. (2020)	"Technostress is the mental and psychological problems afflicting people with technologically dense work"
Weil & Rosen (1997)	"Technostress is any negative impact on attitudes, thoughts, behaviour or psychology caused directly or indirectly by technology".
Champion (1988)	"Stated that rapidly changing technology would have a negative effect on future living, and thereby described technostress as "The Price of Using Technology".
Brod (1984)	"Technostress is .a modern disease of adaptation caused by an inability to cope with the new computer technologies in a healthy manner. It manifests itself in two distinct ways: in the struggle to accept computer technology, and in the more specialized form of over identification with computer technology."
Caro & Sethi (1985)	"Technostress is a perceived, dynamic adaptive state between the person and the environment, mediated by socio psychological processes and influenced by the nature of the technological environment"
Brod (1982)	"Technostress is a situation resulting in difficulty adapting, stemming from the use of a new technology by an individual or organization."
Wang et al. (2008)	"Reflection of one's discomposure, fear, tenseness and anxiety when one is learning and using computer technology directly or indirectly that ultimately ends in psychological and emotional repulsion and prevents one from further learning or using computer technology."
Salanova et al. (2013)	"Negative psychological state associated with the use or threat of ICT use in the future. They found that a technostress experience can be related to feelings of anxiety, mental fatigue, skepticism and inefficacy"
Wes (2007)	"Technostress is made up Technostress is a made-up word that describes a particular sort of reaction to technological change and expectations."
Davis-Millis (2006)	"A condition resulting from having to adapt to the introduction and operation of new technology, particularly when equipment, support, or the technology itself is inadequate."
Kupersmith (2006)	" 'Technostress' (computer-related stress), a common problem for reference librarians in the 1990s, is a combination of performance anxiety, information overload, role conflicts, and organizational factors."

Definitions of Technostress

Techno stress is the incompetence to adapt to or cope with new computer technologies which unveils itself in one of two means that is computer handlers struggle to accept the technologies and computer users' over-identification with the technology (Brod, 1984). There are other terms which have the same meaning with "technostress" used by other scholars contain technophobia, cyber phobia, computer phobia, computer anxiety, and computer stress. Further, the term digital depression has also been employed to recognise the mood of a worker when being dazed by technology (Chua et al., 1999; Durndell & Haag, 2002; Mustaffa et al., 2007).

Technostress is a disorder affected by the inefficiency of "people and professional areas to adapt to rapid changes in technology. In the study of Brod emerged as a factor of technostress also the level and the user's technical expertise, the pressure applied from the outside during use, and the atmosphere lived within the workplace" (Sahin, 2009).

The definition of technostress has evolved over time as research and understanding of the phenomenon has progressed. Initially, technostress was defined as the negative psychological and physiological effects that result from excessive or inadequate use of technology. Later, definitions began to focus on the specific sources of stress, such as the constant pressure to stay connected, the overwhelming amount of information, and the pressure to keep up with new technology. Some researchers have defined technostress as the "emotional, cognitive, and physiological responses to being overwhelmed by technology-related demands." Others have described it as "the negative impact of technology on an individual's psychological and physiological well-being." More recently, definitions of technostress have shifted to include not only the negative effects of technology but also the positive effects, such as increased productivity and connectivity. The definition of technostress now encompasses a balance between the positive and negative effects of technology, which is in line with the idea that technology can be a double-edged sword.

In summary, the definition of technostress has evolved over time to reflect the growing understanding of the phenomenon, and currently it is understood as the emotional, cognitive, and physiological response to the balance of positive and negative impacts of technology-related demands.

Dimensions of Technostress

According to Taraffdar et al. (2007) there are five components for techno stress which are techno-overload, techno-invasion, Techno insecurity, techno complexity and techno uncertainty.

Techno Insecurity

It is the one of the serious issues caused by technological devices. It is related with circumstances where individuals feel endangered about losing their data especially personal and official secrets. Techno-insecurity arises when a user minds nervous about losing his or her profession to other co-workers who have better knowledge of and skills with the technology compared to him or her. Technoinsecurity has been linked to situations where users face threats about technology due to automation from ICTs or other persons who have a improved understanding of ICTs.

Techno insecurity refers to the feeling of insecurity or anxiety that individuals may experience when using technology, particularly in the context of online security and privacy. It encompasses concerns about the safety and security of personal information, the potential for hackers to access sensitive information, and the fear of identity theft.

This insecurity can manifest in a number of ways, such as reluctance to share personal information online, avoidance of certain technology or social media platforms, or a lack of trust in the security of online transactions. It can also take the form of anxiety or stress when using technology, especially when dealing with sensitive information or conducting financial transactions online. Techno insecurity can be a significant barrier to the adoption and use of technology, particularly for those who may not be as familiar with technology. It can also lead to reduced productivity and increased stress for individuals and organizations.

Techno Complexity

Techno complexity refers to the difficulty or complexity that individuals may experience when using technology. It encompasses the feeling of being overwhelmed by the amount of information, the number of options, or the complexity of the technology. This complexity can make it difficult for people to understand how to use technology effectively, which can lead to frustration, decreased productivity, and a reluctance to adopt new technology. It is the conditions where the complexity connected with ICTs can lead users to experience inadequacies with their computer skills and forces them to employ time and energy in learning and understanding ICTs. It indicates the complex technological arena including new updates in the field of technology which are force to teachers or any workers to use lot of time and energy in learning and acquiring the knowledge how to practice new applications and software. Besides, due to complication of the technology, teachers must apply more time to learn what way to practice technology in education process. Stress happens when they sense that the diversity of applications and functioning is threatening and they do not comprehend the necessity of practicing it. Additionally, persistent variations of technology being a reason to make users as strained and discontented with the whole system (Ibrahim & Yusoff, 2015).

Some of the sources of techno complexity include:

- The rapid pace of technological change, making it difficult for individuals to keep up with new developments and features.
- The large number of options and features available, making it difficult for individuals to determine which one best suits their needs.
- The complexity of the technology itself, with multiple layers of menus and options that can be difficult to navigate.

• The lack of clear instructions or explanations, which can make it difficult for individuals to understand how to use a particular technology.

Techno Invasion

It denotes a condition that users feel that they are never free from technology. They belief that technology has occupied their all interactions. (Ibrahim & Yusoff, 2015). Techno invasion refers to the extent to which technology intrudes into an individual's personal and professional life. It encompasses the feeling of being constantly connected and always available, as well as the pressure to respond to notifications and messages immediately, regardless of time or location. This constant connection can make it difficult for people to disconnect and engage in activities that do not involve technology, leading to feelings of stress, anxiety, and burnout. Techno invasion can have a negative impact on an individual's physical and mental well-being, as well as on personal and professional relationships. It can lead to decreased productivity, burnout, and a lack of work-life balance. Some of the sources of techno invasion include: The constant availability of technology, such as smartphones and laptops, which make it difficult for individuals to disconnect and unplug. Second one is the pressure to be always available, especially in the context of work, where employees may feel obligated to respond to emails and messages outside of normal working hours. Third one is the constant bombardment of notifications and messages, which can make it difficult for individuals to focus on other tasks or activities.

Techno Overload

It happens when users are incapable to recognise what is truly beneficial data. They use more time and energy in information handling because they transfer more information than is essential and obtain more data than they can efficiently process. Users also receive more information than they can process and use meritoriously (Ibrahim & Yusoff, 2015). Techno-overload describes situations

where ICTs force users to work faster and longer. Techno overload denotes a condition where teachers are forced to work quicker as a result of the use of connected ICTs each and every time, eventually they are compelled to be the part of increased workload. As a consequence of this overload teachers suffering work exhaustion and other health glitches. (Christian et al., 2020)

Techno overload refers to the feeling of being overwhelmed by the sheer volume of information and technology that is available. It encompasses the difficulty of handling, processing and making sense of the vast amount of information, options, and features that are available. This overload can make it difficult for people to focus, retain information, and make decisions, leading to feelings of stress, anxiety, and frustration. Techno overload can have a negative impact on an individual's productivity, mental well-being and decision-making ability. It can also lead to decreased job satisfaction and a lack of engagement.

Techno Uncertainty

Technological uncertainty refers to the lack of knowledge or understanding about the potential impacts and risks of new or emerging technologies. It can include uncertainty about the safety, efficacy, or long-term effects of a technology, as well as uncertainty about its economic, social, or political implications. Technological uncertainty can create challenges for decision-makers, businesses, and individuals, as they may have difficulty assessing the risks and benefits of a technology and determining how to best use or regulate it. It can also lead to public uncertainty about the technology and mistrust of the entities that develop or promote it.

Techno Overuse

Technological overuse refers to the excessive use of technology, such as smartphones, computers, and the internet, to the point where it negatively impacts a person's physical, mental, and social well-being. This can manifest in symptoms such as insomnia, eye strain, anxiety, depression, and social isolation. It can also

lead to a lack of physical activity, poor posture, and other health problems. Longterm overuse can also lead to addiction, making it difficult for a person to cut back on their technology use.

Types of Technostress (Fisher, 1996)

According to Fisher, there are two types of technostress which are the result of heavy and long term of use of new technologies.

Physical Forms of Technostress

Heavy use of new technologies may result into physical strain such as eyestrain, headaches, and backaches. Mainly two sub types physical forms are there.

- a) Repetitive Strain Injuries. Carpal tunnel syndrome, whose signs comprises ache, tingling and emotionlessness in the hand, wrist and arm, can be caused by constant hasty practice of the fingers, and is common those who use keyboards recurrently.
- b) Overexposure to Visual Display Units (VDUs). Headaches and muscular dysfunction may results from overexposure to computer terminals, while electromagnetic radiation produced by VDUs has been connected to failures and serious medical conditions including cancer (Coghill,1990).

Psychological Forms of Technostress

This form of technostress is more complex in nature. An individual feels technostress because of his milieu such as poor quality gadgets in workplace and need of the timely updating considering the rapid change

Symptoms of Technostress

Technostress always shows some symptoms which one can understand the intense of stress. There are mainly two types of symptoms such as psychological symptoms and Physical symptoms.

Psychological Symptoms

One of the primary psychological symptom is anxiety. This nervousness is articulated in various ways: petulance, headaches, incubus, and confrontation to learning new things. Techno anxiety most frequently troubles those who feel overstretched by employers, peers, or one who have feels somewhat a cultural lag to approach new technologies.

Physical Symptoms

Brillhart (2004), adding to psychological signs such as mental weariness, insomnia, restlessness, explains physical symptoms in the form of headaches, furor, stomach and duodenal difficulties, heart attack and high blood pressure. Wang and Shu (2005) point out that slump in professional competence is a sign of technostress. When someone work with technological devices, they may struggle from increase in adrenalin levels, headache, heart block, digestive complications, violence, sleeplessness, asthma, tonus, increase in heart rate and blood pressure, and in some unusual situations they may shows signs of diabetes and cancer.

- · Emotional symptoms such as anxiety, depression, and irritability
- Behavioral symptoms such as procrastination, decreased productivity, and difficulty concentrating
- Cognitive symptoms such as forgetfulness, confusion, and difficulty making decisions
- Social symptoms such as isolation, difficulty communicating, and decreased face-to-face interactions
- Difficulty disconnecting from technology, leading to an inability to relax or enjoy leisure activities
- Difficulty in sleeping and insomnia as the constant use of technology disrupts the natural sleep pattern.

It's important to note that the symptoms of technostress can also be symptoms of other conditions, and it's essential to seek professional help if the symptoms persist or disrupt daily life.

Contributors for Technostress

Technostress is the dark side of technological usage. Like many other scientific intervention technological device has positive and negative aspects. Chiefly there are four contributors for technostress such as contributors of lack of professional development, contributors of software issues, contributors of hardware issues and contributors of lack of human relationship. Rosen et al. (1987) distributed the signs of technostress into three groups:-Anxious technophobe, Cognitive technophobe and Uncomfortable user.

Contributors of lack of professional development means each and every person who interact with technology in their profession should have essential knowledge and skill of practice in the field, only through the appropriate and timely training can achieve that expertise. Software issues includes difficulty in recall user names and passwords, too many email and complexity of terminology. Hardware issues consist connectivity problem, environment, high cost, power, hardware breakdowns, and computer breakdown. Technology devices affect human relationship in a great extent, low human interaction leads into great personal and social issues.

Normally technostress will happen in two contexts. First one is organisational context and second one is private context. In former situation people use the information technology for the purpose of profession while in the later context individual use the technology for the personal needs. In organisational context they may use different technologies such as network technologies, communication technologies, database and enterprise technologies and software application. In private usage an individual can easily reduce the usage of IT when he realize his technostress but in the case of work context it may lead into severe problems (Maier, 2014). In some cases both two context affect a person simultaneously.

Teachers and Technostress

The phenomena of technostress is presently getting attention in the educational milieu as novel technologies such as digital textbooks, cloud computing, interactive technology and even social media invaded new classrooms. Teachers are the vital part in the incorporation of technology to teaching-learning process. Law et al. (2008) indicated as teachers are the most prominent factor of educational technology in a research conducted in twenty-two countries and on 35000 teachers.

As in the case of every occupational group, techno-stress is as significant issue for teaching profession too. Because of the facilities it provides, ICTs have taken its place as a pedagogical tool in education in last few decades and been combined to all educational systems quickly. Instructors, who have a central part in the integration are affected from several elements. Techno-stress is one of these issues. Further to the shifting sense of learning, the nature of technology, technical assistance, incompetence to practise, idea of the school and social pressure are the reasons for technostress among teachers. Teachers specifically experience techno-stress in the practise of integrating novel technologies, which is a common situation come across with digital technologies. Furthermore, the continuation of the burden for technology incorporation in education both from the institutions and the public sphere, and the deficiency of awareness and support upshot in technostress among educators (Longman, 2013). "The digitalization of education can also be stressful for teachers. There might be various reasons for such technostress. First of all, the digitalization of work often creates a demand to learn new things and adopt new technologies, which requires extra effort and may lead to higher workload and time pressures. Thus, stress can also be seen as an indicator of learning. The digitalization of work might also force the teacher to change the way she or he works." (Tarafdar et al., 2015)

In addition, it has been exposed that Technological Pedagogical Content Knowledge (TPACK) which is teacher's capacity to combine content, pedagogy, and

technology meaningfully in schooling, and school support are connected with instructors' technostress. According to Joo's high level of TPACK and a high level of school support are connected with lower levels of technostress. According to Çoklar et al. (2016) there are five chief causes indicating technostress suffered by teachers such as individual problems, technical problems, education-oriented problems, health problems, Time problem. Teachers' technostress a relevant topic due to the intensity of the integration of new technologies into teaching. Furthermore, technostress seems to influence teachers' intentions to use technology (Joo, et. al., 2016), in general, high levels of technostress are found to be associated with lower job satisfaction and job performance, and even with intentions to quit (Tarafdar et al., 2015).

Earlier studies on the practise of computers for teaching and learning already stated that instructors are the key to technology implementation. For instance, educators' computer familiarities, teachers' apprehensions or approaches toward computer practise, and teachers' involvements with professional enlargement courses on computer use were important to technology incorporation in education (Joo et al., 2016).

Teachers are particularly susceptible to technostress due to the increased use of technology in education. Some of the factors that can contribute to technostress in teachers include: Inadequate training and support: Teachers may feel overwhelmed and stressed if they are not provided with adequate training and support in how to use technology effectively in their classrooms.

Technology Continuance

Technology continuance is an individual's long-term technology usage over a period of time. It is the one of the major reasons behind the issue of technostress which affect our day to day life unfavourably. Technology continuance refers to the ability of a technology or system to continue functioning without interruption. It can also refer to the process of maintaining and updating a technology to ensure its continued operation. This includes regular maintenance and upgrades to hardware and software, as well as the development of contingency plans to minimize the impact of potential disruptions. The Integrative Framework of Technology Use (IFTU) (Kim and Malhotra 2005) ranges study in the stream of IT continuance by exploring the active interaction of beliefs, intents, and persistence conducts. Hence, IFTU follows four renowned mechanisms; (1) technology acceptance model, (2) sequential updating mechanism, (3) feedback mechanism, and (4) repeated behavioural patterns.

Technology Acceptance Model (TAM, Davis 1989)

The Technology Acceptance Model (TAM) is a theoretical framework that explains how users come to accept and use technology. It is based on the idea that an individual's perception of the usefulness and ease of use of a technology are the primary factors that determine their acceptance and use of that technology. The model proposes that perceived usefulness and perceived ease of use are the key factors that influence an individual's attitude towards a technology, and that attitude in turn influences the individual's intention to use the technology, which ultimately leads to actual use. The model also suggests that external factors such as social influence, and perceived behavioral control can also affect the adoption of technology, as they can change the perceived usefulness and ease of use of technology, and the attitude towards it. TAM is widely used as a theoretical framework to study the acceptance and use of information systems, and has been applied to a wide range of technologies including, but not limited to, computer systems, mobile devices, and the Internet.

Technology Acceptance Model (TAM) introduced by Davis (1989) tries to describe why individuals accept information technology and it covers the future social psychological concepts by presenting two technological characteristic grounded attitudinal beliefs (figure 1). First one is an individual's perceived usefulness, which is demarcated as "the degree to which a person believes that using a system would enhance his/her job performance" and perceived ease of use that denotes to "the degree to which a person believes that using a system would be free of effort".

Figure 1

Technology Acceptance Model (Davis, 1989)



Famous technologist John Kupersmith, reference librarian at the University of California, Berkeley, who has studied about technostress in libraries, analyses hitches comprised in technostress and reference librarians into five distinct kinds:

Performance Anxiety

Kupersmith says: "It is hard– and stressful– to suppress one's anger at clumsy design when teaching a user how to get around in a frustrating system, yet we know that we must do this and project a positive attitude for the user's sake."

It is found in studies that 'apprehension and fear of the consequences of being unable to perform a task or of performing it at a level that will raise expectations of even better task achievement. Test anxiety is a common example of performance anxiety. Other examples include fear of public speaking, participating in classes or meetings, playing a musical instrument in public, and eating in public. If the fear associated with performance anxiety is focused on negative evaluation by others, embarrassment, or humiliation, the anxiety may be classified as a social phobia.' Here in this context being disturbed that you are being arbitrated by your skill to practice technology, particularly when trying to exhibit it to someone else.

Information Overload

This includes not just erudition about an ever-expanding set of novel resources and tools, but also hastily accomplishing a level of proficiency so that you can elucidate it to fresh or practised handlers.

Role Conflicts

It is a feeling that they are moving from traditional proficiency work to undertaking more wide-ranging tech based one, which appears like a downgrading of sorts of employees. Mainly there are two types of role conflict such as Intra-role conflict and Inter role conflict.

Organizational Factors

Institutions should have a clear policy on numbers of people desired and how much technology is required to support a specific task. When these numbers reduced or the lack of adequate technology happens, individuals feel overworked.

Pucci (2015) pointed out that "the symptoms caused by technostress often co -exist with other psychological and phenomena such as the syndrome of burnout derivative disorders from work alcoholism. These conditions are not disease but represent an important psychosocial risk factor. There is evidence that these factors can lead serious health consequences and produce in subjects' certain disease. For this reason, the stress derived from technological dependence, which is still a 25 underestimated phenomenon, is diagnosed in a late stage, often after serious events related to health such as heart attacks, for which prescribed a long rest period away from work. Also on the relational level technostress has a strong impact: the techno stressed subject reacts with the isolation and emotional closure, he /she has bouts of anger, conflicts with colleagues and family. These conditions often result in a syndrome of internet addiction (IAD- internet Addiction Disorder), a disorder that is intertwined with technostress". Karasek (1979) viewed that introduction of novel technology into the teaching area is creating stress by increasing professional responsibilities, while simultaneously dipping decision-making freedom.

Measurement Techniques for Technostress

There are few measurement tools related to technostress. Çoklar et al. (2017) describes the technostress scale levels and its Development'. The aim of the study is developing a likert type scale named as TTLDSC ("Teachers' Techno stress Level

Defining Scale") to find out the techno stress level of teachers. The sample of the study consist 395 teachers. The development of the scale included five steps such as, Forming the theoretical framework and pool of items, getting expert opinion, Testing validity and reliability, Factor analysis and Item discrimination. The tool consists 28 items based on five factors that is Learning –teaching process oriented, Profession oriented, Technical issue oriented, Personal oriented and Social oriented.

Diffusion of Innovations

Rogers (2003) introduced the term 'Diffusion of Innovations'. According to this theory innovation means an idea, exercise, or thing that is observed as novel by an individual or other unit of acceptance. Diffusion of Innovations is a concept that look for to clarify how, why, and at what rate novel notions and technology spread through cultures. Rogers describes five intrinsic features such as relative advantage, compatibility, complexity or simplicity, trialability, observability of innovations that effect a person's choice to accept or discard a novelty. Diffusion is the procedure by which an innovation is transferred through certain networks over time among the fellows of a public system.

The Diffusion of Innovation (DOI) theory is a model that explains how, why, and at what rate new ideas and technologies spread through a social system. it has been widely used to understand the adoption and implementation of new technologies and innovations in various fields. According to the DOI theory, the diffusion of an innovation occurs in five stages:

Knowledge

The first stage is when individuals become aware of the innovation and begin to learn about it.

Persuasion

The second stage is when individuals start to form an opinion about the innovation and decide whether or not to try it.

Decision

The third stage is when individuals make a commitment to adopt or reject the innovation.

Implementation

The fourth stage is when individuals put the innovation into use.

Confirmation

The final stage is when individuals evaluate the results of their adoption and decide whether to continue using the innovation or not.

DOI theory also introduces the concept of adopter categories, it identifies five categories of adopters: innovators, early adopters, early majority, late majority, and laggards. These adopter categories are based on their willingness to adopt new ideas and technologies, and they play a key role in the diffusion process. The DOI theory has been widely used in various fields such as marketing, sociology, and innovation management to understand how new products, services, and ideas spread in the market and society.

Studies Related to Technostress

Varanasi et al. (2021) conducted a study on technostress of teachers in lowincome Indian schools. Findings of the study shows that even though smartphones help the teaching and organizational tasks, use of smartphone significantly forecasts burnout among teachers, with technostress contributing a chief clarification for this correlation.

Wang and Li (2019) tried to explore the issue of technostress among teachers in higher education. One of the main objectives of the study is to find out the relationship between multidimensional technostress and job performance were analysed. The sample of the study included 343 teachers from universities in China. The instrument used for the study is a multidimensional person-environment misfit

frame work of techno stress. The findings of the study show that person - organization (P-O) misfit, Person - technology (P-T) misfit, person-people (P-P) misfit mostly captivated how teachers interfere with manifold facets of higher education context in a disparity way that seeds techno stress.

Dong et al. (2019) Exploring the structural relationship among teachers' technostress, technological pedagogical content knowledge (TPACK), computer self-efficacy and school support. The present study made a structural model among TPACK, computer self-efficacy, administration support, and collegial support, teachers' technostress which were observed through a combined tool modified from prior studies. The sample of the research consist 366K-12 in-service teachers in China.

Tarafdar et al. (2017) explains the technostress trifecta - techno eustress, techno distress, design and gives theoretical guidelines and an outline for research of technostress. It is a qualitative study to comprehend the existing academic explanations of technostress and examines the various causes.

Çoklar et al. (2017) describes the technostress scale levels and its Development'. The aim of the study is developing a likert type scale named as TTLDSC ("Teachers' Techno stress Level Defining Scale") to find out the techno stress level of teachers. The sample of the study consist 395 teachers. The development of the scale included five steps such as, Forming the theoretical framework and pool of items, getting expert opinion, Testing validity and reliability, Factor analysis and Item discrimination. The tool consists 28 items based on five factors that is Learning –teaching process oriented, Profession oriented, Technical issue oriented, Personal oriented and Social oriented.

Joo et al. (2016) conducted a study which targets to explore the structural associations between secondary school teachers and TPACK, their understanding of school backing for the practice of technology, technostress, and purpose to use

technology in Korea, where a SMART education programme has been declared newly for K-12 education. The study engaged structural equation modelling to find out the causal associations among the variables, and the sample of the study includes 312 secondary school teachers. The findings of the study point out that TPAK and school support had a significant impact on the Technostress. In addition, Technostress significantly influenced teachers' intentions to use technology.

Çoklar et al. (2016) conducted a study on determining the reasons of technostress experienced by teachers. It is a qualitative study. The objective of this research is to define the causes leading to technostress suffered by teachers, who are the addressee of an exhaustive use of technology as a consequence of an incorporation procedure to which they are exposed in this study. For this requirement, qualitative data were collected from 64 teachers, who advanced from the technology meticulously and themes were organised using 117 diverse thoughts after the content exploration.

Coklar et al. (2016) conducted a study on Techno-Stress Levels of Teachers Who Were Included in Technology Integration Processes. This paper, which designed in survey mode, examines the techno-stress levels of teachers in the technology integration processes and try to define the technostress levels of teachers. The tool used for the study was technostess scale for teachers developed by the researchers and it contains of 5-point likert type items. All teachers who worked in central regions of Konya province of Turkey in 2015-2016 academic year are considered as the population of the study and the sample selected using stratified sample method and it consists 370 teachers from various level of education. As per the result of the present study, teachers had medium level of general techno stress and medium level of social oriented technostress, and low-level profession oriented and personal oriented techno-stress. It also found that general technostress levels didn't varied based on gender, length of service, and it varied by average internet use time variable.

Syvänen et al. (2016) conducted a study to examine the relationship between ICT and technostress among Finnish teachers. The present study try to explore how demographics related with technostress and which elements cause to stress in Finnish school teachers. The investigators used the OPEKA online –self-evaluation questionnaire tool to collect the data and the sample consists of 2,741 school teachers in Finland. The results of the study point out, class teachers were less "technostressed" than Subject teachers, male teachers were less stressed than Females, teachers with 0–15 years of experience were less stressed than teachers with 16–30 years of working experience. The significant predictors of the techno stress were, ICT expertise, the concordance of the educational use of ICT with the teaching style, school support and Attitudes to the educational use of ICT. High amount of these predictors leads to the low level of technostress.

Harahap and Effiyanti (2015) found that high level of specific cognitive belief towards computer technology will decrease the level of technostress without being scared of having larger amount of workload and job uncertainty caused by their lack of computer skill. Sample of this study consists 152 vocational school teachers at a selected area in Indonesia. The result of the study recommended that technostress could be reduce by empowering teacher's belief through the engagement of ICT in their educational service.

Booker et al. (2014) conducted a study entitled as "A Model for Testing Technostress in the Online Education Environment: An Exploratory Study." It presents a tool to measure technology connected stress for virtual pupils. The study was conducted using a convenience population of pupils and the findings show there is an association among technostress and learner outcomes in the virtual milieu. The study explains the bivalent nature of their structural impact.

Maier (2014) conducted a study titled as "information technology as daily stressor: pinning down the causes of burnout". It clearly depicts the amassed dispersal and practise of information technology as main reason for employee's perception of work stress and burnout. The aforesaid, study goals to explore how technostress and work stress are interrelated and how workers respond to these stressors, and it theorizes the influence of technostressors and fatigue. In adding, this study tries to scrutinize the part of workers' professions and hence whether techno stress and its impacts are perceived inversely.

Rolon (2014) indicated that teachers still face the prime issue in integrating technology beyond different classroom activities. The growing use of ICT in the society also makes another burden for the teachers in satisfying their academic related -responsibilities assumed by administrators, to implement ICT at a top level of academic activities. The objective of this study is to explore the relationship among technology acceptance model, workloads, job insecurity and technostress in high school teachers in Indonesia.

Tarafdar et al. (2014) observed in their study that older people have felt less technostress than younger people due to maturity. The difference of results may be due to the level of technology used in their job and work environment.

Mahalakshmi et al. (2013) conducted a study on technostress among library professionals in India .The study reveals that younger people are more acquainted with new technology and so that they would experience a reduced amount of technostress'

Joo et al. (2012) observed that teachers' technostress levels raised up after implementing digital textbooks in South Korea. He explained that teachers felt exasperated when technological system let-downs happened, because they felt like they were incapable to regulate the condition. Also, the study described that direction for teachers on how to assimilate digital textbooks into their teaching was restricted, swelling teachers' psychological and physical overload for technology implementation."

Coklar et al. (2011) tried to find out different levels of technostress among social network users. The sample of the study includes 287 social network users in

Turkey. In this study an online questionnaire is used for the collection of data. This study is to examine the level of technostress among social network users. The study found that social networking users have "medium technostress level", and that social stress about the use of technology, remembering large amounts of passwords and usernames, anxiety about data loss, and the formation of a technology professional are the main causes of technostress. Adding, it was dogged that technostress altitudes differ grounded on gender, profession and age.

Fudail and Mellar (2008) employed 'teacher-technology environment interaction model' identify the problem of the stress affected by teachers while using ICT in the classroom. The chief findings of this research were the demonstration that teachers do suffer stress connected with the practise of technology in the classroom and the identification of reasons, indicators and managing tactics related with technostress in the schoolroom. This study, thus, opinions to another way of thinking about the glitches of applying e-learning by theorising some of these execution complications in terms of technostress and in specific of teacher-technology milieu fit.

Al-Fudail and Mellar (2008) conducted a qualitative study with nine in-service teachers to know their level of technostress. Observations of classroom teachings and face-to-face interviews discovered that there exists numerous elements instigating technostress such as technological system fiasco, inadequate technical and social support for technology practice, increased time for setting up and preparing class lectures, and inadequate institutional culture for technology implementation.

Murray and Rostis (2007) observed that the technology can create burnout and makes employees unproductive because of no interpersonal dealings.

Agut et al. (2001) explored the 'technostress and Burnout among Spanish Workers'. The study explains a constant connection between Technostress and particular measures of employees' well-being, such as burnout. Thus, when familiarizing any new technology at work consideration should be given not only to organisational and work aspects, but also the psychosocial variables such as attitude

towards ICT or ICT self-efficacy as it is these psychosocial variables that play a part in the association between ICT and reduced well-being. Furthermore, and against hopes, gender does not play a central role in neither technostress nor burnout. This suggests that the traditional sex role stereotypes, which consider ICT to be more linked to men, may be disappearing.

Table 3

Research Studies on Technostres	SS	
---------------------------------	----	--

Year	Author	Findings
2021	Varanasi et al.	Use of smartphone significantly predicts burnout among teachers, with technostress providing a chief clarification for this correlation.
2019	Wang & Li	Person - organization (P-O) misfit, Person - technology (P-T) misfit, person-people (P-P) misfit mostly captivated how teachers interfere with manifold facets of higher education context in a disparity way that seeds techno stress.
2016	Joo et al.	TPAK and school support had a significant impact on the Technostress. In addition, Technostress significantly influenced teachers' intentions to use technology.
2016	Çoklar et al.	Teachers had medium level of general techno stress and medium level of social oriented technostress, and low-level profession oriented and personal oriented techno-stress. It also found that general technostress levels didn't varied based on gender, length of service, and it varied by average internet use time variable.
2016	Syvanen et al.	Class teachers were less techno stressed than Subject teachers, male teachers were less stressed than Females, teachers with 0–15 years of experience were less stressed than teachers with 16–30 years of working experience. The significant predictors of the techno stress were, ICT expertise, the concordance of the educational use of ICT with the teaching style, school support and Attitudes to the educational use of ICT. High amount of these predictors leads to the low level of technostress.

$48 \ \mathrm{Influence} \ \mathrm{OF} \ \mathrm{Technostress} \ \mathrm{\&} \ \mathrm{Teacher} \ \mathrm{autonomy} \ \mathrm{on} \ \mathrm{Burnout}$

Year	Author	Findings
2015	Tarafdar et al.	high levels of technostress are found to be associated with lower job satisfaction and job performance, and even with intentions to quit
2015	Harhap & Effyanti	High level of specific cognitive belief towards computer technology will decrease the level of technostress. Technostress could be reduce by empowering teacher's belief through the engagement of ICT in their educational service.
2014	Tarafdar	older people have felt less technostress than younger people due to maturity
2013	Mahalakshmi et al.	Younger people are more acquainted with new technology and so that they would experience a reduced amount of technostress.
2011	Ahmet et al.	Social networking users have "medium technostress level", and that social stress about the use of technology, remembering large amounts of passwords and usernames, anxiety about data loss, and the formation of a technology professional are the main causes of technostress. Adding, it was dogged that technostress altitudes differ grounded on gender, profession and age.
2008	Fudail & Mellar	Teachers do suffer stress connected with the practise of technology in the classroom and the identification of reasons, indicators and managing tactics related with technostress in the schoolroom.
2007	Murray & Rostis	Technology can create burnout and makes employees unproductive because of no interpersonal dealings.
2001	Agut et al.	Constant connection between Technostress and particular measures of employees' well-being, such as burnout. Thus, when familiarizing any new technology at work consideration should be given not only to organisational and work aspects, but also the psychosocial variables such as attitude towards ICT or ICT self-efficacy as it is these psychosocial variables that play a part in the association between ICT and reduced well-being

The review of literature exposed that there are ongoing studies in the field of technostress in education all over the world. The majority of available research studies are from European countries. During the review procedure, the researchers witnessed the studies that controverted and supported each other.

Theoretical Overview of Teacher Autonomy

The very word Autonomy indicates the basic human requirement and capacity to perform independently. Exclusively in educational field teachers need right autonomy to act for the betterment of the teaching -learning process. NCERT (2014), mentioned, "The origin of Autonomy has been one of the central concerns of philosophers since ancient times. The concept first came into prominence in ancient Greece and was derived from the Greek words 'auto' (self) and 'nomos' (rule or law), meaning one who gives oneself their own laws". Immanuel Kant was the one of the chief exponents who expresses his views on the notion of autonomy which is based on his 'Moral theory' has great philosophical importance. To attaining self-sufficiency one can make himself free from external context that affects them. Autonomous individual has their own control and choice.

Emerging the Concept of Autonomy

The word autonomy derived from ancient Greek which denoted the political status of occupied cities that were reigned according to their own rules rather than those of subjugators. Cockeram (1968) documented firstly the term in 'English Dictionary: or an interpreter of hard English words,' as the 'liberty to live after one's own law'. It appeared first as a political notion, then used as philosophical thought which applied to individuals. Apart from politics and philosophy the concept of autonomy is seen in various fields like: psychology, anthropology, medicine, sociology, bioethics, artificial intelligence, business and organizational development and so on.

The word autonomy means self-directing freedom particularly moral independence. Maloney (1997), defines the "Autonomous Teacher as one who is

aware of why, when, where, and how pedagogical skills can be acquired in the self – conscious awareness of teaching practice itself". Huang (2005) highlighted three words, namely, willingness, capacity, and freedom and defined teacher autonomy as "teachers' willingness, capacity and freedom to take control of their own teaching and learning". Pearson and Moomaw (2005) defines teacher autonomy as teacher's freedom to make decisions concerning what is taught and how it is taught. Because teacher autonomy correlates with feelings of professionalism. Mausethagen and Mølstad (2015) stated teacher autonomy as pedagogical freedom and absence of control, the will and capacity to justify practices, and a local responsibility.

The concept of autonomy in education has been an area of increasing interest in recent years. The concept of autonomy in education is not new, but it has been more widely discussed and studied in recent years. The emergence of the concept of autonomy in education can be traced back to several factors, including:

Changes in the educational landscape: The increasing use of technology and the rise of online learning have led to a shift in the traditional model of education and have created new opportunities for teachers and students to have more control over their learning experiences.

The focus on student-centered learning: There has been a growing emphasis on student-centered learning, which emphasizes the importance of giving students more control over their own learning, and providing them with opportunities to take ownership of their education. The emphasis on teacher professionalism: The increasing emphasis on teacher professionalism has led to a greater focus on the importance of teachers having the knowledge, skills, and autonomy to make informed decisions about their teaching practice.

The increasing awareness of the importance of student well-being: There has been a growing awareness of the importance of student well-being, and the role that autonomy can play in promoting positive mental health and well-being among students. The demand for more individualized and flexible education: With the growing diversity of students and their needs, the demand for more individualized and flexible education has increased, which in turn has led to more focus on giving teachers and students more autonomy in their learning.

All in all, the concept of autonomy in education has emerged as a response to the changing needs and demands of education in the 21st century, and it continues to be an important topic of research and discussion in the field of education.

European liberal democratic and liberal humanist believed the notion of individual autonomy as the central to their ideology (Lindley, 1986) and Kant regarded this autonomy as the basis of human dignity (Hill, 1991). "Teacher autonomy can be defined as teachers' capacity to independently their actions within the given the given contextual frame of constraints in which they operate (Wermke & Höstfält 2014; Paulsrud & Wermke 2019).

Teacher Autonomy; Offshoot of Autonomy

The concept of teacher autonomy has rooted in the concept of freedom. It is a specific form of autonomy that refers to the freedom and authority that teachers have to make decisions and take actions in their classrooms without external interference. It is considered as an offshoot of the broader concept of autonomy in education, as it emphasizes the importance of giving teachers the autonomy to make informed decisions about their teaching practice. It includes autonomy over disciplinary practices, pedagogical autonomy, curricular autonomy, professional development, classroom environment, and assessment. Willner (1990) explains "an older concept of teacher autonomy, based on independence through isolationism and alienation, and a newer concept of teacher autonomy, based on collaborative decision making and the freedom to make prescriptive professional choice concerning the services rendered to students".

Teacher autonomy is considered as a key aspect of teacher professionalism, as it allows teachers to use their professional knowledge and skills to make decisions

that best support student learning. It also allows teachers to take ownership of their teaching practice, which can lead to greater job satisfaction and motivation. There are several factors that can impact teacher autonomy, including school culture, school policies, and the level of trust and support provided by administrators and colleagues. Factors such as these can either support or constrain teacher autonomy. Research has shown that teacher autonomy can have positive effects on student learning and teacher well-being. Teachers who have autonomy in their classrooms tend to be more motivated, have a greater sense of job satisfaction and have better relationships with their students. However, it is worth noting that autonomy should be balanced with accountability, as teachers need to be held accountable for the decisions they make and the actions they take in their classrooms. It is also worth mentioning that providing autonomy does not mean that teachers should be left alone and unsupported, but rather that they should be provided with the necessary resources and support to make informed decisions and take effective actions.

There is clear difference between independence and autonomy. Independence is related with doing things by oneself or not depend on others, autonomy indicates "to act freely, with a sense of volition and choice" (Deci, 1995). According to Little (1995) "total independence is not autonomy but autism". There is clear separation between teacher autonomy and teacher empowerment. While teacher empowerment labels the process of collective decision making that takes place when administrators of school empower teachers to involve in the cooperative process of participative leadership for the amalgamation of a more democratic and reasonable educational system (Short, 1994), teacher autonomy can be conceptualized as the perceived and real capacity to exercise control over teaching and assessment, curriculum development, school functioning, or professional development matters, within the limits of the educational goals accepted by the school community.

Teacher autonomy is not independence or isolation. It involves interdependence, responsibility, mutual support, professional discretion, and obligation to the educational community. Perceiving teacher autonomy as isolation justifies educational policies that impose practices of collaborative work. Little (1995) explains the teacher autonomy "as exercise of continuous reflection and analysis of one's own teaching process." It has been defined by Lamb (2008) as a commitment to empower the learners by creating appropriate learning spaces"

A compilation of explanations presented in literature discloses an extensive diversity of views in teachers and scholars. While some researchers have delivered direct definitions taken from prevailing literature on student autonomy and well-defined teacher autonomy as freedom for control over instruction (Shaw, 2002) or as the promotion of student autonomy (Thavenius, 1999), other researchers have observed the notion in a more inclusive way. For example, Barfield et al. (2002, p.3) outline teacher autonomy as "a continual process of inquiry into how teaching can best promote autonomous learning for learners" which contains, among other doctrines, action, co-operation, understanding of restrictions, and collective support. On this line of thought, other scholars have enlarged on the idea and explained it as a multidimensional capacity connected with shared decision making grounded on pupils' requirements and interests, teachers' self-regulation, professional competence, and freedom from outwardly forced plans (Castle & Aichele, 1994).

According to the description in the learners' dictionary "autonomy is the ability to make your own decision about what to do rather being influenced by someone else or told what to do rather than being influenced by someone else or told what to do (Sinclair, 1995). Anderson (1987) points out that teacher autonomy is a freedom from direction by others over professional action or growth." National Policy on Education (1986) recommended in its ninths chapter about 'The Teacher', "the status of the teacher reflect the traditional cultural echoes of our society; it is said that no pupil can rise above the level of its teachers. The government and community should endeavour to create conditions, which will help motivate and inspire teachers on constructive and creative line. Teachers should have the freedom to innovate, to devise appropriate methods of communication and activities relevant

to the needs and capabilities of and the concerns of the community". Walker (2016), states that "Teachers believe their classroom autonomy suffered during the No child Left behind era. According to just-realised federal data from the National Centre for Education Statistics (NCES), educationalists informed fewer classroom autonomy in school year 2011-2012 compared to 2003-2004".

Dealings between instructors and pupils was at the core of the educational practise; Enticing and retaining great eminence teachers was a prime requirement for education. The justification underlying this view was that upgrading the teaching profession led to enhancements in teachers' show, which eventually led to progresses in pupil education (Rudolph, 2006). Teacher autonomy consists control over instructional practices which makes them creative and gave a sense of confidence and comfort in their teaching competencies. It includes control over discipline that keystone of effective classroom. Control over the curriculum is another area. (Rudolph, 2006).

Self-directed teachers do not oversee accountability. "They give more importance to update their knowledge and never use the same lecture notes in the class year after which were prepared by them or other teachers. They never spoil the students by spoon-feeding them. They never get satisfied with what they have learnt but find delight in enhancing their knowledge and techniques of fostering learner autonomy every day. They share what they have learnt with their colleagues. To summarize, autonomous teachers work with learners and not for learners." (Raya & Vieiria, 2015)

Pedagogical autonomy refers to teachers' choice to define instruction objectives, amass teaching materials, select instructional methods, utilising resources, and design student assessment that complies with teachers' teaching philosophy and beliefs (Skaalvik & Skaalvik, 2014). According to this comprehensive definition, with the given autonomy, instructors do not only have the freedom to teach in the

style that matches their ethics and viewpoint, but also have the right to refuse the instructional materials, approaches, techniques and demands that they do not feel contented with. (Kameshwara et al., 2020)

According to Wilches (2007), the narrow pedagogical autonomy can be understood as teachers' liberty to design classroom instruction and pupil assessment, while other vital academic works, such as curriculum change and teachers' specialized training, fall out of these theoretical margins.

"Conditions of Autonomy" developed by Wallace and Little (1998)

Teacher autonomy refers to the extent to which teachers have the freedom and authority to make decisions and take actions in their classrooms without external interference. There are several theoretical frameworks that have been proposed to understand the concept of teacher autonomy. One of the most widely cited frameworks is the "Conditions of Autonomy" framework, developed by Wallace and Little (1998). This framework identifies four conditions that are necessary for teachers to have autonomy in their classrooms:

Professional Knowledge: Teachers must have the knowledge and skills to make informed decisions about their teaching practice. Professional trust: Teachers must be trusted by their colleagues, administrators, and the community to make professional judgments about their teaching. Professional community: Teachers must be part of a community of professionals who support and collaborate with one another. Professional autonomy: Teachers must have the authority to make decisions about their teaching practice and to be held accountable for the results of those decisions.

Self-Determination Theory

Another theoretical framework for understanding teacher autonomy is the "Self-Determination Theory" (SDT) developed by Deci and Ryan (1985). This theory posits that autonomy is a basic psychological need that leads to optimal

motivation and well-being. According to SDT, autonomy-supportive contexts (e.g. school) are those that allow individuals to experience their actions as self-determined, and those contexts that do not support autonomy lead to a decrease in motivation and well-being.

Additionally, the "Autonomy-Supportive Teaching" framework developed by Vansteenkiste, Simons, Lens, Sheldon, and Deci (2004) suggests that teacher autonomy can be increased by creating autonomy-supportive climates in the classroom, which includes providing choice and opportunities for personal relevance, fostering autonomy through autonomy-supportive communication, fostering autonomy through competence-supportive feedback, and promoting relatedness.

Overall, these theoretical frameworks suggest that teacher autonomy is a multi-dimensional concept that depends on various factors such as trust, knowledge, community, and autonomy-supportive teaching and learning contexts.

Types of Autonomy

There are three categories of autonomy according to Boson.

- 1. *Technical Autonomy*. The act of teaching a subject outside the framework of an educational institution and without the interference of an administrator.
- 2. *Psychological Autonomy*. It is a capacity which allows teachers to take more obligation of their own teaching; an internal transformation that may be supported by situational autonomy without being reliant on it.
- Political Autonomy. Control over the methods and content of teaching. (Benson 1996)

There are some other types of autonomy, including:

• Behavioral autonomy, which refers to the ability of an entity to perform tasks without direct human supervision.
- Operational autonomy, which refers to the ability of an entity to operate independently and make decisions based on its environment.
- Situational autonomy, which refers to the ability of an entity to adapt its behavior to changing circumstances.
- Tactical autonomy, which refers to the ability of an entity to make decisions and take actions in pursuit of a specific goal.
- Strategic autonomy, which refers to the ability of an entity to set its own goals and objectives.
- Functional autonomy, which refers to the ability of an entity to perform specific functions without external direction.

Types of Teacher Autonomy

Teacher autonomy refers to the level of freedom and decision-making power that teachers have in their classrooms and in their professional lives. There are several types of teacher autonomy, including:

- Curricular autonomy, which refers to the ability of teachers to design and implement their own curriculum, lesson plans, and teaching methods.
- Assessment autonomy, which refers to the ability of teachers to design and administer their own assessments and evaluate student learning.
- Professional autonomy, which refers to the ability of teachers to make decisions about their own professional development and career paths.
- Classroom management autonomy, which refers to the ability of teachers to make decisions about classroom management and discipline.
- Decision-making autonomy, which refers to the ability of teachers to make decisions about the day-to-day operations of their classrooms and to have a voice in school-wide decisions.
- Pedagogical autonomy, which refers to the ability of teachers to design and implement their own teaching strategies, methods and approaches.

• Technological autonomy, which refers to the ability of teachers to use technology in their teaching and their classrooms.

Transformative Autonomy

Transformative autonomy is one of the main subdivision of autonomy. Willy Nel (2015) explains Transformative Autonomy as 'the form of autonomy in which school role players, such as teachers, have the urge to be involved with fellow role players in education development initiative towards social reformation which contributes democracy'. Transformative autonomy refers to a type of autonomy that empowers individuals and communities to actively shape their own lives and the world around them. It is characterized by an emphasis on self-determination, agency, and the ability to make meaningful choices. In education, transformative autonomy refers to the ability of teachers and students to actively shape their own learning experiences, to engage in critical reflection, and to develop the skills and knowledge needed to participate in society as informed and active citizens. Transformative autonomy also emphasizes the need for social and political change, and aims to empower individuals and communities to challenge and transform oppressive systems and structures.

In practice, transformative autonomy is often associated with critical pedagogy, which aims to help students develop the critical thinking skills needed to question and challenge dominant power structures. It is also often associated with student-centered and constructivist approaches to teaching and learning, which prioritize student autonomy, agency, and self-direction in the learning process.

Stagnation in Professional Development

There is a report as named A Nation at Risk, which is published in 1983 and it was the one of the first exertions that initiated school reform movement. Based on this report there are explosion of rules and regulations which published to empower the efforts to improve the education system. As part of school reform movement, there arise the need of competent individuals in the field of teaching. To achieve the quality numbers in the field should be upsurge the feeling of professionalism in the field. This could be only be recognised if autonomous of teachers was accepted and executed (Rudolph, 2006).

If teachers are to be professionally empowered and uplifted, teachers must have the autonomy to recommend the best dealing for their pupils as doctors and lawyers do for their clients.

While most research supports the use of intrinsic reward to encourage teachers, teachers and principals point out that their biggest shortcomings are safety and autonomy (Nero, 1986). Autonomy is one aspect of teacher motivation (Khmelkov, 2000; Losos, 2000; White, 1992). A 1997 research by the National Centre for Education Statistics and some other studies have established that the degree of autonomy perceived by teachers is suggestive of present job satisfaction (Charters, 1976; Franklin, 1988; Gnecco, 1983; Hall et al., 1989; Pearson & Hall, 1993). And a majority of latest literature backings this dogma (Brunetti, 2001; Kim & Loadman, 1994; Klecker & Loadman, 1996; Ulriksen, 1996).

The The Buona Scuola or The Good School reform (La Buona Scuola, 2015, Law 107/15) was introduced in 2015 to address the main issues in education and to convert it into an effective education system." This reform suggests granting more autonomy to schools. "In fact, some studies have highlighted the negative effects of autonomy on other proximate factors such as teacher attendance, motivation, and stress which might, in turn, affect student achievement (Bardhan & Mookherjee, 2006; Lucia & Cristian, 2010). Empirical studies show ambiguous results concerning the effects of autonomy and its effective implementation is argued to be dependent on the context where it is implemented" (Hanushek et al., 2013; Faguet & Sanchez, 2008, Treisman, 2007).

A comprehensive explanation of educators' academic autonomy denotes to teachers' independence to explain instructional objectives, accumulate teaching

resources, select pedagogical methods, using resources, and plan pupil assessment that fulfils with teachers' teaching philosophy and principles (Skaalvik & Skaalvik, 2014). According to this comprehensive description, with the given autonomy, teachers do not only have the choice to teach in the style that equals their morals and thinking, but also have the right to reject the pedagogical resources, methods, and demands that they do not feel contented with."

Previous researches that observed the consequence of teachers' pedagogical autonomy on students' academic achievement have revealed some constructive outcomes. After studying the PISA 2015 data, Bédard (2015) stated that "teachers' pedagogical autonomy, together with school leaders' accountability, influence students' achievement scores across 65 nations. Likewise, Gurganious (2017) claimed that there was a positive relationship amid science teachers' perceptions of autonomy and district-level pupil achievement scores. Teachers who have a accountability in decision making at the classroom level are found to be more effective to advance students' academic performance (Khodabakhshzadeh et al., 2018; Berry et al., 2010; Hulpia et al., 2009)".

Teachers who works in Finland have high level autonomy, guaranteed by the lack of over control and unnecessary observations (Simola, 2005). Carlgren and Klette (2008) highlights "the influence of educational reforms on Nordic teachers' working conditions found that curricular documents in both Finland and Sweden place an emphasis on teachers as professional curriculum makers, provided with a large amount of flexibility and responsibility for creating a good learning environment".

Previous researches have shown that autonomy is indispensable for teachers to achieve intrinsic motivation and to withstand their emotional comfort (Skaalvik & Skaalvik, 2014).

Friedman (1999) introduced different levels of autonomy. They are as follows.

- a) **No Autonomy**. Teachers do not have any control to take any initiation to modify the curriculum or any other elements in the institution.
- b) Scant-Autonomy. Teachers have limited independence of choice within the clear margins of existing programs, rules and regulations as defined by the management.
- c) Moderate Autonomy. This is in way similar to the previous one where teachers are allowed or motivated to share and implement their ideas but are compelled to go through stern authorization procedures before executing.
- d) High Autonomy. teachers have freedom to innovate or modify the curriculum and methods, and execute them within the common boundaries which are previously granted upon principles and norms
- e) Complete Autonomy. Teachers are given complete freedom to innovate and employ new ideas, programs, or curriculum within generally accepted ethical and legal principles.

Teacher Autonomy and Learner Autonomy

There is a close a relationship between teacher autonomy and learner autonomy. Both are interdependent. Teacher autonomy enhances learner autonomy. Teacher autonomy refers to the degree of freedom and independence that a teacher has in making decisions related to their classroom and curriculum, while learner autonomy refers to the degree of independence and self-direction that a student has in their own learning process. Research suggests that increasing teacher autonomy can lead to increased learner autonomy, as teachers are better able to personalize instruction and create an environment that supports self-directed learning. On the other hand, when teachers have less autonomy and follow a more rigid curriculum, learners may have less opportunity to take ownership of their own learning.

Additionally, when teachers give students autonomy, it helps them to become more self-directed learners and to develop the skills and confidence they need to succeed in the classroom and in life. Teachers can do this by providing opportunities for students to make choices, set goals, and take responsibility for their own learning.

In summary, Teacher autonomy and Learner autonomy are related in the sense that when teachers have more autonomy, they can create an environment that supports self-directed learning, which in turn helps students become more autonomous learners.

Figure 2

The Autonomy Triangle (Vieria, 2008, p.200)



Studies Related to Teacher Autonomy

Lyle et al. (2022) conducted a comparative case study on instructionally focused education system (IFESs) in US. They investigated two IFESs in which efforts to advance instruction pushed against historic norms of teacher autonomy. The study found that these new systems are not at odds with teacher autonomy but rather these systems reflect a transition to move interdependent notions of teacher autonomy.

Elo (2021) conducted a study on perception of Finnish teachers on autonomy. Findings of the result reveals that upper secondary teachers experience high autonomy. Result also shows that teachers use their mutual autonomy from coworkers to make a school profile that chiefly improved their perceived individual autonomy. Li and Allen (2021) conducted a study on "three–level hierarchical linear modelling analyses of the relationship between political culture and teacher autonomy". The study investigates the relationship between teacher autonomy and political culture and how this association differs with teacher and school features. The study shows teacher autonomy associate positively to teacher work satisfaction and retention.

Collie (2021) collected data between March and May 2020 from 325 Australian teachers and found that autonomy-supportive leadership was associated with greater buoyancy and, in turn, lower somatic burden, stress related to change, and emotional exhaustion. In contrast, autonomy-thwarting or controlling practices involve pressuring individuals to feel, act, and think in particular ways, and have been positively associated with emotional exhaustion.

Salokangas et al. (2020) conducted a study on teacher autonomy among Irish and Finnish teachers. Results of the study show that teachers in both countries think themselves high independent in their teaching process and decision making.

Kameshwara et al. (2020) conducted a study on Teachers' Pedagogical Autonomy, Professional Development and Students' Digital Skills: This study was conducted in the context of the good school reforms, Italy with an attention only on the levels of autonomy and digital skills. The empirical analysis was conducted using the newest International Computer and Information Literacy Study (ICILS 2018) data to observe the relation between teachers' academic autonomy and pupils' digital skills. In this study the researchers used the survey method to collect data. The study was also concerned in viewing at how educators' professional development interrelates with teachers' autonomy in determining pupil achievement. In the finding of the study, teacher autonomy displays a statistically significant positive effect on student achievement but not statistically significant association with students' digital skills performance.

Han (2020) conducted a study on the relationship between Teacher Autonomy and Learner Autonomy. This study indicates to foreign language learning

and teacher education in that language educators should modify their old-fashioned roles to ones catering and assisting the advance of learner autonomy, and teacher education and training programs should contain and plan courses, workshop to improve and endorse teacher-learner autonomy.

Paulsrud and Wermke (2020) conducted a study on perception of autonomy of Swedish and Finnish teachers. The survey method was used in this study. The sample of the study consists 708 Swedish and 1583 Finnish teachers. Result of the study shows Finnish teachers normally perceived themselves to be more individually autonomous, while Swedish teachers were more collegially oriented.

Paradise et al. (2019) conducted a study on "rational understanding of teacher autonomy: the role of trust for Canadian and Finnish teachers". The study explores the role of trust regarding teachers' perceptions of their own autonomy, entrenched in diverse milieus and dealings. The sample of the study consists upper-secondary school mathematics teachers in Canada and Finnish. The results of the study shows "that trust plays a significant role in whether teachers feel autonomous or not, and that trust plays a central role in relationships teachers have to work. Furthermore, trust is articulated differently according to the teachers' context. Based on findings, a reconceptualization of teacher autonomy is needed. A relational conceptualisation of teacher autonomy, highlights its inherent rationality and contextual sensitivity, provides new insight into the teachers' work, and supports more empowering ways for teachers to exert a proactive influence on their autonomy".

Prichard and Moore (2016) conducted a study on the balance of Teacher Autonomy and top down coordination in ESOL programs. The main objectives of the study is to recognize the levels of teacher autonomy. The survey method used for the collection of data from the sample which is included core faculty members from 130 programmes. The researchers used validated teacher autonomy scale by Pearson & Hall (1993), which measures both Curricular Autonomy (CA) General Autonomy (GA). The findings of the study shows the much lower level of agreement in Curricular Autonomy constructs and teachers have General Autonomy (GA)

Oberfield (2016) conducted a study on "teacher autonomy and accountability in traditional public schools and public charter schools". The study compares teachers' perceptions of autonomy and accountability in public charter schools and traditional public schools and in EMO-run (educational management organizations) and non- EMO – run PCS. It displays that PCS stated better autonomy than teachers in TPS; similarly, teachers in non-EMO-run schools indicated better autonomy than teachers in EMO-run schools. On the other hand there is no variance in perceptions of accountability across these various school types.

Wang and Zhang (2014) conducted a study on "promoting teacher autonomy through university-school collaborative action research". The study reports on collaborative action research project done by a group of university researchers with a group of senor secondary school English teachers in an effort to encourage teacher autonomy in the Chinese context.

According to Wilches (2007), the narrow pedagogical autonomy can be understood as teachers' liberty to design classroom instruction and pupil assessment, while other vital academic works, such as curriculum change and teachers' specialized training, fall out of these theoretical margins.

Rudolph and Supovitz (2006) conducted a study on types of teacher autonomy and tried to find out how it associated with job satisfaction. Four public school districts located in Michigan, New York, and Pennsylvania was included in the investigation. One of the main objective of the study is to know the extent does a leader allowing autonomy to teacher's influence teacher job satisfaction. The design of the study consisted the use of a survey along with interviews of heads of the institution and teachers. The study explored the succeeding areas of autonomy: freedom over curriculum, how leaders allowed independence to instructors, and what characteristics were considered to decide how much freedom was agreed to educators.

Pearson and Mooma (2005) conducted a study on the relationship between teacher autonomy and stress, work satisfaction, empowerment, and professionalism. Objective of this research was to check the connection between teacher autonomy and on-the-job stress, work satisfaction, empowerment, and professionalism. The assessment tool used for the study is a reliable and valid Teacher Autonomy Scale (TAS) which include curricular autonomy and general autonomy. The result of the study shows when the curriculum autonomy increased, the job stress reduced. The findings of this study also show that autonomy does not vary across teaching level (elementary, middle, high school).

Table 4

Year	Author	Findings
2022	Lyle et al.	New systems of education are not at odds with teacher autonomy but rather these systems reflect a transition to move interdependent notions of teacher autonomy.
2021	Elo	Teachers use their mutual autonomy from co-workers to make a school profile that chiefly improved their perceived individual autonomy.
2021	Li & Allen	Teacher autonomy associate positively to teacher work satisfaction and retention.
2021	Collie	Autonomy-supportive leadership was associated with greater buoyancy and, in turn, lower somatic burden, stress related to change, and emotional exhaustion. In contrast, autonomy- thwarting or controlling practices involve pressuring individuals to feel, act, and think in particular ways, and have been positively associated with emotional exhaustion.
2020	Salokangas et al.	Teachers in Irish and Finnish think themselves high independent in their teaching process and decision making.
2020	Kameshwara et al.	Teacher autonomy displays a statistically significant positive effect on student achievement but not statistically significant association with students' digital skills performance.

Recent Research Studies on Teacher Autonomy

Year	Author	Findings
2020	Paulsrud and Wermke	Finnish teachers normally perceived themselves to be more individually autonomous, while Swedish teachers were more collegially oriented.
2019	Paradise et al.	Trust plays an important role in whether teachers feel autonomous or not, and that trust plays a chief role in relationships teachers have to work. In addition, trust is expressed differently according to the teachers' milieu.
2016	Prichard & Moore	Lower level of agreement in Curricular Autonomy constructs and teachers have General Autonomy (GA).
2016	Oberfield	PCS stated better autonomy than teachers in TPS; similarly, teachers in non-EMO-run schools indicated better autonomy than teachers in EMO-run schools. On the other hand there is no variance in perceptions of accountability across these various school types.
2005	Pearson & Mooma	When the curriculum autonomy increased, the job stress reduced. The findings of this study also show that autonomy does not vary across teaching level (elementary, middle, high school).

Theoretical Overview of Teacher Burnout

Burnout is a burning issue in all professions. It is not only the issues of present but also it was a big problem in past and will be in future. The difference lays in the causes. Causes, components and levels of burnout may vary based on the nature of profession and whole environment of society. Burn-out is a condition conceptualized as consequential from long-lasting stress that has not been effectively coped or another word it is uncontrolled stress.

Definitional Aspect of Burnout

There is no one single definition of burnout which is collectively accepted. One central point that academics approve on is that burnout is a response to long-

$68\ {\rm Influence}\ {\rm of}\ {\rm technostress}\ {\rm \&}\ {\rm teacher}\ {\rm autonomy}\ {\rm on}\ {\rm burnout}$

standing stress at work. It is very useful to analyse the various definition of burnout to make this concept as clear as day light.

Table 5

Definitions of Burnout

Definer	Definition
Merriam Webster's Dictionaries	"Exhaustion of physical or emotional strength or motivation usually as a result of prolonged stress or frustration"
Freudenberger (1977)	"Physical and emotional exhaustion resulting from the excessive demands on energy strength or resources".
Fadden, M.C. (1980)	"Organizational burnout is a collection of symptoms which are characterized by low morale in the workers, declining in rates of overall production elevated levels of work absenteeism. Poor or in adequate communication among workers and increased level of job attribution."
Maslach (1976)	"Loss of concern for the people with whom one is working"
Cherniss (1980)	"A process in which a previously committed professional disengaged from his or her work in response to stress and strain experienced in the job."
Maslach (1982)	"An accumulation of intensive negative feeling that is so debilitating that a person withdraws from the situation in which those feelings are generated."
Maslach (1982)	"A condition produced by working too hard and too long in a high-pressure environment."
Maslach (1982)	"A process in which a professional's attitude and behaviour changes in negative ways in response to job strain."
Daley (1979)	"Burnout is the inability to handle stress on the job that results in demoralization, frustration and reduced efficiency".
(Maslach, 1976).	"A reaction to job related stress that results in the workers becoming emotionally detached from clients, treating clients in dehumanizing ways and becoming less effective on the job"

Maslach and Jackson (1981)	"Burnout is a syndrome of emotional exhaustion and cynicism that occurs frequently among persons who do "people work" of some kind and often results in the development of negative, cynical attitude and feelings about the client"
Paine (1982)	"Burnout Stress Syndrome (BOSS), the consequences of high level of job stress personal frustration and inadequate coping skills, has major personal organisational and social costs and these costs are probably increasing".
Maslach (1983)	"Burnout is physiological Syndrome of emotional exhaustion, depersonalization and reduced personal accomplishment that can occur among individuals who work with other people in some capacity."
Suran and Sheridan (1985)	"Burnout, is the feeling of personal/emotional depletion that arises from the stress that seems to be inherent in many professional careers."
Pines and Aronson (1988)	"Burnout is a condition of physical, emotional and mental exhaustion that is the result of chronic emotional strain."
Gaster and Schaubrock (1991)	"burnout is a type of stress - a chronic effective response pattern to stressful work conditions that features high level of interpersonal contact"
Brouwers & Tomic, 2000).	"It is a psychological syndrome of cynicism, emotional exhaustion, and reduced personal accomplishment which occurs among individuals working with other people".
Pareek (1983)	"Burnout is the end result of stress experienced, but not properly coped up, resulting in symptoms of exhaustion, irritation, ineffectiveness, inaction and several health problems".

Oranje (2001) divides studies on burnout into three categories. First, burnout is considered to be a coping problem (the interaction model), i.e. burnout derives from the negative outcome of an individual's judgment of their own abilities in relation to real or imagined stressors in their environment (Eskridge and Coker, 1985). Second, some studies view burnout as a state of both physical and mental

exhaustion that strikes the individuals involved for a long time. Third, some studies take the view that it is the environment that produces stressors responsible for the onset of burnout. Examples of such environmental stressors are the social relationships of the teachers with students, colleagues and principals (Brouwers & Tomic, 1999), and the organizational working circumstances (Van-Dierendonck et al., 1998). There are some general reasons for burnout which are may be seen in individuals' different profession and some specific reasons.

General Causes of Burnout

General causes can be classified into two; as environmental factors and personal or career factors. Freudenberg (1977) points out some environmental or organizational reasons such as over commitment, Extreme commitment, Lack of awareness of one's restriction. In addition to this reason, Cunningham, (1983) indicates few other general reasons like, Result of role vagueness, Workmate stress, unfitting incentive systems, Lack of promotional opportunities, Excessive work hours, insufficient monetary rewards, client overload, etc. (Cunningham, 1983; Edelwich, 1980; Glicken, 1983). Pareek used the word glow up to state the after effect of the stress and add some other contributors of burnout like level of stress, type of stress, non-work life, life style and coping style.

Specific Causes of Burnout

There are lot of specific causes of burnout. Farber, (1984) describes some of them as Impractical or excessive need for accomplishment, High principles, Workaholic propensities, Failure to make productive use of leisure, Idealistic prospects, Obligation without authority or resources to accomplish tasks, Depleting relationships with others and isolation (Farber, 1984; Glicken, 1983; Meier, 1983). Personality, Nature of the job or the role, Role style. Organizational climate are other contributors of burnout mentioned by Pareek.

Emotional exhaustion denotes to the state of mind of being emotionally worn out and strained. "Energy is lacking and low mood prevails. Depersonalization characterizes a distant and callous attitude toward one's job and the people connected to it (e.g., students, patients, clients). The worker shows weakened motivation and withdraws from an occupation that once sustained a great deal more interest for him or her. Lastly, reduced personal accomplishment includes feelings of inadequacy and incompetence associated with loss of self-confidence. This third dimension, however, was not present in the original working definition of burnout and its belonging to the burnout syndrome is doubtful (Hakanen & Schaufeli, 2012; Schaufeli & Taris, 2005).

Burnout as a Transactional Process

Burnout has been described as a transactional process, meaning that it is the result of the interactions between the individual and their work environment. This transactional process model suggests that burnout occurs when the demands of the work environment exceed the resources and coping mechanisms of the individual. Burnout is not an incident, on the other hand it is a process, a transactional process involving of phases. Cherniss (1980) suggested a three step Transactional model of Burnout as presented in Fig.

Figure 3

Transactional Model of Burnout [Cherniss, 1980]

STRAIN
Tension
Fatigue
Irritability
JOB STRESS
Demand
Resources
DEFENSIVE COPING
Emotional Detachment
Withdrawal
Cynicism
Rigidity

The initial phase titled as job stress consist of a disparity between resources and demand. The next step is an abrupt, short-term emotional reaction to this disparity. This phase which is known as Strain stage is featured by feelings of nervousness, pressure, weariness and exhaustion .The third phase which is known as defensive coping, involves of a number of variations in attitude and behaviour, such as, trend to deal clients in an isolated and mechanical style or a sceptical pre-occupation with indulgence of one's own requirements . Hence, Burnout is a transactional process comprising of Occupation stress, Employee strain and psychological coping.

Teacher Burnout

Professionals who related with human beings vastly happened to this burnout issue. Teaching profession highly related with students. So there are great possibility of burnout among teachers. Numerous reasons are there which can result in burnout among teachers. Pareek (1983) has pointed out nine factors which lead to burnout that is level of stress, type of stress, personality, and organizational climate, nature of job, life style, role style, coping style, and non-working life. Above these common reasons there are some specific reasons for burnout of unaided teachers.

Fiscal

One of the keen issue of unaided teachers are their low salary. Economic status of unaided teachers are in pitiable state. They are fighting to meet both ends of their life.

Lack of Training

Another stress is related to lack of training. Every year more than one thousand prospective teachers are completing their course. Most of them working in the field of unaided section. However it should be noted that sufficient in-service training are not available in the unaided sector. Problem of autonomy, problem related parents, issues related to pupils, issues of management and problem of attitude are other causes which lead into burnout. **High workload**. Teachers often have heavy workloads, including lesson planning, grading, and administrative tasks, which can be overwhelming and contribute to burnout.

Role Conflict and Ambiguity. When teachers experience conflicting demands or unclear expectations in their work, they may experience feelings of confusion and frustration, which can lead to burnout.

Poor Work-life Balance. When teachers are unable to balance their work and personal responsibilities, they may experience feelings of exhaustion and burnout.

Organizational Culture. Organizational culture can contribute to burnout when there is a lack of trust, poor communication, and low levels of support and recognition.

Stages of Teacher Burnout

Teacher burnout can have negative consequences for both the individual teacher and the education system as a whole, such as decreased job satisfaction, lower job performance, increased absenteeism and high turnover. Burnout is not an instant phenomenon; somewhat it is gradual in nature, comprising lot of steps or paces. Clouse and Whitekar (1981) indicated three phase of teacher burnout:

Phase 1-Loss of Enthusiasm

Every teachers enter to their profession with great passion. But when their expectation and can't fulfil, then starts loss of interest.

Phase II-Frustration

Frustration or desperation out of the nervousness of the profession is one of major staring sign of burnout.

Phase III-Alienation

Alienation is the third stage of burnout, where the teachers try to be detached from the students and co-workers.

Neuroticism

Neuroticism is a personality attribute; the word is used to explain persons who "express more negative emotions, emotional instability and stress reaction and, therefore, they become more vulnerable to both burnout and the majority of psychopathological disturbances" (Cano- García et al., 2005, p. 937; Watson et al., 1994)

Measurement Tools for Teacher Burnout

Teacher burnout scale was developed by the investigator with the help of her supervising teacher based on study by Steven A.Seidman(1986) .Based on this theory there are four components for teacher burnout which are career satisfaction, perceived administrative support, career satisfaction and attitude towards students. Here the investigator selected the aforesaid components for his tool. In fact Maslach Burnout Inventory is very popular but it is meant for general workers. So it didn't include the specific components related to teachers such as administrative and leadership skills. So that Steven A Seidman developed teacher burnout scale which consists four components such as Career satisfaction, Perceived administrative support, Coping with job related stress, Attitude towards students.

Oldenburg Burnout Inventory (OLBI)

This measure assesses burnout in four dimensions: exhaustion, detachment, personal accomplishment, and self-esteem. The Copenhagen Burnout Inventory (CBI): This measure assesses burnout in three dimensions: emotional exhaustion, cynicism and inefficiency.

Studies Related to Teacher Burnout

Tian and Guo (2022) conducted a study entitled as "how does transformational leadership relive teacher burnout: the role of self-efficacy and emotional intelligence." The study explained the basic mechanisms and boundary environments of the association between transformational leadership and teacher burnout. It explored the intermediating part of teacher self-efficacy and the regulating part of emotional

intelligence in this connection. The design of the study was structural equation model and hierarchical regression. The sample consists 539 primary and secondary school teachers. The result of the study showed that transformational leadership predicted teacher burnout negatively, self-efficacy mediated the relationship of transformational leadership with burnout partially, and emotional intelligence strengthened the relationship between transformational leadership and burnout.

Saloviita and Pakarinen (2021) conducted a study on teacher burnout in related to student and organisation-level variables consisting category of teacher, size of class, number of pupil. The sample of the study consist of 4567 primary school teachers in Finland. The findings of the study shows that the extent of burnout is different among various categories. Subject teachers possess highest level of burnout and size of class had a slight relation with teachers' burnout. It also pointed out that additional help was connected with lesser level of burnout.

Hassan and Ibourk (2021) conducted a study on "burnout, self-efficacy and job satisfaction among primary school teachers in Morocco". The present study objects the analysis the constructions of the scales including burnout scale. Sample of the study consists 404 teachers from El Kalaa Des Sraghna city. The result of the study established the two-dimensionality of the burnout measurement scale, depersonalisation and emotional exhaustion.

Sokmen and Kilic (2019) conducted a study to find out the relationship between between primary school teachers' autonomy and burnout variables. the research design used for the study consists quantitative research method and a correlational model. The sample of the study consists of 716 primary teachers. convenient sampling method was used. The findings of the study are shows that teaching self-efficacy predicted, job satisfaction, teacher engagement and teacher autonomy positively while it predicted burnout negatively at significant levels. As for teacher autonomy, it predicted teacher engagement positively.

Perrone et al. (2019) conducted a study on "administrative climate, early career teacher burnout and turnover." The study explores the relationship between administrative climate and early career teacher burnout. The sample of the study includes 184 early career teachers from Michigan Indiana. The result of the study shows that higher measures of administrative climate are strongly connected with lower levels of burnout.

Onal and Tatar (2017) conducted a study on Teacher burnout and participation in professional learning activities. This study try to find out the burnout and participation in professional learning activities among Trukish EFL (English as a foreign language). The sample of the study were collected from 224 Turkish EFL teachers through a questionnaire containing three segments, comprising the Maslach Burnout Inventory for Educators. The study revealed that "state university English Preparatory Program instructors (SUEPPI) and private university instructors (PUEPPI) significantly differ in terms of the levels of sense of personal accomplishment and participation in professional learning activities. Results of the study suggest that, SUEPPI did not feel as accomplished and did not participate in professional learning activities as repeatedly as PUEPPI, and instructors' sense of personal accomplishment is positively associated with their better participation in professional learning activities."

Esfandiari and Kamali (2016) conducted a study on the relationship between job satisfaction, teacher burnout, and teacher autonomy. The present study try to find out the association between teacher burnout, teacher autonomy and job satisfaction. The sample of the study includes two hundred and seven teachers and institute in Karaj and Tehran, they were given three tools for the study. The sampling method used in the study is convenience sampling. According to the result of this study teacher autonomy correlated negatively with job satisfaction and job satisfaction had a negative relationship with teacher burnout.

Jacobson (2016) conducted a study on Causes and effects of teacher burnout The aim of this case study was to investigate teacher burnout, its potential causes, and effects at an urban high school in the northeastern United States. The study's framework was based on Vygotsky's social development theory and the research questions focused on identifying the factors that contribute to teacher burnout and understanding teachers' perceptions of how burnout affects instruction and relationships with parents, colleagues, and administrators. The study used indepth interviews with five high school teachers who had experienced burnout as a method of data collection. The interview data was analysed and coded to identify recurring themes and patterns. A narrative was developed for each of the five teachers, and a cross-case analysis of the data was conducted. The results suggest that teacher burnout may be caused by various factors such as educational mandates, classroom discipline issues and it affects classroom instruction and interactions with all educational stakeholders. The study's implications for positive social change focus on identifying and eliminating the factors that contribute to teacher burnout in order to retain highly qualified and motivated teachers who will provide students with consistent, high-quality and equal educational opportunities that help them reach their full academic potential.

Bianchi et al. (2014) conducted a study entitled as "Is burnout a depressive disorder? A re-examination with special focus on atypical depression." The study examines the relationship between burnout and depression. The sample of the survey included .The present study proposes that the burnout–despair overlay has been largely misjudged. Atypical depression may account for a considerable part of this overlay. Generally, findings indicate to depressive indicators and depressive disorders as vital apprehensions in controlling of burnout. The experimental research on behaviours for depression propose resolutions that may support labours recognised as burned out.

Antoniou et al. (2013) explore the levels of occupational stress and professional burnout of teachers of primary and secondary education. It also objects to

investigate the coping strategies that they adopt, and the relationship between them. The survey involved 388 teachers who teach in public schools in Attica. Three instruments were administrated to teachers: "Teachers' Occupational Stress" (Antoniou et al., 2006), the Maslach Burnout Inventory (Maslach & Jackson, 1986) and the "Stress Coping Strategies Scale" (Cooper, Sloan, & Williams, 1988). The findings showed that teachers of Primary Education experience higher levels of stress compared to the teachers of Secondary Education. Female teachers experience more stress and lower personal accomplishment than men. Rational coping behaviours are a resource which help teachers overcome work-related stressors and burnout and achieve their valued outcomes with students, while avoidance coping predicted high level of stress and burnout.

Antoniou et al. (2013) conducted a study entitled as "Occupational Stress and Professional Burnout in Teachers of Primary and Secondary Education: The Role of Coping Strategies". The study tried to find out the levels of occupational stress and professional burnout of primary and secondary education. It also analyse the coping strategies that they adopt and the link between them. The sample of survey included 388 teachers who work in public schools in Africa. Three instruments were administrated to teachers: Teachers' Occupational Stress" (Antoniou, et al. 2006), the Maslach Burnout Inventory (Maslach & Jackson, 1986) and the "Stress Coping Strategies Scale" (Cooper et al., 1988). The study found that primary teachers experience higher levels of stress compared to the secondary teachers. Male teachers experience less stress and higher personal accomplishment than female.

Pandey (2012) conducted "a comparative study of teacher burnout, job satisfaction and job involvement among private and government school teachers". Main objectives of the study are to compare burnout of secondary school teachers with respect to their age, experience, gender, and educational qualification. The study find out that private school teachers sense more emotionally exhausted and depersonalized in contrast to government school teachers.

Jude (2011) studied the collaborative and comparative effect of emotional intelligence and locus of control on burnout among secondary school teachers. The design of the study was ex-post facto research design. Sample of the study consists 300 secondary school teachers which collected using stratified random sampling technique. Three tools were used for the study namely emotional intelligence scale, locus of control scale and teacher burnout scale. The result of the study showed that locus of control and emotional intelligence when taken as a complete significantly forecast the burnout.

Shukla (2009) conducted a study to know the level of teacher burnout in relation to job satisfaction among unaided engineering college teachers in Jalgaon District. The investigator used the burnout inventory and job satisfaction scale. The study shows that workers with burnout signs have a tendency to take away emotionally from the demands of the job.

Patricia (2009) examined the burnout of teachers in the public schools affecting the value of education the pupils were receiving. The objective of the study was to know the relationship between creativity and the intensity of the indicators of burnout in teachers. The result of the study shows there were no correlation between creativity and burnout components. On the other hand, there was a negative relationship between years of experience and depersonalization.

Pillay et al., (2005) conducted a study on Well-Being, Burnout and Competence: Implications for Teachers. This study is about the relationship between burnout and expertise for a sample of mid-career teachers in primary and secondary schools in Queensland. The findings break new ground in reporting a negative association between the MBI subscale Depersonalization and competence that may be attributed to a distancing mechanism in difficult human interactions. Overall, the assumptions of this study hold implications for teacher training courses and the wellbeing and competence of teachers.

Seidman and Zager (1986) developed a 21-item Likert scale to measure burnout among public school teachers. The validity of the Teacher Burnout Scale is

supported by the results of a factor analysis of data from 365 teachers in which the 4 factors that emerged are consistent with the theoretical constructs believed to underlie burnout. The factor analysis determined the subscales of the instrument: (1) Career Satisfaction, (2) Perceived Administrative Support, (3) Coping with Job-Related Stress, and (4) Attitudes towards Students. Additional analyses with 490 teachers indicated that the Teacher Burnout Scale has good internal consistency, test–retest reliability, and construct and predictive validity.

Conclusion

Analysis on related literature exposed a number of studies on teacher autonomy and teacher burnout. But review shows that there are only few studies on the variable technostress of teachers that too are conducted in other countries especially European countries. Studies on technostress among teachers are very rare in India. Especially the studies to know the relationship between technostress and burnout are less. Therefore the investigator claims that the present study is a relevant one. The above reviews of related literature give a broader outlook of the present problem. Various studies disclose the nature of the research. There are studies to explore the relationship of self-efficacy, leadership, locus of control, and emotional intelligence to the burnout (Jude, 2011; Tian & Guo, 2022; Hassan & Ibourk, 2021; Sokmen & Kilic, 2019). However there are found less studies know the relationship of technostress and teacher autonomy on burnout. These reviews of related literature made the investigator to feel a necessity to explore the influence of technostress and teacher autonomy on burnout among teachers.

From the review of literature related to the variables, it is apparent that most of the studies on technostress, teacher autonomy were conducted abroad. Apart from these, the investigator could not locate any study showing the interaction of technostress and teacher autonomy on burnout among unaided secondary school teachers. In this context the present study is found significant.

Chapter 3

METHODOLOGY

9

ব

- ✤ Variables of the study
- ✤ Objectives of the study
- ✤ Hypotheses of the study
- * Tool used for data collection
- $\boldsymbol{\textbf{\diamond}}$ Sample for the study
- Data collection procedure, scoring and consolidation of data
- ✤ Statistical Techniques used for analysis of data

The present study is an attempt to find out the influence of Technostress and Teacher Autonomy on Burnout among unaided secondary school teachers of Kerala. It efforts to study the influence of two independent variables on the dependent variable burnout of teachers. The major purpose of the study is to identify the main effect and interaction effect of Techno stress and Teacher autonomy on burnout among unaided secondary school teachers of Kerala.

The study also aims to identify the individual and joint contributions of the independent variables (techno stress and teacher autonomy) on the dependent variable, teacher burnout. The methodology adopted by the investigator for the present study is described under the following heading viz,

- Variables
- Objectives
- Hypotheses
- Method used
- Tools used
- Sample selected for the study
- Data collection procedure, scoring and consolidation of data
- Statistical techniques used for analysis of data

The details of each of the above are given in the following.

Variables

The independent and dependent variables selected for the present study are the following.

Independent Variables

Two independent variables were selected for the study. They are.

- Techno Stress
- Teacher Autonomy

$82 \ \mathrm{influence} \ \mathrm{of} \ \mathrm{technostress} \ \mathrm{\&} \ \mathrm{teacher} \ \mathrm{autonomy} \ \mathrm{on} \ \mathrm{burnout}$

Dependent Variables

• Teacher burnout

Categorical Variables

Categorical variables selected for the study are gender, locale, type of school, educational qualification, subjects of teaching, and teaching experience

Variables are diagrammatically represented in the following

Figure 4





Figure 5





Objectives of the Study

- 1. To find out the extent of technostress, teacher autonomy and burnout among unaided secondary school teachers of Kerala.
- To find out whether there exists any significant difference in technostress, teacher autonomy and burnout among unaided secondary school teachers based on relevant sub groups viz. gender, locale, type of management and subject of teaching.
- To find out the main effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant sub groups.
- To find out the first order interaction effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant sub groups.
- To find out the individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.
- 6. To develop regression equation to predict teacher burnout from technostress and teacher autonomy.

Hypotheses of the Study

- There exists significant gender difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- 2. There exists significant locale difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- There exists significant school difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.

- 4. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on educational qualification.
- There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers-based subject of teaching.
- 6. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on teaching experience.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for total sample.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for male unaided secondary school teachers.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for female unaided secondary school teachers.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for rural unaided secondary school teachers.
- 11. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for urban unaided secondary school teachers.
- 12. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for CBSE unaided secondary school teachers.

- 13. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for state syllabus unaided secondary school teachers.
- 14. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for basic qualification unaided secondary school teachers.
- 15. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for highly qualified unaided secondary school teachers.
- 16. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for language unaided secondary school teachers.
- 17. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for social science unaided secondary school teachers.
- 18. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for science unaided secondary school teachers.
- 19. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for mathematics unaided secondary school teachers.

- 20. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for novice unaided secondary school teachers.
- 21. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for less experience unaided secondary school teachers.
- 22. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for more experienced unaided secondary school teachers.
- 23. There is significant individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.

Method of Study

The present study is aimed to investigate the influence of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala. Survey method was used by the investigator to find out the influence of independent variables on dependent variable.

Tools Used for Data Collection

The selection and design of suitable instrument and Collection of appropriate data is a chief part of any research work and indispensable for a successful research. The data required for the study was collected by using three tools. All of them are constructed and standardized by the investigator. The tools used for the present study are the following.

- 1. Technostress Scale (Shafeek & Koya, 2019)
- 2. Teacher Autonomy Scale (Shafeek & Koya, 2019)
- 3. Teacher Burnout Inventory (Shafeek & Koya, 2019)

Detailed description of the construction of each scale is given below under the following headings.

- Planning of the scale
- Preparation of the scale
- Try out of the preliminary scale
- Item analysis
- Finalization of the scale

Technostress Scale

Planning of the Scale

The first step of the construction of scale is planning. The present study is an attempt to find out the level of technostress among secondary school teachers. Hence the investigator attempts to develop a scale on technostress for unaided secondary school teachers on the basis of objectives of the study by discussing with the supervising teacher, informal discussion with teachers, practitioner and experts in the field of education and after a crucial analysis of review of related literature. From the review investigator identified the theoretical constructs of technostress. Investigator also searched existing tools for measuring technostress. Studies used interview, opinionnaire, scale etc. It was found that most of them were developed and used for measuring technostress in IT professional context. The investigator could not find out appropriate instrument used for technostress among teachers in Kerala. So, the investigator himself developed a technostress scale. It includes two parts. Part A consists of personal information and Part B consists of 47 questions.

Preparation

The draft scale composed of two parts. Section A and section B. Section A contains personal information which is prepared to collect general data that is name of the teacher, name of the school, locale, type of school, subject, teaching

experience, educational qualification and gender. Section B consists of 47 items. The investigator wrote all items by the help of supervising teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The brief report on the two sections described under below

Technostress scale was developed by the investigator with the help of his supervising teacher based on the study by Tarafdar et al. (2007). Based on this theory there are five components for techno stress which are techno-overload, techno-invasion, Techno insecurity, techno complexity and techno uncertainty. Here the investigator through the discussion with experts sum up the dimensions of technostress of teachers as

- Techno Insecurity
- Techno Complexity
- Techno Invasion
- Techno Awareness
- Techno Facility

Description of each of these components is given in the following.

Techno Insecurity. Techno insecurity is the one of the serious issues caused by technological devices. It is related with circumstances where individuals feel endangered about losing their data especially personal and official secrets. Technoinsecurity arises when a user minds nervous about losing his or her profession to other co-workers who have better knowledge and skills with the technology compared to him or her. Techno-insecurity has been linked to situations where users face threats about technology due to automation from ICTs or other persons who have a improved understanding of ICTs.

Techno insecurity outlines a condition where teachers feel doubtful about the existence of technology because it is dreaded that it can substitute the part of teachers in educational process. It comprises intimidations to the profession, promotion skill sets to evade being interchanged, co-teachers as a menace, anxiety of being substituted by contemporaries, self-assurance in using technology to teach, nervousness in using technology to teach, and familiarity in functioning instruction plans. (Christian et al., 2020)

Item numbers 7, 8, 9,10,11,19 come under this category. 6 positive items prepared for this component.

Illustration.

- I have lot of concerns about the loss of privacy through online systems.
- I am afraid of cyber-attacks like hacking

Techno Complexity. Techno-complexity defines conditions where the complexity connected with ICTs can lead users to experience inadequacies with their computer skills and forces them to employ time and energy in learning and understanding ICTs. It indicates the complex technological arena including new updates in the field of technology which are force to teachers or any workers to use lot of time and energy in learning and acquiring the knowledge how to practice new applications and software. Besides, due to complication of the technology, teachers must apply more time to learn what way to practice technology in education process. Stress happens when they sense that the diversity of applications and functioning is threatening and they do not comprehend the necessity of practicing it. Additionally, persistent variations of technology being a reason to make users as strained and discontented with the whole system (Ibrahim et al., 2015).

Techno complexity includes complex to comprehend, complex to practise efficiently, and an ample amount of time and endeavour. Educational practices with the use of technology turn into more convoluted than traditional approaches where there is only human interaction. There are several complicated issues relating to the

integration of technology with education. (Christian, M., Purwanto, E., & Wibowo, S. (2020). Techno complexity includes techno uncertainty too. Techno-uncertainty refers to contexts where continuing ICT changes and upgrades unsettle users and create uncertainty so that they must constantly learn and educate themselves about new ICTs.

Item numbers 2, 3, 4, 18, 20, 23, 30, 41 come under this category. 8 positive items are prepared for this component.

Illustration.

- I am forced to make changes in teaching preparation to adapt new technology.
- I think that the information received through the internet is not reliable

Techno Invasion. Techno-invasion denotes a condition that users feel that they are never free from technology. They belief that technology has occupied their all interactions. (Ibrahim et al., 2015). Here it includes techno overload and techno overuse. Techno overload denotes a condition where teachers are forced to work quicker as a result of the use of connected ICTs each and every time, eventually they are compelled to be the part of increased workload. As a consequence of this overload teachers suffering work exhaustion and other health glitches. (Christian et al., 2020).

Techno invasion denotes a situation in which technology invaded all life of a person. We concentrate on techno-invasion as a special factor of technostress, which describes the excessive use of ICT, a state in which employees can be reached at any time and are compelled to always be connected (Tarafdar et al., 2010; Tarafdar et al., 2011). Techno-invasion describes the invasive effect of ICTs in situations where employees can be reached anytime and feel the need to be constantly connected, thus blurring non-personal and personal contexts.

Techno overload happens when users are incapable to recognise what is truly beneficial data. They use more time and energy in information handling because they transfer more information than is essential and obtain more data than they can efficiently process. Users also receive more information than they can process and use meritoriously. (Ibrahim, & Yusoff, 2015) Techno-overload describes situations where ICTs force users to work faster and longer.

Item numbers 5, 6, 12, 13, 14, 15, 16, 21, 22, 29, 45 come under this category. 1 negative item and 10 positive items are prepared for this component.

Illustration.

- My personal life has become chaotic due to new technology
- I felt some physical difficulties when using technology for a long time

Techno Awareness. Techno awareness refers to a person's understanding and knowledge of technology and its capabilities. It includes understanding the potential benefits and drawbacks of technology, as well as the ability to use technology effectively and safely. Being techno aware also includes keeping up-to-date with new developments in technology and understanding how to use new technologies for personal and professional benefit. Everyone who lives in this tech-world should have good awareness on different types of technology and should have a positive attitude.

Item numbers 1, 24, 25, 26, 31, 32, 33, 34, 36, 37, 39, 40 come under this category. 5 negative items (1, 26, 33, 34, 40) and 7 positive items are prepared for this component.

Illustration.

- In my opinion technology-based teaching is better than any other method of teaching.
- I think that the teaching method using technology adversely affects the creativity of the teacher
Techno Facility. It is related to the technological facility available for the teachers. Lack of adequate facility for the integration of technology is one of the big issue of teachers who want to adapt new ways of teaching. Even now most of the schools have only nominal facilities.

Item numbers 17, 27,28,35,38, 42, 43, 44,46,47 come under this category. 10 items are prepared for this component.

Illustration.

- The availability of apps and software related to my subject is low
- I didn't get adequate facility in classrooms for practicing technology based teaching methods.

Preparation

The draft scale composed of two parts namely personal information, statements part. Personal information is meant to collect general data name of the teacher, name of the school, locale, type of school, work experience, educational qualifications, IT literacy, subject and gender. Second part consists statements based upon the above-mentioned components. The investigator wrote all items by the help of supervising teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The draft scale consists of 47 items of which 37 items are positive and 10 items are negative. Positive and negative items listed below.

Type of items	Question numbers
Positive items	2,3,4,5,6,7,8,9,10,11,13,14,15,16,17,18,19,20,21,22,23,
	24,25,27,28,29,30,31,32,36,37,39,44,45,46,47
Negative items	1,12,26,33,34,35,38,40,42,43

Details of Negative and Positive Items in Technostress Scale

Scoring Procedure

As the present scale is a Likert type scale, responses should be made in a fivepoint scale as strongly agree, Agree, Undecided, Disagree and strongly disagree. For the positive statements the respective score of the five responses are 5,4,3,2, and 1. For the negative statements the scoring is done in the reverse order. The scores on all the items are added to get the total score on Technostress.

Pilot Testing

The draft scale was administered to representative group of 395 unaided secondary school teachers. Due representation was given to different subsample while selecting the sample and some of them are partially completed the scale. So Incomplete response sheets were cast-off. After random rejection, the sample size was fixed as 370. The response of each item by all the teachers in the sample were scored and subjected to item analysis. The draft version of the Technostress Scale (Malayalam) is presented as Appendix I.

Item Analysis

Item analysis is done using the method suggested by Likert(1932). The responses collected from the 370 teachers were first organized in ascending order from low scoring teachers to higher scoring teachers. The upper 27 percentages and lower 27 percentage of teachers' responses sheet were taken separately for item analysis. The number of teachers in lower and upper 27 percentages is 100 each. The responses for each item in lower and upper group were scored and subjected to item analysis. For this the t value for each item were calculated. The mean and standard deviation of each item were calculated separately for lower and upper groups. Items having the t value above 2.58 were selected for the final version of technostress Scale. The t values were calculated using the following formula

$$t = \frac{\overline{X}_{1} - \overline{X}_{2}}{\sqrt{\frac{SD_{1}^{2}}{N_{1}} + \frac{SD_{2}^{2}}{N_{2}}}}$$

Where,

- \overline{X}_1 = Mean for the first group
- \overline{X}_2 = Mean for the second group
- SD_1 = Standard deviation for the first group
- SD_2 = Standard deviation for the second group
- N_1 = Size of the sample for the first group
- N_2 = Size of the sample for the second group. (Best & Kahn, 2011)

The t values of each item are presented in Table 7

Data and Results of Item Analysis for Technostress Scale

SI No	High Group		Low Group		tvoluo	Status
51. INO.	Mean	SD	Mean	SD	- t value	(Accepted/Rejected)
1	1.98	1.00	1.37	0.72	4.935	Accepted
2	4.41	0.82	4.32	0.80	.786	Rejected
3	4.46	0.58	3.95	0.89	4.804	Accepted
4	4.34	0.67	3.72	0.95	5.317	Accepted
5	4.54	0.58	3.95	1.14	4.618	Accepted
6	4.18	0.77	3.39	1.20	5.551	Accepted
7	4.31	0.76	3.12	1.24	8.172	Accepted
8	4.25	0.76	3.60	1.04	5.039	Accepted
9	4.40	0.62	3.88	0.96	4.563	Accepted
10	4.79	5.02	3.41	1.06	2.692	Accepted
11	3.53	0.97	2.40	0.98	8.180	Accepted
12	2.40	1.12	1.95	0.93	3.099	Accepted
13	4.29	0.71	3.29	1.22	7.086	Accepted
14	4.17	0.85	3.38	1.14	5.535	Accepted
15	3.91	0.96	2.40	0.96	11.070	Accepted
16	4.02	0.89	2.62	1.17	9.533	Accepted
17	3.89	0.97	2.84	1.19	6.841	Accepted

<u> </u>	High (Group	Low	Group	, 1	Status
SI. No.	Mean	SD	Mean	SD	- t value	(Accepted/ Rejected)
18	3.69	0.93	2.10	0.85	12.651	Accepted
19	3.51	0.96	2.07	0.83	11.345	Accepted
20	3.34	1.03	2.17	0.92	8.478	Accepted
21	4.12	0.82	2.85	1.28	8.345	Accepted
22	4.44	0.76	4.15	1.02	2.286	Rejected
23	4.18	0.86	2.84	1.06	9.824	Accepted
24	3.33	0.97	2.24	0.92	8.121	Accepted
25	3.57	0.99	2.24	0.92	9.843	Accepted
26	2.09	0.89	1.86	0.79	1.933	Rejected
27	3.25	1.07	2.12	0.89	8.129	Accepted
28	3.60	1.08	2.10	0.95	10.424	Accepted
29	4.11	0.86	3.22	1.30	5.705	Accepted
30	3.70	0.85	2.58	1.01	8.513	Accepted
31	3.83	1.01	2.86	1.13	6.418	Accepted
32	3.17	1.14	2.25	0.94	6.245	Accepted
33	2.21	0.88	1.62	0.69	5.269	Accepted
34	1.70	0.56	1.56	0.74	1.505	Rejected
35	2.29	0.89	1.80	0.82	4.055	Accepted
36	3.90	0.88	2.36	0.96	11.820	Accepted
37	3.98	0.95	2.59	1.08	9.633	Accepted
38	2.15	0.90	1.74	0.76	3.473	Accepted
39	3.64	0.87	2.39	0.92	9.869	Accepted
40	1.98	0.92	1.70	0.82	2.267	Rejected
41	3.59	1.00	2.30	1.01	9.094	Accepted
42	2.47	0.96	1.90	0.86	4.430	Accepted
43	2.00	0.89	1.81	0.97	1.444	Rejected
44	3.49	1.12	2.44	1.10	6.666	Accepted
45	4.11	0.87	2.87	1.18	8.452	Accepted
46	3.85	0.90	2.67	1.05	8.498	Accepted
47	3.87	0.86	2.47	1.01	10.554	Accepted

Finalization of the Scale

Item which critical ratio greater than 2.58 the tabled value of 't' required for significance level at .01 were selected for the final scale. Thus, from the total 47 items, 41 items were selected for the final scale. A copy of the final version of the tool Technostress Scale (Malayalam and English version) are Appended as Appendices II and III respectively.

Establishment of Validity and Reliability. Burney points out that "validity is an indication of accuracy in terms of the extent to which a research conclusion corresponds with reality. "Validity of the present scale was ensured using face validity". A test is said to have face validity when it appears to measure whatever the author mind, what he thought he was measuring (Garret, 2005, p.355). Initially validity of the tool was established using face validity and content validity were established in consultation with experts. For establishing criterion related validity, the scores obtained for a representative group of 40 unaided secondary school teachers were correlated with an external independent criteria technostress scale (Ragunathan, 2002) which is appropriate for teachers. The validity coefficient obtained is 0.64

The reliability of the scale was established using Cronbach's alpha, α which is the most common measure of reliability. Cronbach's alpha is calculated measuring splitting data into every possible way and computing the correlation coefficient for each split. The average of these values is equivalent to Cronbach's alpha. The Cronbach's alpha obtained for the technostress is 0.76. Also, reliability of technostress Scale established using test retest method. The reliability value of technostress scale is 0.68.

Teacher Autonomy Scale

The tool Teacher Autonomy scale is constructed and standardized by the investigator with the help of supervising teacher. Detailed description of the construction of the scale is given below.

Planning of the Scale

The first step is the construction scale is planning. The present study is an attempt to find out the level of teacher autonomy among unaided secondary school teachers. Hence the investigator attempts to develop a scale on teacher autonomy among secondary school teachers on the basis of objectives of the study by discussing with the supervising teacher, informal discussion with teachers, practitioner and experts in the field of education and after a crucial analysis of review of related literature. From the review investigator identified the theoretical constructs of teacher autonomy. Investigator also searched existing tools for measuring teacher autonomy. Studies used questionnaire, scale etc. It was found that in the last decade we unconsciously neglected the autonomy of teachers while it became a serious issue. There are some tools to measure the autonomy but that is not fully appropriate in the present scenario of Kerala. So, the investigator himself developed a teacher autonomy scale. It includes two parts. Part A consists of personal information and Part B consists of 44 questions

Preparation

The draft scale composed of two parts namely personal information, and Section B. Personal information is meant to collect general data name of the teacher, name of the school, locale, and type of management, experience, educational qualifications, subject and gender. Section B consists of 44 item. The investigator wrote all items by the help of supervising teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The brief report on the two sections described under below

Teacher autonomy scale was developed by the investigator with the help of his supervising teacher based on the study by Friedman. Based on this theory there are six components for teacher autonomy which are establishing school identity and praxis, Teaching and achievement evaluation, Parental involvement, Staff development, extracurricular subjects and Curriculum change and development.

Here the investigator selected the aforesaid components for his tool. Description of each of these components is given in the following

Establishing School Identity and Praxis. Praxis denotes doing or action: the exercise or practice of an art, skill, or science (Chapman, 1999). School identity refers to the unique characteristics and qualities that distinguish a school from others. This can include things like the school's mission and values, its educational philosophy, and the types of programs and activities offered. A school's identity is often shaped by the community it serves, as well as the vision and goals of the school's leadership and staff. Praxis refers to the practical application of a school's educational philosophy and mission.

Item numbers 10, 11, 12,13,14,15,16,29,34 are comes under this component.

Teaching and Assessment. It refers to the ability of teachers to make decisions about the methods, techniques, and materials they use in the classroom, as well as the ability to evaluate student learning in ways that align with their own teaching philosophy and the school's mission and values. This can include things like designing lesson plans, and selecting assessment tools. Allowing teacher's autonomy in teaching and assessment can lead to more creativity, innovation and ownership among teachers.

Item numbers 1,2,6,7,17,20,21,28,30,31,40 are comes under this component.

Parental Involvement. Some parents may become overly involved in their child's education, which can lead to lack of autonomy for teachers and students. In unaided schools, the involvement of parents is extensive, which forces teachers to cater to their demands. As a result, the authority does not permit teachers to share or discuss the actual issues of students with their parents. Teachers are only allowed to highlight the positive traits of students. In unaided schools, parents are the source of funding, and as a result, they exhibit an overbearing attitude towards teachers, lacking respect.

Item numbers 3, 22, 23, 43, 44 are comes under this component.

Professional Development. It refers to the ability of teachers to pursue ongoing education and training to improve their skills and knowledge in their field. This can include attending workshops, conferences, and other professional development opportunities, as well as participating in online learning and self-directed study. Having freedom for professional development allows teachers to stay current with the latest teaching methods, technologies, and educational research.

Item numbers 9, 18, 19, 24, 35, 39, 42, are comes under this component.

Extracurricular Subjects. Teachers should have the necessary freedom to conduct extracurricular activities in order to promote the overall development of students. These activities enliven the academic experience. It refers to the level of independence and decision-making power that teachers have in organizing and running extracurricular activities. This can include clubs, sports teams, and other activities.

Item number 4, 8, 32, 38, 27, 33 are comes under this component.

Curriculum Development. Teachers should be granted the appropriate liberty to impart the curriculum in its entirety, in order to achieve the desired results. The opinions of teachers should be taken into account when any curriculum changes or developments occur. Keeping pace with the rapidly changing social, technological, and cultural landscape is crucial in the evolution of the curriculum.

Item numbers 25, 26, 36, 37, 41 are come under this component

Preparation

The draft scale composed of two parts namely personal information, statements part. Personal information is meant to collect general data name of the teacher, name of the school, locale, type of school, teaching experience, educational qualifications, IT literacy, subject and gender. Statements based upon the abovementioned components. The investigator wrote all items by the help of supervising

$100 \ \mathrm{Influence}$ of technostress & teacher autonomy on burnout

teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The draft scale consists of 44.

Items of which 31 items are positive and 13 items are negative. Positive and negative items listed below.

Table 8

Details	01	f Negati [,]	ve and	l Positi	ve Items	in	Teacher	Autonomy	Scale
	~	0						~	

Type of Items	Question Numbers
Positive items	1,3,4,6,8,9,11,12,13,14,15,16,17,18,20,21,23,27,28,31,32,33,
	34,35,36,37,38,39,40,41,42
Negative items	2,5,7,10,19,22,24,25,26,29,30,43,44

Scoring Procedure

As the present scale is a Likert type scale, responses should be made in a five-point scale as strongly agree, Agree, Undecided, Disagree and strongly disagree. For the positive statements the respective score of the five responses are 5,4,3,2, and 1. For the negative statements the scoring is done in the reverse order. The scores on all the items are added to get the total score on Teacher autonomy.

Pilot Testing

The draft scale was administered to representative group of 395 unaided secondary school teachers. Due representation was given to different subsample while selecting the sample and some of them are partially completed the scale. So Incomplete response sheets were cast-off. After random rejection, the sample size was fixed as 370. The response of each item by all the teachers in the sample were scored and subjected to item analysis. The draft version of the teacher autonomy scale (Malayalam) is presented as Appendix IV.

Item Analysis

Item analysis is done using the method suggested by Likert (1932). The responses collected from the 370 teachers were first organized in ascending order from low scoring teachers to higher scoring teachers. The upper 27 percentages and lower 27 percentage of teachers' responses sheet were taken separately for item analysis. The number of teachers in lower and upper 27 percentages is 100 each. The responses for each item in lower and upper group were scored and subjected to item analysis. For this the t value for each item were calculated. The mean and standard deviation of each item were calculated separately for lower and upper groups. Items having the t value above 2.58 were selected for the final version of teacher autonomy Scale. The t values were calculated using the following formula

$$t = \frac{\overline{X}_{1} - \overline{X}_{2}}{\sqrt{\frac{SD_{1}^{2}}{N_{1}} + \frac{SD_{2}^{2}}{N_{2}}}}$$

Where,

- \overline{X}_1 = Mean for the first group
- \overline{X}_2 = Mean for the second group
- SD_1 = Standard deviation for the first group
- SD_2 = Standard deviation for the second group
- N_1 = Size of the sample for the first group
- N_2 = Size of the sample for the second group. (Best & Kahn, 2011)

The t values of each item are presented in Table 9

Sl.	High	Group	Low (Group	. 1	Status
No.	Mean	SD	Mean	SD	t value	(Accepted/Rejected)
1	4.47	0.66	3.83	0.92	5.65	Accepted
2	3.78	1.01	3.35	0.97	3.07	Accepted
3	4.40	0.74	3.31	1.08	8.33	Accepted
4	4.27	0.75	3.13	1.03	8.94	Accepted
5	3.25	1.23	2.53	1.07	4.43	Accepted
6	4.18	0.76	3.02	1.04	8.99	Accepted
7	3.67	0.95	3.00	1.08	4.64	Accepted
8	4.58	0.62	3.63	0.97	8.24	Accepted
9	4.13	0.87	2.77	1.06	9.90	Accepted
10	3.45	1.10	2.63	1.10	5.29	Accepted
11	3.87	0.66	2.80	0.89	9.67	Accepted
12	3.65	0.95	2.76	1.04	6.34	Accepted
13	4.15	0.90	2.89	1.12	4.02	Accepted
14	3.55	1.08	2.56	1.07	6.53	Accepted
15	3.94	0.79	2.89	1.12	7.67	Accepted
16	3.77	0.83	2.73	0.95	8.25	Accepted
17	4.20	0.91	3.06	1.03	8.28	Accepted
18	3.99	0.93	3.18	1.05	5.79	Accepted
19	3.67	1.01	2.84	1.10	5.57	Accepted
20	3.40	1.10	2.62	0.95	5.36	Accepted
21	4.40	0.74	3.62	0.87	6.82	Accepted
22	2.78	1.12	2.41	1.07	2.38	Rejected
23	3.60	0.92	2.74	0.79	7.10	Accepted
24	3.23	0.83	2.65	0.91	4.71	Accepted
25	3.50	0.97	2.75	1.09	5.15	Accepted
26	3.03	1.00	2.55	0.93	3.52	Accepted
27	3.97	0.81	3.37	0.92	4.90	Accepted
28	4.10	0.97	3.33	0.91	5.79	Accepted
29	3.57	0.99	2.70	0.99	6.22	Accepted
30	3.26	1.12	2.45	0.95	5.54	Accepted

Data and Results of Item Analysis for Teacher Autonomy Scale

Sl.	High Group Low		Low C	Group	t voluo	Status
No.	Mean	SD	Mean	SD	- t value	(Accepted/Rejected)
31	4.12	0.86	3.40	0.94	5.65	Accepted
32	4.29	0.76	2.92	1.02	10.78	Accepted
33	3.48	1.11	2.64	0.98	5.66	Accepted
34	3.12	1.03	2.29	0.83	6.28	Accepted
35	3.25	0.98	2.59	0.82	5.18	Accepted
36	3.60	0.98	2.47	0.87	8.60	Accepted
37	4.53	0.63	3.67	0.85	8.12	Accepted
38	4.20	0.80	2.65	0.88	13.00	Accepted
39	4.43	0.57	3.28	0.92	10.59	Accepted
40	4.38	0.65	3.00	0.89	12.56	Accepted
41	4.40	0.62	3.54	0.87	8.06	Accepted
42	4.35	0.74	3.38	1.09	7.35	Accepted
43	3.87	0.66	2.80	0.89	9.67	Accepted
44	3.65	0.95	2.76	1.04	6.34	Accepted

Finalization of the Scale

Item which critical ratio greater than 2.58 the tabled value of 't' required for significance level at .01 were selected for the final scale. Thus, from the total 44 items, 43 items were selected for the final scale. A copy of the final version of the tool teacher autonomy scale (Malayalam and English version) are appended as Appendices V and VI respectively.

Establishment of Validity and Reliability. Donald H Mc Burney points out that "validity is an indication of accuracy in terms of the extent to which a research conclusion corresponds with reality. "Validity of the present scale was ensured using face validity". A test is said to have face validity when it appears to measure whatever the author mind, what he thought he was measuring (Garret, 2005, p.355). Initially validity of the tool was established using face validity and content validity were established in consultation with experts. For establishing criterion related validity, the scores obtained for a representative group of 40 unaided secondary

school teachers were correlated with an external independent criteria teacher autonomy scale (Friedman, 1999) which is suitable for teachers. The validity coefficient obtained is 0.78.

The reliability of the scale was established using Cronbach's alpha, α which is the most common measure of reliability. Cronbach's alpha is calculated measuring splitting data into every possible way and computing the correlation coefficient for each split. The average of these values is equivalent to cronbach's alpha. The cronbach's alpha obtained for the teacher autonomy is 0.781. Also, reliability of teacher autonomy Scale established using test retest method. The reliability value of teacher autonomy scale is 0.72.

Teacher Burnout Inventory

The tool Teacher Burnout inventory is constructed and standardized by the investigator with the help of supervising teacher. Detailed description of the construction of the inventory is given below.

Planning of the Inventory

The first step of the construction of inventory is planning. The present study is an attempt to find out the level of teacher burnout among unaided secondary school teachers. Hence the investigator attempts to develop an inventory on teacher burnout among secondary school teachers on the basis of objectives of the study by discussing with the supervising teacher, informal discussion with teachers, practitioner and experts in the field of education and after a crucial analysis of review of related literature. From the review investigator identified the theoretical constructs of teacher burnout. Investigator also searched existing tools for measuring teacher burnout. Studies used inventory, scale etc. It was found that in the last decade we unconsciously neglected the burnout of teachers while there are raised lot of new burnout creators. There are some tools to measure the burnout but that is not fully apt in the present scenario of Kerala. So, the investigator himself developed a teacher burnout inventory based on famous Maslach Inventory. It includes two parts. Part A consists of personal information and Part B consists of 28 items.

Preparation

The draft inventory composed of two parts namely personal information, and Section B. Personal information is meant to collect general data name of the teacher, name of the school, locale, type of management, subject and gender. Section B consists of 28 item. The investigator wrote all items by the help of supervising teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The brief report on the two sections described under below.

Teacher burnout inventory was developed by the investigator with the help of his supervising teacher based on the Maslach Burnout Inventory – Educator's Survey (MBI-ES, 1986). Based on this theory there are three components for burnout which are emotional exhaustion (EE), Depersonalization (DP), and Perished Personal Accomplishment (PA). Description of each of these components is given in the following.

Emotional Exhaustion (EE). Emotional exhaustion denotes to state of mind of being emotionally drained and exhausted of one's emotional strength consequence of long-term stress in private or work life. People undergoing emotional exhaustion feel like they have no control over what happens in their daily life. Lack of motivation, sleeplessness, and struggle in attention, nervousness and depression are the symptoms of emotional exhaustion.

Item numbers 1, 2, 3,4,8,13,20,25,26,28 are come under this component

Depersonalization (DP). Depersonalization denotes to an undesirable, insensitive, or extremely detached reaction to other people who are regularly the receivers of one's services or interaction. Depersonalization lead to heartless and destructive approach towards their clients and subsequently treat them impersonally by separating from them (Maslach et al., 2001).

Item numbers 5,6,7,9,10,11,12,14,18,22 are come under this component.

Inefficacy/Reduced Personal Accomplishment. Reduced personal accomplishment denotes to a decay in an individual's feelings of competence and affluent achievement in his or her work. Individuals those who suffers from diminished personal accomplishment feels a general discontent and frustration with themselves, their specialized skills, and their efficiency (Maslach et al., 2001).

Item numbers 15,16,17,19,21,23,24,27 are come under this component

Preparation

The draft inventory composed of two parts namely personal information, statements part. Personal information is meant to collect general data name of the teacher, name of the school, locale, type of management, work experience, educational qualifications, subject and gender. Statements based upon the abovementioned components. The investigator wrote all items by the help of supervising teacher. The items were subjected to expert criticism. Then some of the items were deleted and some were modified. The draft scale consists of 28 items of which 23 items are positive and 5 items are negative. Positive and negative items listed below.

Table 10

Type of Items	Question numbers
Positive items	2,3,4,5,6,7,8,9,10,11,12,14,15,16,17,18,19,20,21,23,24,25,26
Negative items	1,13,22,27,28

Details of Negative and Positive Items in Teacher Burnout Inventory

Scoring Procedure

The present inventory is a five-point scale in which items can be responded as strongly agree, Agree, Undecided, Disagree and strongly disagree. For the positive statements the respective score of the five responses are 5,4,3,2, and 1. For the negative statements the scoring is done in the reverse order. The scores on all the items are added to get the total score on Teacher Burnout.

Pilot Testing

The draft scale was administered to representative group of 395 unaided secondary school teachers. Due representation was given to different subsample while selecting the sample and some of them are partially completed the scale. So Incomplete response sheets were cast-off. After random rejection, the sample size was fixed as 370. The response of each item by all the teachers in the sample were scored and subjected to item analysis. A copy of the Burnout Inventory (Malayalam) is presented as Appendix VII.

Item Analysis

Item analysis is done using the method suggested by Likert (1932). The responses collected from the 370 teachers were first organized in ascending order from low scoring teachers to higher scoring teachers. The upper 27 percentages and lower 27 percentage of teachers' responses sheet were taken separately for item analysis. The number of teachers in lower and upper 27 percentages is 100 each. The responses for each item in lower and upper group were scored and subjected to item analysis. For this the t value for each item were calculated. The mean and standard deviation of each item were calculated separately for lower and upper groups. Items having the t value above 2.58 were selected for the final version of technostress Scale. The t values were calculated using the following formula

$$t = \frac{\overline{X}_{1} - \overline{X}_{2}}{\sqrt{\frac{SD_{1}^{2}}{N_{1}} + \frac{SD_{2}^{2}}{N_{2}}}}$$

Where,

 \overline{X}_1 = Mean for the first group

 $\overline{\mathbf{X}}_2$ = Mean for the second group

- SD_1 = Standard deviation for the first group
- SD_2 = Standard deviation for the second group
- N_1 = Size of the sample for the first group

$108 \ \mathrm{Influence} \ \mathrm{of} \ \mathrm{technostress} \ \mathrm{\&} \ \mathrm{teacher} \ \mathrm{autonomy} \ \mathrm{on} \ \mathrm{burnout}$

 N_2 = Size of the sample for the second group. (Best & Kahn, 2011)

The t values of each item are presented in Table 11

Data and Result of Item Analysis	for Teacher Burnout Inventory
----------------------------------	-------------------------------

Cl Ma	High C	High Group		Low Group		Status
51. INO.	Mean	SD	Mean	SD	t value	Accepted /rejected
1	1.74	0.97	1.23	0.42	4.82	Accepted
2	4.32	0.82	3.22	1.18	7.68	Accepted
3	4.05	0.94	2.40	0.94	12.42	Accepted
4	4.01	0.95	2.25	0.90	13.44	Accepted
5	2.69	1.06	1.77	0.63	7.45	Accepted
6	3.58	1.07	1.68	0.60	15.53	Accepted
7	4.09	0.83	3.04	1.20	7.21	Accepted
8	3.09	1.19	1.71	0.76	9.79	Accepted
9	2.55	1.00	1.52	0.64	8.67	Accepted
10	2.64	1.10	1.52	0.59	8.98	Accepted
11	2.41	0.91	1.57	0.64	7.55	Accepted
12	2.69	1.03	1.70	0.85	7.42	Accepted
13	2.47	1.04	1.98	0.94	3.49	Accepted
14	2.65	0.90	1.68	0.66	8.65	Accepted
15	2.74	1.09	1.54	0.63	9.56	Accepted
16	3.31	1.01	1.68	0.72	13.11	Accepted
17	4.45	0.76	3.60	1.31	5.62	Accepted
18	2.73	1.01	1.79	0.70	7.63	Accepted
19	3.10	0.99	1.88	0.84	9.38	Accepted
20	4.35	0.90	3.40	1.32	5.95	Accepted
21	2.26	1.00	1.62	0.74	5.15	Accepted
22	3.52	1.08	4.10	1.04	3.87	Accepted
23	3.72	0.98	2.21	1.01	10.77	Accepted
24	3.77	0.99	2.30	0.95	10.70	Accepted
25	3.69	0.97	1.86	0.77	14.80	Accepted
26	3.40	0.99	2.13	0.97	9.14	Accepted
27	2.79	1.07	1.56	0.61	10.02	Accepted
28	2.37	0.88	1.44	0.66	8.45	Accepted

Finalization of the Tool

Item which critical ratio greater than 2.58 the tabled value of 't' required for significance level at .01 were selected for the final tool. Thus, from the total 28 items, all of them were selected for the final tool. A copy of the tool Teacher Burnout Inventory (English version) are appended as Appendix VIII.

Establishment of Validity and Reliability. Initially validity of the tool was established using face validity and content validity were established in consultation with experts. For establishing criterion related validity, the scores obtained for a representative group of 40 unaided secondary school teachers were correlated with an external independent criteria teacher burnout scale (Seidman, 1986) which is suitable for teachers. The validity coefficient obtained is 0.52.

The reliability of the inventory was established using Cronbach's alpha, α which is the most common measure of reliability. Cronbach's alpha is calculated measuring splitting data into every possible way and computing the correlation coefficient for each split. The average of these values is equivalent to Cronbach's alpha. The Cronbach's alpha obtained for the teacher burnout inventory is 0.81. Also, reliability of teacher burnout inventory established using test retest method. The reliability value of teacher burnout inventory is 0.71.

Sample Selected for the Study

Unaided Secondary school teachers of Kerala are the population of the study and the sample for the present study is collected from this population. As it is impossible to collect data from a large number of populations the study was limited to 540 unaided secondary school teachers. Investigator made all effort to make the sample the true representation of the population.

Schools from Northern, Central and Southern part of the Kerala state were included in order to get a cross section of the Kerala state. Stratified sampling

technique was used to select the participants of the study. The participants were selected from 58 schools of ten districts of Kerala state. Thirteen schools from Kannur, three schools from Palakkad, four schools from Trivandrum, fifteen schools from Kozhikode, fourteen schools from Malappuram, four schools from Thrissur, two schools from Kollam, one school from Pathanamthitta, one from Ernakulum and one school from Kasaragod were selected randomly. Care was taken to include schools from urban and rural areas as well as government and aided schools.

Due representation as given to strata such as Gender, locale, and Stream of study. Samples were drawn from all districts of Kerala. Stratified random sampling techniques were used to select the sample. Data was collected from 560 unaided secondary school teachers. Some responses were not complete thus it reduced to 510. The details of the sample presented in table 12

Sl. No.	Name of Unaided School	Male	Female	Urban/ Rural	Type of School
1	Majlis English Medium Uliyil (Kannur)	3	9	Rural	CBSE
2	Cordova Public School (Trivandrum)	2	13	Urban	CBSE
3	Thunjan Smaraka School (Trivandrum)	2	10	Urban	State
4	KEMS Manjery (Malappuram)	4	4	Urban	CBSE
5	Majmau Therattammal (Malappuram)	3	7	Rural	State
6	Al Farook Residential School (Calicut)	5	10	Urban	CBSE
7	Chinmaya Vidyalaya (Trivandrum)	1	4	Urban	CBSE
8	Qadisssiyya English Medium School Feroke (Calicut)	1	9	Urban	State
9	Al-Azhar English Medium school, Manakkody (Trissur)	2	5	Urban	CBSE
10	Hidaya Public School, Manakkadav (Calicut)	1	8	Urban	CBSE

List of Schools and Number of Students for Each Category

Sl. No.	Name of Unaided School	Male	Female	Urban/ Rural	Type of School
11	The Oxford School (Trivandrum)	1	7	Rural	CBSE
12	V.M.M. English Medium School, Palottupally (Kannur)	2	9	Urban	State
13	Little Flower E.M. H.S. Kelakam (Kannur)	2	13	Rural	State
14	Tha-E-Leemul Islam Orphanage High School (Malappuram)	2	3	Rural	State
15	MET English Medium School(Palakkad)	1	7	Rural	CBSE
16	Al-Falah AMM English School Kakkidippuram (Malappuram)	1	3	Rural	CBSE
17	Ma'din Public School (Malappuram)	1	4	Urban	CBSE
18	Irshad English School Melattoor (Malappuram)	2	3	Urban	CBSE
19	Apex international school (Calicut)	3	6	Urban	CBSE
20	Al Maqar English Medium,(Kannur)	2	6	Rural	CBSE
21	Quadisiya Engliish Medium School, Thazuthala (Kollam)	2	8	Urban	CBSE
22	Oxford Pantheerankavu (Calicut)	2	13	Urban	CBSE
23	Gurudeva Public School Varakara (Trissur)	2	4	Urban	CBSE
24	Iqrah JDT Public School (Calicut)	2	8	Urban	State
25	Little Flower, (Calicut)	1	7	Urban	CBSE
26	DHEMS Nadapuram (Calicut)	2	14	Urban	CBSE
27	Hidaya Vellila (Malappuram)	2	8	Rural	State
28	St Paul's English Medium Higher Secondary School, Kohinoor (Malappuram)	1	8	Urban	State
29	MEMS School Chelari (Malappuram)	3	7	Rural	CBSE
30	Bhavan's Vidya Mandir (Pathanamthitta)	1	4	Urban	CBSE
31	White International School, PERUMANNA (Calicut)	Nil	5	Urban	CBSE
32	Bavan's Vidya Peedam, Chelembra (Malappuram)	2	15	Rural	CBSE
33	Markaz Public School, Aikarapadi (MPSA) (Malappuram)	1	8	Rural	State
34	Nivedita Vidya Peedam, Ramanattukara (Calicut)	2	13	Urban	State
35	Sappaire Olavanna (Calicut)	3	17	Rural	CBSE

Sl. No.	Name of Unaided School	Male	Female	Urban/ Rural	Type of School
36	CIRHSS (Calicut)	2	10	Urban	State
37	Mount Flower, Narayanpara (Kannur)	3	12	Rural	CBSE
38	Malabar English Medium School (Kannur)	1	9	Urban	CBSE
39	Sree Vidya Matttanur (Kannur)	3	9	Urban	State
40	Global Public School Vilakkod (Kannur)	2	11	Rural	CBSE
41	Niveditha Vidyalayam Punnad (Kannur)	1	5	Rural	State
42	Benhill (Kannur)	3	12	Rural	CBSE
43	MGM Kelakam (Kannur)	2	12	Rural	CBSE
44	Shanthinikethan English School, Peravoor (Kannur)	2	6	Rural	CBSE
45	Cresent Public School Chaliyam (Calicut)	3	10	Urban	CBSE
46	Al Hidayath English Medium School Kondotty (Malappuram)	2	5	Rural	State
47	MES English Medium School, Kannanallur (Kollam)	4	7	Urban	State
48	Alhuda EM H S Panaikulam (Eranakulam)	1	7	Rural	State
49	Islamic Central School Ottappalam (Palakkad)	2	Nil	Rural	CBSE
50	Izzathul Islam H S S Kuzhimanna (Malappuram)	2	Nil	Urban	State
51	Bukari English School Kondotty (Malappuram)	1	Nil	Urban	CBSE
52	St. Paul's HSS Kurichira (Thrissur)	2	Nil	Rural	State
53	Saadiya English Medium Residential Senior Secondary School, Deli (Kasargod)	1	Nil	Urban	CBSE
54	Moloor Central School Nellaya (Palakkad)	Nil	2	Rural	CBSE
55	Sri Sri Ravishankar Vidyamandir (Trissur)	Nil	1	Urban	CBSE
56	Safa English Medium School, Mattool (Kannur)	Nil	1	Rural	CBSE
57	Cresent Public school Mavoor (Calicut)	3	5	Urban	State
58	Ilahiya HSS Kappad (Calicut)	1	NIL	Rural	State

Out of the total sample of 510, 107 were male and 403 were female. 230 teachers belong to rural schools and 280 teachers belongs to urban schools. 325 secondary

school teachers were selected from CBSE schools while 185 samples were selected from state syllabus unaided schools. The final break-up of the sample is presented in Table 13.

Table 13

Final Break-up of the Sample

Total Sample 510							
G	ender	Lo	cale	Type of School CBSE State syllbus			
Male	Female	le Rural Urban CBS		CBSE	State syllbus		
107	403	230	280	325	185		

Data Collection Procedure

The data were collected from the sample drawn. As the first step investigator visited the proposed school to sought permission for data collection, data were collected from each school. The mode of responding to the tools was briefed to the principal of the school before administering. All the three tools were administered to secondary school teachers. Necessary directions were given to teachers. It was difficult to administer all the three tools at a stretch. Hence, the investigator distributed the tools to teachers. Two days are allowed them to fill their response and collected it. Sometimes it took more than one week to administer all the tools in a single school. The data collected were consolidated and codified suitably for the analysis.

Statistical Techniques used for the Study

For testing the hypotheses formulated, different statistical techniques were used. As the first step of analysis, the independent variables were classified into various levels.

Classification Techniques

Two independent variables were selected for the present study. These variables were classified into three levels each. The classification technique of each independent variable is presented in the following selection.

Classification of Technostress

The data collected using scales were classified in to three groups namely high technostress group, moderate technostress group and low technostress group. For this the mean and standard deviation of the scores obtained in the technostress scale were calculated first. teachers who fall above the mean + $\frac{1}{2}$ SD were considered as high technostress group, students who scores under mean - $\frac{1}{2}$ SD were considered as low technostress group and students coming in between mean - $\frac{1}{2}$ SD and mean + $\frac{1}{2}$ SD were categorized as moderate technostress group.

Classification of Teacher Autonomy

The data collected using scale were classified in to three groups namely scant autonomy group, moderate autonomy group and high autonomy group. For this the mean and standard deviation of the scores obtained in the technostress scale were calculated first. teachers who fall above the mean + $\frac{1}{2}$ SD were considered as high autonomy group, students who scores under mean - $\frac{1}{2}$ SD were considered as low autonomy group and students coming in between mean - $\frac{1}{2}$ SD and mean + $\frac{1}{2}$ SD were categorized as moderate autonomy group.

Statistical Techniques Used for Analysis

For analysing the data, different statistical techniques were used. The statistical techniques used in the present study can be divided into four types. The statistical techniques used in the present study is summarised as figure basic descriptive statistics such as mean, median, mode, SD, skeweness and kurtosis of each of the independent variables and dependent variables were calculated. Descriptive Statistics were calculated for total sample and distinctly for male, female, CBSE, State, rural and urban secondary school teachers. Descriptive statistics were calculated to identify the nature of the distribution of independent variables and dependent variables. Mean difference analysis As a preliminary

analysis gender difference, locality difference and difference based on type of schools were calculated for technostress, teacher autonomy and teacher burnout. Test of significance of difference between two means of large independent sample were used to compare the mean scores.

3x3 Factorial ANOVA

The main effect and interaction effect of two independent variables on the dependent variable were estimated using three-way analysis of variance. Three fixed factors were identified for each of the independent variable, technostress were classified into high, average and low category. The levels of teacher autonomy were high, moderate and slant autonomy. Hence 3x3 ANOVA, in which two independent variables at three levels, were used to analyse data. Data were analysed for total sample and separately for male, female, urban, rural, CBSE, and State Syllabus secondary school teachers. The significant F value were subjected to Scheffe's test of post hoc comparison. Multiple Regression Analysis To predict the individual and joint contribution of independent variables on the dependent variables, multiple regression analysis was used. Multiple regression was done using enter method in which all independent variables were entered simultaneously. A regression equation was also developed to predict the dependent variable from the select independent variables.

All statistical analysis has been done using SPSS for windows version 21.

A summary of the methodology used in the present study is presented in a concept map as figure 6.

Figure 6

Summary of the Methodology





ANALYSIS AND INTERPRETATION

- ✤ Preliminary Analysis
- ✤ Investigation of Group Difference
- ✤ Analysis of Variance

C

✤ Multiple Regression Analysis

ANALYSIS AND INTERPRETATION

The present study is envisioned to find out the influence of technostress and teacher autonomy on burnout among unaided secondary school teachers. for the analysis of the present data, suitable statistical techniques such as basic descriptive statistics, test of significance of difference between two means, 3x3 factorial ANOVA and multiple regression analysis were performed. The statistical analysis was carried out based on the objectives and hypotheses framed for the present study.

The entire analysis done for the present study is elucidated under the following titles.

- □ Preliminary analysis
- □ Investigation of group difference
- $\hfill\square$ Analysis of variance
- □ Multiple regression analysis

Preliminary Analysis

Preliminary analysis was conducted to find the distribution of scores of Teacher Burnout, Technostress and Teacher Autonomy.

The relevant statistical constants for the distribution of the variable Technostress and its components of total sample were calculated and presented in table 14

Statistical Constants for the Distribution of Scores of Technostress and its Components for Total Sample

Statistical Constants	Techno Insecurity	Techno Complexity	Techno Invasion	Techno Awareness	Techno Facility	Techno stress
Ν	510	510	510	510	510	510
Mean	21.29	23.28	34.87	25.25	24.92	129.60
Median	22.00	23.00	35.00	25.50	25.00	130.00
Mode	23.00	25.00	39.00	28.00	27.00	142.00
Std. Deviation	4.43	3.79	5.73	5.17	4.56	18.29
Skewness	3.38	.271	266	040	372	.052
Kurtosis	4.15	.078	.213	.066	.011	.580

Mean (129.60) and median (130) of Technostress for total sample are approximately equal. Mode (142) of Technostress for total sample is slightly deviated from mean and median. The indices of skewness (.052) and kurtosis (0.580) indicate positively skewed, leptokurtic distribution of Technostress. Mean (21.29), median (22), and mode (23) of techno insecurity for total sample are approximately equal. The indices of skewness (3.38) and kurtosis (4.15) indicate positively skewed, leptokurtic distribution of techno insecurity. Mean (23.28), median (23), and mode (25) of techno complexity for total sample are approximately equal. The indices of skewness (.271) and kurtosis (.078) indicate positively skewed, leptokurtic distribution of techno complexity. Mean (34.87), median (35), and mode (39) of techno invasion for total sample are approximately equal. The indices of skewness (-.266) and kurtosis (.213) indicate negatively skewed, leptokurtic distribution of techno invasion. Mean (25.25), median (25.50), and mode (28) of techno awareness for total sample are approximately equal. The indices of skewness (-.040) and kurtosis (.066) indicate negatively skewed, leptokurtic distribution of techno awareness. Mean (24.92), median (25), and mode (27) of techno facility for total sample are approximately equal. The indices of skewness (-.372) and kurtosis (.011) indicate negatively skewed, leptokurtic distribution of techno facility. Obtained values of mean, median, mode, skweness and kurtosis for Technostress and its components indicate that the distribution is approximately normal.

The histogram with normal plot and p-p plot of variable Technostress for total sample is presented in figure 7 and figure 8 respectively

Figure 7



Histogram with Normal Plot of Variable Technostress for Total Sample



P-P Plot of Variable Technostress for Total Sample



From indices of statistical constants, histogram with normal plots and P-P plot, it is clear that the variable Technostress is not deviated largely from normality.

The pertinent statistical constants of the distribution of the variable Teacher Autonomy and its components of total sample were calculated and presented in table 15

Table 15

Statistical Constants	Establishing School Identity	Teaching and Assessment	Parental Involvement	Professional Development	Extra Curricular Subjects	Curriculum development	Teacher Autonomy
N	510	510	510	510	510	510	510
Mean	30.14	35.66	14.37	24.52	21.96	15.62	142.26
Median	30.00	36.00	14.00	24.00	22.00	16.00	143.00
Mode	29.00	34.00	13.00	23.00	23.00	15.00	148.00
Std. Deviation	4.43	4.46	2.54	3.75	2.77	2.78	14.74
Skewness	110	138	.281	192	.038	423	.073
Kurtosis	.254	.940	.035	.444	022	1.13	.636

Statistical Constants for the Distribution of Scores of Teacher Autonomy and its Components for Total Sample

Mean (142.26) and median (143) of Teacher Autonomy for total sample are approximately equal. Mode (148) of Teacher Autonomy for total sample is slightly deviated from mean and median. The indices of skewness (.073) and kurtosis (0.636) indicate positively skewed, leptokurtic distribution of Teacher Autonomy. Mean (30.14), median (30), and mode (29) of establishing school identity for total sample are approximately equal. The indices of skewness (-.110) and kurtosis (.254) indicate negatively skewed, leptokurtic distribution of establishing school identity. Mean (35.66), median (36), and mode (34) of Teaching and assessment for total sample are approximately equal. The indices of skewness (-.138) and kurtosis (.940) indicate negatively skewed, leptokurtic distribution of Teaching and assessment. Mean (14.37), median (14), and mode (13) of parental involvement for total sample are approximately equal. The indices of skewness (.281) and kurtosis (.035) indicate positively skewed, leptokurtic distribution of parental involvement. Mean (24.52), median (24), and mode (23) of professional development for total sample are approximately equal. The indices of skewness (-.192) and kurtosis (.444) indicate negatively skewed, leptokurtic distribution of parental involvement. Mean (21.96), median (22), and mode (23) of extra curricular subjects for total sample are approximately equal. The indices of skewness (.038) and kurtosis (-.022) indicate positively skewed, platykurtic distribution of extra curricular subjects. Mean (15.62), median (16), and mode (15) of Curriculum development for total sample are approximately equal. The indices of skewness (-.423) and kurtosis (1.13) indicate negatively skewed, leptokurtic distribution of Curriculum development. Obtained values of mean, median, mode, skweness and kurtosis for Teacher Autonomy and its components indicate that the distribution is approximately normal.

The histogram with normal plot and p-p plot of variable Teacher Autonomy for total sample is presented in figure 9 and figure 10 respectively.

Figure 9





Figure 10

P-P plot of variable Teacher Autonomy for Total Sample



From indices of statistical constants, histogram with normal plots and P-P plot, it is clear that the variable Teacher Autonomy is not deviated largely from normality.

The significant statistical constants for the distribution of the variable Teacher Burnout and its components of total sample were calculated and presented in table 16

Statistical Constants	Emotional Exhaustion	Depersonalization	Reduced Personal Accomplishment	Teacher Burnout
Ν	510	510	510	510
Mean	27.65	24.60	21.32	73.56
Median	27.00	24.00	21.00	73.00
Mode	26.00	24.00	22.00	73.00
Std. Deviation	5.64	4.96	4.72	13.13
Skewness	030	.710	.327	.156
Kurtosis	.288	1.320	.803	.609

Statistical Constants for the Distribution of Scores of Teacher Burnout and its Components for Total Sample

Mean (73.56), median (73), and mode (73) of Teacher Burnout for total sample are approximately equal. The indices of skewness (.156) and kurtosis (0.609) indicate positively skewed, leptokurtic distribution of Teacher Burnout. Mean (27.65), median (27), and mode (26) of emotional exhaustion for total sample are approximately equal. The indices of skewness (-.030) and kurtosis (0.288) indicate negatively skewed, leptokurtic distribution of emotional exhaustion. Mean (24.60), median (24), and mode (24) of depersonalization for total sample are approximately equal. The indices of skewness (1.320) indicate positively skewed, leptokurtic distribution of emotional exhaustion. Mean (24.60), median (24), and mode (24) of depersonalization for total sample are approximately equal. The indices of skewness (.10) and kurtosis (1.320) indicate positively skewed, leptokurtic distribution of depersonalization. Mean (21.32), median (21), and mode (22) of reduced personal accomplishment for total sample are approximately equal. The indices of skewness (.327) and kurtosis (.803) indicate positively skewed, leptokurtic distribution of reduced personal accomplishment. Obtained values of mean, median, mode, skweness and kurtosis for Teacher Burnout and its components indicate that the distribution is approximately normal.

The histogram with normal plot and p-p plot of variable Teacher Burnout for total sample is presented in figure 11 and figure 12 respectively

Figure 11





Figure 12

P-P Plot of Variable Teacher Burnout for Total Sample



From indices of statistical constants, histogram with normal plots and P-P plot, it is clear that the variable Teacher Burnout is not deviated largely from normality.

Extent of Technostress, Teacher Autonomy and Teacher Burnout

To find the extent of Technostress, Teacher Autonomy and Teacher Burnout among unaided secondary school teachers of Kerala percentile analysis and one sample t test were used.

To know the extent of Technostress and its components percentiles of obtained data were calculated. Data and results of percentile calculation for total sample is presented in Table 17.

Table 17

Data and Results of Percentile Calculation of Technostress and its Components for Total Sample

Percentiles	Techno Insecurity	Techno Complexity	Techno Invasion	Techno Awareness	Techno Facility	Technostress
10	16.0	18.1	28.0	19.0	19.0	107.1
20	18.0	20.0	30.0	21.0	21.0	115.0
30	19.0	21.0	32.0	22.3	23.0	120.0
40	20.4	22.0	34.0	24.0	24.0	126.0
50	22.0	23.0	35.0	25.5	25.0	130.0
60	22.6	24.0	37.0	27.0	27.0	134.0
70	23.0	25.0	38.0	28.0	27.0	140.0
80	24.0	27.0	39.0	29.0	29.0	143.0
90	25.0	28.0	41.0	32.0	30.0	152.0

From the table 17, it is clear that for total sample 10th Percentile of the scores of Technostress is 107.1. That means only 10 percent of the unaided school teachers scores less than 107.1 on the Technostress. 50th Percentile of the scores of Technostress is 130. That means 50 percent of the unaided school teachers scores less than 130 and 50 percent scores higher than 130. 50th Percentile of the scores of techno insecurity is 22. That means 50 percent of the unaided school teachers scores less than 22 and 50 percent scores higher than 22. 50th Percentile of the scores of techno complexity is 23. That means 50 percent of the unaided school teachers scores less than 23 and 50 percent scores higher than 23. 50th Percentile of the scores of techno invasion is 35. That means 50 percent of the unaided school teachers scores less than 35 and 50 percent scores higher than 35. 50th Percentile of the scores of techno awareness is 25.5. That means 50 percent of the unaided school teachers scores less than 25.5 and 50 percent scores higher than 25.5. 50th Percentile of the scores of techno facility is 25. That means 50 percent of the unaided school teachers scores less than 25.5 and 50 percent scores higher than 25.5. 50th Percentile of the scores scores less than 25.5 and 50 percent scores higher than 25.5.

To assess the extent of Technostress among unaided school teachers one sample t test was also used. Mid score of the tool was used as the test value for the calculation t value. The maximum score obtainable for Technostress scale is 205 and minimum score obtainable is 41. Mid score of the tool is 123, and it is taken as the test value in one sample t test. Similarly, mid score of each component were calculated and used as test value. Results of the one sample t test conducted for Technostress and its components are presented in table 18

Table 18

Variable	Ν	Mean	Std. Deviation	Test value	t-value
Techno Insecurity	510	21.29	4.43	18	16.77**
Techno Complexity	510	23.28	3.79	21	13.57**
Techno Invasion	510	34.87	5.73	30	19.17**
Techno Awareness	510	25.25	5.17	27	7.65**
Techno Facility	510	24.92	4.56	27	10.13**
Technostress	510	129.60	18.29	123	8.15**

One Sample t test for the Variable Technostress and its Components

**Significant at 0.01 level

Table 18 reveals that there exists a significant difference in the mean Technostress score (129.60) and mid score (123) of unaided school teachers (t= 8.15, p<0.01). Mean Technostress score is higher than the mid score. It means that Technostress scores of the unaided school teachers are significantly higher than the mid score. From the result it can be concluded that the unaided school teachers possess above average level of Technostress.

Components wise analysis shows that there exists significant difference in the mean scores of components of Technostress such as techno insecurity (t=16.77, p<0.01), techno complexity (t=13.57, p<0.01), techno invasion (t=19.17, p<0.01), techno awareness (t=7.65, p<0.01) and techno facility (t=10.13, p<0.01) and corresponding mid values of unaided school teachers. Mean scores of techno
insecurity, techno complexity and techno invasion are higher than corresponding mid values. It means that mean scores of techno insecurity, techno complexity and techno invasion are significantly higher than mid values. Mean scores of techno awareness and techno facility are lower than corresponding mid values. It means that mean scores of techno awareness and techno facility are significantly lower than mid values. From results it can be concluded that unaided school teachers possess above average level of techno insecurity, techno complexity and techno invasion and below average level of techno awareness and techno facility.

To know the extent of Teacher Autonomy and its components percentiles of obtained data were calculated. Data and results of percentile calculation for total sample is presented in table 19.

Table 19

Percentiles	Establishing School Identity	Teaching and assessment	Parental Involvement	Professional Development	Extra Curricular Subjects	Curriculum development	Teacher Autonomy
10	25.00	30.00	11.00	20.00	18.00	12.00	123.00
20	26.00	32.00	12.00	22.00	20.00	14.00	130.00
30	28.00	34.00	13.00	23.00	21.00	15.00	135.00
40	29.00	35.00	14.00	23.00	21.00	15.00	140.00
50	30.00	36.00	14.00	24.00	22.00	16.00	143.00
60	31.00	37.00	15.00	25.00	23.00	16.00	147.00
70	32.70	38.00	16.00	26.00	23.00	17.00	150.00
80	34.00	39.00	16.00	27.00	24.00	18.00	154.00
90	35.00	41.00	18.00	30.00	26.00	19.00	159.00

Data and Results of Percentile Calculation of Teacher Autonomy and its Components for Total Sample

From the table it is clear that for total sample 10th Percentile of the scores of Teacher Autonomy is 123.00. That means only 10 percent of the unaided school teachers

128 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

scores less than 123.00 on the Teacher Autonomy. 50th Percentile of the scores of Teacher Autonomy is 143.00. That means 50 percent of the unaided school teachers scores less than 143 and 50 percent scores higher than 143. 50th Percentile of the scores of establishing school identity is 30. That means 50 percent of the unaided school teachers scores less than 30 and 50 percent scores higher than 30. 50th Percentile of the scores of Teaching and assessment is 36. That means 50 percent of the unaided school teachers scores less than 36 and 50 percent scores higher than 36. 50th Percentile of the scores of parental involvement is 14. That means 50 percent of the unaided school teachers scores less than 14 and 50 percent scores higher than 14. 50th Percentile of the scores of professional development is 24. That means 50 percent of the unaided school teachers scores less than 24 and 50 percent scores higher than 24. 50th Percentile of the scores of extra-curricular subjects is 22. That means 50 percent of the unaided school teachers scores less than 22 and 50 percent scores higher than 22. 50th Percentile of the scores of Curriculum development is 16. That means 50 percent of the unaided school teachers scores less than 16 and 50 percent scores higher than 16.

To assess the extent of Teacher Autonomy among unaided school teachers one sample t test was also used. Mid score of the tool was used as the test value for the calculation t value. The maximum score obtainable for Teacher Autonomy is 215 and minimum score obtainable is 43. Mid score of the tool is 126, and it is taken as the test value in one sample t test. Similarly, mid score of each component were calculated and used as test value. Results of the one sample t test conducted for Teacher Autonomy and its components are presented in table 20

One Sample t test for the Variable Teacher Autonomy and its Components

Variables	Ν	Mean	Std. Deviation	Test value	t
Establishing School Identity	510	30.14	4.43	27	16.01**
Teaching and assessment	510	35.66	4.46	30	28.62**
Parental Involvement	510	14.37	2.54	15	5.58**
Professional Development	510	24.52	3.75	21	21.17**
Extra-Curricular Subjects	510	21.96	2.77	18	32.33**
Curriculum development	510	15.62	2.78	15	4.99**
Teacher Autonomy	510	142.26	14.74	126	195.04**

**significant at 0.01 level

Table 20 conveys there exists a significant difference in the mean Teacher Autonomy score (142.26) and mid score (126) of unaided school teachers (t= 195.04, p<0.01). Mean Teacher Autonomy score is higher than the mid score. It means that Teacher Autonomy scores of the unaided school teachers are significantly higher than the mid score. From the result it can be concluded that the unaided school teachers possess above average level of Teacher Autonomy.

Components wise analysis shows that there exists significant difference in the mean scores of components of Teacher Autonomy such as establishing school identity (t=16.01, p<0.01), Teaching and assessment evaluation (t=28.62, p<0.01), parental involvement (t=5.38, p<0.01), professional development (t=21.17, p<0.01) extra-curricular subjects (t=32.33, p<0.01) and curriculum development (t=4.99, p<0.01) and corresponding mid values of unaided school teachers. Mean scores of establishing school identity, Teaching and assessment evaluation, professional development, extra-curricular subjects and Curriculum development are higher than corresponding mid values. It means that mean scores establishing school identity, Teaching and assessment evaluation, professional development, extra-curricular subjects and Curriculum development, extra-curricular

130 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

mid values. Mean score of parental involvement is lower than corresponding mid value. It means that mean scores of parental involvement is significantly lower than mid value. From results it can be concluded that unaided school teachers possess above average level of establishing school identity, Teaching and assessment evaluation, professional development, extra-curricular subjects and Curriculum development and below average level of parental involvement.

To know the extent of Teacher Burnout and its components percentiles of obtained data were calculated. Data and results of percentile calculation for total sample is presented in table 21.

Table 21

Data and Results of Percentile Calculation of Teacher Burnout and its Components for Total Sample

Percentiles	Teacher Burnout	Emotional Exhaustion	Depersonalization	Reduced personal Accomplishment
10	57.0	21.0	18.1	15.0
20	63.0	23.0	20.0	18.0
30	67.0	25.0	22.0	19.0
40	70.0	26.0	23.0	20.0
50	73.0	27.0	24.0	21.0
60	76.0	29.0	25.0	22.0
70	80.0	30.0	26.0	23.0
80	84.0	32.0	28.0	25.0
90	90.0	35.0	31.0	27.0

From the table 21 it is clear that for total sample 10th Percentile of the scores of Teacher Burnout is 57.0 that means only 10 percent of the unaided school teachers scores less than 57 on the Teacher Burnout. 50th Percentile of the scores of Teacher Burnout is 73. That means 50 percent of the unaided school teachers scores less than 73 and 50 percent scores higher than 73. 50th Percentile of the score of emotional

exhaustion is 27. That means 50 percent of the unaided school teachers scores less than 27 and 50 percent scores higher than 27. 50th Percentile of the scores of depersonalization is 24. That means 50 percent of the unaided school teachers scores less than 24 and 50 percent scores higher than 24. 50th Percentile of the score of reduced personal accomplishment is 21. That means 50 percent of the unaided school teachers scores less than 21 and 50 percent scores higher than 21.

To assess the extent of Teacher Burnout among unaided school teachers one sample t test was also used. Mid score of the tool was used as the test value for the calculation t value. The maximum score obtainable for Teacher Burnout inventory is 140 and minimum score obtainable is 28. Mid score of the tool is 84, and it is taken as the test value in one sample t test. Similarly, mid score of each component were calculated and used as test value. Results of the one sample t test conducted for Teacher Burnout and its components are presented in table 22.

Table 22

0	ne S	Sampl	le t	test	for	the	V_{i}	aria	ble	1	Feache	r İ	Burnout	and	its	C	compo	onei	its
---	------	-------	------	------	-----	-----	---------	------	-----	---	--------	-----	---------	-----	-----	---	-------	------	-----

Variables	N	Mean	Std. Deviation	Test value	t-value
Emotional Exhaustion	510	27.65	5.64	30	9.41**
Depersonalization	510	24.60	4.96	30	24.59**
Reduced personal accomplishment	510	21.32	4.72	24	12.82**
Teacher Burnout	510	73.56	13.13	84	17.94**

**Significant at 0.01 level

Table 22 discloses that there exists a significant difference in the mean Teacher Burnout score (73.56) and mid score (84) of unaided school teachers (t= 17.94, p<0.01). Mean Teacher Burnout score is lower than the mid score. It means that Teacher Burnout scores of the unaided school teachers are significantly lower than the mid score. From the result it can be concluded that the unaided school teachers possess below average level of Teacher Burnout.

132 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

Components wise analysis shows that there exists significant difference in the mean scores of components of Teacher Burnout such as emotional exhaustion (t=9.41, p<0.01), depersonalization (t=24.59, p<0.01), and reduced personal accomplishment (t=12.82, p<0.01), and corresponding mid values of unaided school teachers. Mean scores of emotional exhaustions. depersonalization, and reduced personal accomplishment are lower than corresponding mid values. It means that mean scores emotional exhaustion, depersonalization, and reduced personal accomplishment are significantly lower than mid values. From results it can be concluded that unaided school teachers possess below average level of emotional exhaustion, depersonalization, and reduced personal accomplishment.

Mean Difference Analysis

Comparison of technostress, teacher autonomy and teacher burnout of unaided school teachers based on Gender, Type of school, Experience, Locality, subject, computer course and educational qualification were conducted and presented under relevant headings. Detailed discussions of comparison are presented below.

Investigation of Mean Difference based on Gender

The test of significance difference between mean scores of male and female aided school teachers for the variables technostress, teacher autonomy and teacher burnout were calculated to find gender difference.

The details of results of mean difference analysis of technostress and its components are presented in table 23

Variables Gender Ν Std. Deviation t-value Mean 107 22.35 Male 3.73 2.80** Techno Insecurity 403 21.01 4.56 Female Male 107 24.31 3.65 3.19** Techno Complexity 3.79 Female 403 23.01 107 35.86 5.91 Male Techno Invasion 2.02* Female 403 34.60 5.66 Male 107 25.81 5.80 Techno Awareness 1.27 Female 403 25.10 4.98 107 24.73 4.98 Male **Techno Facility** .481 Female 403 24.97 4.45 Male 107 133.06 20.23 Technostress 2.21* Female 403 128.68 17.65

Results of Mean Difference Analysis of Technostress and its Components

**Significant at 0.01 level, *Significant at 0.05 level

Table 23 shows that, mean scores of technostress of the unaided school male teachers and female teachers differ significantly [t = 2.21; p<.05]. Mean score showed that male teachers (M = 133.06) have higher technostress than female teachers (M = 128.68).

Table explains that, mean scores of techno insecurity [t = 2.80; p < .01], techno complexity [t = 3.19; p<.01] and techno invasion [t = 2.02; p<.05] of the unaided school male teachers and female teachers differ significantly. Mean score showed that male teachers have higher techno insecurity (M = 22.35), techno complexity (M =24.31) and techno invasion (M = 35.86) than female teachers (M = 21.01).

Mean scores of techno awareness [t = 1.27; p>05] and techno facility [t =.481; p>05] of the unaided school male teachers and female teachers do not differ significantly.

The details of results of mean difference analysis of Teacher autonomy and its components are presented in table 24.

Variables	Gender	Ν	Mean	Std. Deviation	t-value
Establishing Sahaal Identity	Male	107	29.08	4.95	2 55*
Establishing School Identity	Female	403	30.42	4.24	2.33
Teaching and accordment	Male	107	35.62	5.31	001
reaching and assessment	Female	403	35.67	4.22	.091
Derentel Involvement	Male	107	14.26	2.64	508
	Female	403	14.40	2.51	.308
Profossional Davalonment	Male	107	24.59	4.35	202
Professional Development	Female	403	24.50	3.58	.205
Extra Curricular Subjects	Male	107	21.45	3.13	1 07*
Extra-Cumcular Subjects	Female	403	22.10	2.65	1.97
Curriculum development	Male	107	15.92	2.62	1 26
Curriculum development	Female	403	15.54	2.82	1.20
Tanahar Autonomy	Male	107	140.92	17.07	051
reacher Autonomy	Female	403	142.62	14.05	.931

Results of Mean Difference Analysis of Teacher Autonomy and its Components

*Significant at 0.05 level

Table 24 reveals that, mean scores of Teacher autonomy of the unaided school male teachers and female teachers do not differ significantly [t = .951; p > .05].

Table explains that, mean scores of Establishing School Identity [t = 2.55; p<.05] and extra-curricular subjects [t = 1.97; p<.05] of the unaided school male teachers and female teachers differ significantly. Mean score of Establishing School Identity (M = 30.42) and extra-curricular subjects (M = 22.10) indicated that female teachers have higher mean scores than male teachers.

Mean scores of Teaching and assessment [t =.091; p>05], parental involvement [t =.508; p>.05] and, Curriculum development [t = 1.26; p>05] of the unaided school male teachers and female teachers do not differ significantly.

The details of results of mean difference analysis of Teacher burnout and its components are presented in table 25

Variable	Gender	Ν	Mean	Std. Deviation	t-value	
Emotional Exhaustion	Male	107	29.64	6.28	2 70**	
Emotional Exhaustion	Female	403	27.12	5.35	5.79	
Depersonalization	Male	107	25.71	5.04	2 (2**	
	Female	403	24.30	4.90	2.63	
Reduced personal	Male	107	23.33	5.47	1 1 1 **	
accomplishment	Female	403	20.79	4.35	4.44	
	Male	107	78.67	14.40	4 2 4**	
reacher Burnout	Female	403	72.21	12.45	4.24	

Results of Mean Difference Analysis of Teacher Burnout and its Components

**Significant at 0.05 level

Table 25 conveys that, mean scores of Teacher burnout of the unaided school male teachers and female teachers differ significantly [t = 4.24; p<.05]. Mean score showed that male teachers (M = 78.07) have higher mean scores than female teachers (M = 72.21).

Table describes that, mean scores of emotional exhaustion [t = 3.79; p<.05], depersonalization [t = 2.63; p<.05] and reduced personal accomplishment [t = 4.44; p<.05] of the unaided school male teachers and female teachers differ significantly. Mean scores indicated that male teachers have higher emotional exhaustion (Male = 29.64; Female= 27.12), depersonalization (Male= 25.71; Female = 24.30) and reduced personal accomplishment (Male = 23.33; Female = 20.79) than female teachers.

Investigation of mean difference based on Locale

The test of significance difference between mean scores of urban and rural unaided school teachers for the variables technostress, teacher autonomy and teacher burnout were calculated to find locale difference

136 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

The details of results of mean difference analysis of technostress and its components are presented in table 26.

Table 26

Results o	f Mean	Difference	Analysis of	^c Technostress	and its Co	omponents
	/	././	~ ./			

Variables	Locality	Ν	Mean	Std. Deviation	t-value	
Tashna Ingaanity	Rural	228	21.40	3.61	500	
reclino insecurity	Urban	282	21.20	5.00	.528	
Tashna Camplavity	Rural	228	23.38	3.87	550	
Techno Complexity	Urban	282	23.20	3.74	.332	
Techno Invasion	Rural	228	34.88	5.39	027	
	Urban	282	34.86	6.00	.037	
T 1	Rural	228	25.42	5.32	(7(
Techno Awareness	Urban	282	25.11	5.04	.0/0	
Tashna Fasility	Rural	228	25.36	4.41	1.05	
Techno Facility	Urban	282	24.56	4.66	1.95	
Technostress	Rural	228	130.44	17.54	021	
	Urban	282	128.92	18.88	.931	

Table 26 explains that, mean scores of technostress of the rural and urban unaided secondary school teachers do not differ significantly [t = .931, p > .05].

Table reveals that, mean scores of techno insecurity [t =.528; p>.05], techno complexity [t =.552; p>.05], techno invasion [t =.037 p>.05]techno awareness [t =.676; p>05] and techno facility [t =1.95; p<05] of rural and urban unaided secondary school teachers do not differ significantly.

The details of results of mean difference analysis of Teacher autonomy and its components are presented in table 27

Variables	Locality	N	Mean	Std. Deviation	t-value	
Establishing	Rural	228	30.60	4.53	2.12*	
School Identity	Urban	282	29.77	4.32	2.12	
Teaching and assessment	Rural	228	36.02	4.30	1 (4	
	Urban	282	35.37	4.58	1.04	
Parental Involvement	Rural	228	14.31	2.62	524	
	Urban	282	14.43	2.47	.324	
Professional	Rural	228	25.06	3.64	2.07**	
Development	Urban	282	24.07	3.79	2.97	
Extra Curricular	Rural	228	22.38	2.84	2 10**	
Subjects	Urban	282	21.62	2.67	5.10	
Curriculum	Rural	228	15.89	2.77	2.04*	
development	Urban	282	15.39	2.78	2.04	
Teacher	Rural	228	144.26	14.86	> 77 ^{**}	
Autonomy	Urban	282	140.65	14.46	2.77	

Results of Mean Difference Analysis of Teacher Autonomy and its Components

**Significant at 0.01 level, *Significant at 0.05 level

Table 27 reveals that, mean scores of Teacher autonomy of the rural and urban unaided secondary school teachers differ significantly [t = 2.77; p<.01]. Mean score revealed that rural teachers (M = 144.26) have higher mean scores than urban teachers (M = 140.65).

Table elucidates that, the mean scores of Establishing School Identity [t = 2.12; p < .05] professional development [t = 2.97; p < .05], extra-curricular subjects [t = 3.10; p < .01] and Curriculum development [t = 2.04; p < 01] of the rural and urban unaided secondary school teachers differ significantly. Mean score revealed that rural teachers have higher autonomy of establishing school identity (rural = 30.60; urban = 29.77), professional development (rural = 25.06; urban = 24.07), extra-curricular subjects (rural = 22.38; urban = 21.62) and Curriculum development (M = 15.89; M = 15.39) than urban teachers.

Mean scores of Teaching and assessment [t= 1.64; p>05] and parental involvement [t=.524; p>.05] of the rural and urban unaided secondary school teachers do not differ significantly.

The details of results of mean difference analysis of Teacher burnout and its components are presented in table 28.

Table 28

Variables	Locality	Ν	Mean	Std. Deviation	t-value
Emotional Exhaustion	Rural	228	26.79	5.09	2 10**
Emotional Exhaustion	Urban	282	28.34	5.97	3.18
Demonstration	Rural	228	24.52	4.85	201
Depersonalization	Urban	282	24.66	5.05	.321
Reduced personal	Rural	228	20.68	4.47	1 70**
accomplishment	Urban	282	21.84	4.85	2.78
	Rural	228	71.98	12.21	2 46*
Teacher Burnout	Urban	282	74.84	13.72	2.40

Results of Mean Difference Analysis of Teacher Burnout and its Components

**Significant at 0.01 level, *Significant at 0.05 level

Table 28 reveals that, critical ratio gained for rural and urban teachers for burnout is 2.46 which is significant at 0.05 level. So the level of burnout of unaided secondary school teachers differ significantly with respect to locale [t = 2.46; p<.05]. Mean score exhibited that unaided school teachers in urban area (M = 74.84) have higher level of burnout than unaided school teachers in rural area (M = 71.98).

Table explicates that, emotional exhaustion of the rural and urban unaided secondary school teachers differ significantly [t = 3.18; p<.05]. Mean score revealed that urban teachers (M =28.34) have higher mean scores than rural teachers (M = 26.79). Mean scores of depersonalization of the rural and urban unaided secondary school teachers do not differ significantly [t = .321; p>.05]. Mean scores of reduced personal accomplishment of the rural and urban unaided secondary school teachers

differ significantly [t =2.78; p<.01]. Mean score exposed that urban teachers (M =21.84) have higher mean scores than rural teachers (M = 20.68).

Investigation of Mean difference based on Type of School

The test of significance difference between mean scores of CBSE and unaided State syllabus school teachers for the variables technostress, teacher autonomy and teacher burnout were calculated to find school difference.

The details of results of mean difference analysis of technostress and its components are presented in table 29.

Table 29

Results of Mean Difference Analysis of Technostress and its Components

Variables	Type of School	Ν	Mean	Std. Deviation	t-value	
Taskes Ingestrity	CBSE	325	21.22	4.98	504	
Techno Insecurity	State syllabus	185	21.41	3.25	.304	
Techno Complexity	CBSE	325	23.22	3.68	172	
	State syllabus	185	23.38	3.98	.475	
Techno Invasion	CBSE	325	34.97	6.02	551	
	State syllabus	185	34.68	5.20	.551	
Tachno Awaranass	CBSE	325	24.98	5.35	1 5 5	
Techno Awareness	State syllabus	185	25.72	4.81	1.55	
Tachna Facility	CBSE	325	24.51	4.74	2 60**	
Techno Facility	State syllabus	185	25.63	4.15	2.09	
Technostress	CBSE	325	128.90	19.16	1 1/	
1001110501055	State syllabus	185	130.82	16.63	1.14	

**Significant at 0.05 level

Table 29 conveys that, mean scores of technostress of the CBSE and state syllabus teachers do not differ significantly [t = 1.14; p>.05].

Table explains that, mean scores of techno insecurity [t =.504; p>.05], techno complexity [t =.473; p>.05], techno invasion [t =.551; p>.05] and techno awareness [t = 1.55; p>05] of the CBSE and state syllabus unaided teachers do not differ

significantly. Mean scores of techno facility of the CBSE and state syllabus teachers differ significantly [t =2.69; p<05]. Mean score revealed that state syllabus teachers (M = 25.63) have higher techno facility stress than CBSE teachers (M = 24.51).

The details of results of mean difference analysis of Teacher autonomy and its components are presented in table 30

Table 30

Variables	Type of School	Ν	Mean	Std. Deviation	t-value
Establishing School	CBSE	325	30.08	4.42	421
Identity	State syllabus	185	30.25	4.46	.421
Teaching and	CBSE	325	35.54	4.55	014
assessment	State syllabus	185	35.87	4.31	.814
Parental Involvement	CBSE	325	14.54	2.51	1.02
	State syllabus	185	14.09	2.55	1.95
Professional	CBSE	325	24.27	3.89	1.93
Development	State syllabus	185	24.94	3.47	
Extra Curricular	CBSE	325	22.02	2.76	625
Subjects	State syllabus	185	21.86	2.79	.033
Curriculum	CBSE	325	15.56	2.75	565
development	State syllabus	185	15.71	2.85	.303
Tanahar Autonomy	CBSE	325	142.01	15.26	521
	State syllabus	185	142.71	13.79	.321

Results of Mean Difference Analysis of Teacher Autonomy and its Components

Table 30 conveys that, mean scores of Teacher autonomy of the unaided CBSE and state syllabus unaided teachers do not differ significantly [t = .521; p > .05].

Table explains that, mean scores of Establishing School Identity [t =.421; p>.05], Teaching and assessment [t =.814; p>05], parental involvement [t = 1.93; p>.05], professional development [t = 1.93; p>.05], extra-curricular subjects [t =.635; p>.05] and Curriculum development [t =.565; p>05] of the CBSE and state syllabus unaided teachers do not differ significantly. The details of results of mean difference analysis of Teacher burnout and its components are presented in table 31

Variables	Type of School	N	Mean	Std. Deviation	t-value	
Emotional	CBSE	325	27.66	5.87	002	
Exhaustion	State syllabus	185	27.62	5.25	.095	
Depersonalization	CBSE	325	24.39	4.71	1.250	
	State syllabus	185	24.96	5.36	-1.239	
Reduced personal	CBSE	325	21.54	4.78	1 277	
accomplishment	State syllabus	185	20.94	4.60	1.377	
Teacher Burnout	CBSE	325	73.59	13.21	050	
	State syllabus	185	73.52	13.04	.059	

Results of Mean Difference Analysis of Teacher Burnout and its Components

Table 31 reveals that, mean scores of Teacher burnout of the CBSE and unaided State syllabus teachers do not differ significantly [t = .059; p > .05].

Table describes that, mean scores of depersonalization of the CBSE and unaided State syllabus teachers differ significantly [t = 1.259; p > .05]. Mean scores of emotional exhaustion [t = 0.93; p > .05], reduced personal accomplishment [t = 0.59; p > .05] of the CBSE and state syllabus unaided teachers do not differ significantly.

Investigation on Mean Difference based on Qualification

The test of significance difference between mean scores of highly qualified unaided school teachers and unaided teachers with basic qualification for the variables technostress, teacher autonomy and teacher burnout were calculated to find qualification difference.

The details of results of mean difference analysis of technostress and its components are presented in table 32.

Variables	Educational Qualification	N	Mean	Std. Deviation	t-value
Techno	Basic Qualification	295	21.33	3.54	.262
Insecurity	Highly qualified	215	21.23	5.42	
Techno	Basic Qualification	295	23.56	3.72	1.98^{*}
Complexity	Higher Qualification	215	22.89	3.86	
Techno	Basic Qualification	295	34.97	5.81	.474
Invasion	Higher Qualification	215	34.73	5.63	
Techno	Basic Qualification	295	25.68	5.26	2.20^{*}
Awareness	Higher Qualification	215	24.66	4.99	
Techno	Basic Qualification	295	25.36	4.39	2.58**
Facility	Higher Qualification	215	24.31	4.74	
Technostress	Basic Qualification	295	130.90	17.91	1.88
	Higher Qualification	215	127.81	18.69	

Results of Mean Difference Analysis of Technostress and its Components

**Significant at 0.01 level, *Significant at 0.05 level

Table 32 conveys that, mean scores of technostress of unaided secondary school teachers with higher qualification and teachers with basic qualification do no differ significantly [t = 1.88; p>.05].

Table explains that, mean scores of techno complexity [t =1.98; p<.01], techno awareness [t = 2.20; p<05] and techno facility [t =2.58; p<05] of unaided secondary school teachers with higher qualification and teachers with basic qualification differ significantly. Mean score revealed that teachers with basic qualification have higher techno complexity (M =23.56), techno awareness (M =25.68) and techno facility (M = 24.31) than teachers with higher qualification. Mean scores of techno insecurity [t =. 262; p>.05] and techno invasion [t =.474 p>.05] of unaided secondary school teachers with higher qualification and teachers with basic qualification do no differ significantly.

The details of results of mean difference analysis of Teacher autonomy and its components are presented in table 33.

Variables	Educational Qualification	N	Mean	Std. Deviation	t-value	
Establishing	Basic Qualification	295	30.34	4.53	1 10	
School Identity	Higher Qualification	215	29.87	4.29	1.19	
Teaching and	Basic Qualification	295	35.96	4.52	1.02	
assessment	Higher Qualification	215	35.24	4.36	1.82	
Parental	Basic Qualification	295	14.11	2.60	7 01**	
Involvement	Higher Qualification	215	14.74	2.41	2.81	
Professional	Basic Qualification	295	24.81	3.62	2 0.0*	
Development	Higher Qualification	215	24.11	3.89	2.08	
Extra Curricular	Basic Qualification	295	22.01	2.81	420	
Subjects	Higher Qualification	215	21.90	2.71	.420	
Curriculum	Basic Qualification	295	15.90	2.69	2 72**	
development	Higher Qualification	215	15.22	2.86	2.75	
Teacher	Basic Qualification	295	143.13	14.72	1.55	
Autonomy	Higher Qualification	215	141.08	14.71	1.55	

Results of Mean Difference Analysis of Teacher Autonomy and its Components

**Significant at 0.01 level, *Significant at 0.05 level

Table 33 discloses that, mean scores of Teacher autonomy of unaided secondary school teachers with higher qualification and teachers with basic qualification do no differ significantly [t = 1.55; p>.01].

Table elucidates that, mean scores of parental involvement [t =2.81; p<.01], professional development [t = 2. 08; p<.05] and Curriculum development [t =2.73; p<01] of unaided secondary school teachers with higher qualification and teachers with basic qualification differ significantly. Mean score revealed that teachers with basic qualification have higher professional development (M = 24.81), and Curriculum development (M =15.90) than teachers with higher qualification. Mean score revealed that teachers with higher qualification have higher parental involvement((M =14.741) than teachers with basic qualification.

Mean scores of Establishing School Identity [t = 1.19; p > .05], Teaching and assessment [t = 1.82; p > 05] and extra-curricular subjects [t = .420; p > .05]of unaided secondary school teachers with higher qualification and teachers with basic qualification do not differ significantly.

The details of results of mean difference analysis of Teacher burnout and its components are presented in table 34.

Table 34

Variables	Educational Qualification	N	Mean	Std. Deviation	t- value
Emotional	Basic Qualification	295	27.43	5.61	000
Exhaustion	Higher Qualification	215	27.94	5.69	.999
Depersonalization	Basic Qualification	295	24.51	5.12	105
	Higher Qualification	215	24.72	4.74	.403
Reduced personal	Basic Qualification	295	21.22	4.89	510
accomplishment	Higher Qualification	215	21.46	4.48	.348
Teacher Burnout	Basic Qualification	295	73.16	13.45	800
	Higher Qualification	215	74.12	12.70	.009

Results of Mean Difference Analysis of Teacher Burnout and its Components

Table 34 conveys that, mean scores of burnout of unaided secondary school teachers with higher qualification and teachers with basic qualification do not differ significantly [t = .809; p > .05].

Table explicates that, mean scores of emotional exhaustion [t = .999; p > .05], depersonalization [t = .485; p > .05] and reduced personal accomplishment [t = .548; p > .05] of unaided secondary school teachers with higher qualification and teachers with basic qualification do not differ significantly.

Investigation of Mean Difference based on Subject

To find whether the subject of unaided school teachers (language, social science, science and mathematics) can significantly affect the technostress, teacher autonomy, teacher burnout and its components, one-way ANOVA was employed.

Results of the one-way ANOVA of technostress and its components are given in table 35

Table 35

Results of the One-way ANOVA of Technostress and its Components

Variables	Source of variance	Sum of Squares	Df	Mean Square	F	Sig.
	Between Groups	53.321	3	17.774	000	429
Techno	Within Groups	9921.309	506	19.607	.900	.438
mseeditty	Total	9974.629	509			
TT 1	Between Groups	77.276	3	25.759	1 800	146
Techno Complexity	Within Groups	7241.187	506	14.311	1.800	.140
complexity	Total	7318.463	509			
TT 1	Between Groups	152.773	3	50.924	1 5 5 5	100
Techno	Within Groups	16570.160 506 32.7		32.747	1.555	.199
mvasion	Total	16722.933	509			
TT 1	Between Groups	102.327	3	34.109	1 270	201
l echno A wareness	Within Groups	13489.048	506	26.658	1.279	.201
1 Wureness	Total	13591.375	509			
T 1	Between Groups	161.294	3	53.765	2 600	051
Techno Facility	Within Groups	10429.248	506	20.611	2.009	.031
1 definty	Total	10590.541	509			
	Between Groups	1835.991	3	611.997	1 820	120
Technostress	Within Groups	168408.409	506	332.823	1.039	.139
	Total	170244.400	509			

Table 35 conveys that there is no significant effect of subject on technostress of unaided school teachers (F (3, 506) =.139, p>.05). There is no significant effect of subject on techno insecurity of unaided school teachers (F (3, 506) =.438, p>.05). There is no significant effect of subject on techno complexity of unaided school teachers (F (3, 506) =.146, p>.05). There is no significant effect of subject on techno invasion of unaided school teachers (F (3, 506) =.146, p>.05). There is no significant effect of subject on techno invasion of unaided school teachers (F (3, 506) =.199, p>.05). There is no significant effect of subject on techno techno invasion of unaided school teachers (F (3, 506) =.199, p>.05). There is no significant effect of subject on techno awareness of unaided school teachers (F (3, 506) =.281, p>.05). There is no significant effect of subject on techno facility of unaided school teachers (F (3, 506) =.051, p>.05).

$146 \ \mathrm{Influence} \ \mathrm{of} \ \mathrm{technostress} \ \mathrm{\&} \ \mathrm{teacher} \ \mathrm{autonomy} \ \mathrm{on} \ \mathrm{burnout}$

Results of the one-way ANOVA of teacher autonomy and its components are given in table 36.

Table 36

Results of the One-way ANOVA	of Teacher Autonomy	and its Components
------------------------------	---------------------	--------------------

Variables	Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
Establishing	Between Groups	72.032	3	24.011	1 226	200
School	Within Groups	9913.083	506 19.591		1.220	.300
Identity	Total	9985.116	509			
	Between Groups	72.846	3	24.282	1 220	202
Teaching and	Within Groups	10070.104	506	19.901	1.220	.302
VariablesEstablishingSchoolIdentityTeaching andassessmentParentalInvolvementProfessionalDevelopmentExtraCurricularSubjectsCurriculumDevelopment	Total	10142.951	509			
	Between Groups	16.666	3	5.555	967	460
Parental Involvement	Within Groups	3256.550	506	6.436	.805	.400
Involvement	Total	3273.216	509			
	Between Groups	90.732	3	30.244	2 1 6 5	001
Professional	Within Groups	7068.642	506	13.970	2.103	.091
VariablesEstablishing School IdentityTeaching and assessmentParental 	Total	7159.375	509			
Extra	Between Groups	35.351	3	11.784	1 5 4 2	202
Parental Involvement Professional Development Extra Curricular Subjects Curriculum Development	Within Groups	3864.942	506	7.638	1.343	.205
Subjects	Total	3900.292	509			
I	Between Groups	47.262	3	15.754	2.046	106
Curriculum	Within Groups	3895.413	506	7.698	2.040	.100
Establishing School Identity Teaching and assessment Parental Involvement Professional Development Extra Curricular Subjects Curriculum Development Teacher Autonomy	Total	3942.675	509			
	Between Groups	392.317	3	130.772	601	615
Teacher	Within Groups	110126.475	506	217.641	.001	.013
¹ Satononity	Total	110518.792	509			

Table 36 reveals that there is no significant effect of subject on teacher autonomy of unaided school teachers (F (3, 506) = .615, p > .05).

Table describes that there is no significant effect of subject on component establishing school identity of unaided school teachers (F (3, 506) = .300, p > .05). There is no significant effect of subject on component Teaching and assessment of unaided school teachers (F (3, 506) = .302, p > .05). There is no significant effect of subject on

component parental involvement of unaided school teachers (F (3, 506) = 0.460, p>.05). There is no significant effect of subject on component professional development of unaided school teachers (F (3, 506) =.091, p>.05). There is no significant effect of subject on component extracurricular subjects of unaided school teachers (F(3, 506) =.203, p>.05). There is no significant effect of subject on component Curriculum development of unaided school teachers (F (3, 506) = 0.106, p>.05).

Results of the one-way ANOVA of teacher burnout and its components are given in table 37.

Table 37

Results o	of the	One-way	<i>v ANOVA</i>	of '	Teacher	Burnout	and i	ts Cor	nponents
		~		~					1

Variables	Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
	Between Groups	72.394	3	24.131	756	510
Emotional	Within Groups	16144.077	506	31.905	.750	.319
Exilustion	Total	16216.471	509			
Depersonalization	Between Groups	47.975	3	15.992	640	581
	Within Groups	12458.817	506	24.622	.049	.364
	Total	12506.792	509			
	Between Groups	142.216	3	47.405	2 1 4 4	004
Accomplishment	Within Groups	11187.047	506	22.109	2.144	.094
Variables Emotional Exhaustion Depersonalizatio Reduced Persona Accomplishment Teacher Burnout	Total	11329.263	509			
	Between Groups	717.148	3	239.049	1 2 2 0	245
Teacher Burnout	Within Groups	87094.217	506	172.123	1.369	.243
	Total	87811.365	509			

Table 37 reveals that there is no significant effect of subject on teacher burnout of unaided school teachers (F (3, 506) = .245, p > .01).

Table explains that there is a significant effect of subject on emotional exhaustion of unaided school teachers (F (3, 506) =.519, p>.01). There is no significant effect of subject on component depersonalization of unaided school teachers (F (3, 506) =.584, p>.05). There is a significant effect of subject on component reduced personal accomplishment of unaided school teachers (F (3, 506)

=.094, p<.01). Mean scores of reduced personal accomplishment differs significantly among novice, less experienced and experienced groups.

Investigation of Mean Difference based on Experience

To find whether the experience of unaided school teachers (Novice, less experienced and experienced) can significantly affect the technostress, teacher autonomy, teacher burnout and its components, one-way ANOVA was employed.

Results of the one-way ANOVA of technostress and its components are given in table 38

Table 38

Results of the One-way ANOVA of T	Technostress and its Components
-----------------------------------	---------------------------------

Variables	Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
	Between Groups	9.450	2	4.725	240	796
Techno	Within Groups	9965.179	507	19.655	.240	./80
msecurity	Total	9974.629	509			
	Between Groups	23.758	2	11.879	076	420
Techno	Within Groups	7294.705	507	14.388	.820	.439
complexity	Total	7318.463	509			
Techno Invasion	Between Groups	8.777	2	4.388	122	075
	Within Groups	16714.157	507	32.967	.133	.873
	Total	16722.933	509			
	Between Groups	53.730	2	26.865	1.01	266
Techno Awareness	Within Groups	13537.645	507	26.701	1.01	.300
1 Wareness	Total	Irce of VarianceSum of SquaresDfween Groups9.4502hin Groups9965.179507Total9974.629509ween Groups23.7582hin Groups7294.705507Total7318.463509ween Groups8.7772thin Groups16714.157507Total16722.933509ween Groups53.7302thin Groups13537.645507Total13591.375509ween Groups4.2392thin Groups10586.302507Total10590.541509ween Groups174.1282thin Groups170070.272507Total170244.400509				
	Between Groups	4.239	2	2.119	102	002
Techno Facility	Within Groups	10586.302	507	20.880	.102	.903
ruenny	Total	10590.541	509			
	Between Groups	174.128	2	87.064	260	770
Technostress	Within Groups	170070.272	507	335.444	.200	.112
	Total	170244.400	509			

Table 38 conveys that there is no significant effect of experience on technostress of unaided school teachers (F (2, 507) =.260, p>.05). There is no significant effect of

experience on techno insecurity of unaided school teachers (F (2, 507) =.240, p>.05). There is no significant effect of experience on techno complexity of unaided school teachers (F (2, 507) = 826, p>.05). There is no significant effect of experience on techno invasion of unaided school teachers (F (2, 507) =.875, p>.05). There is no significant effect of experience on techno awareness of unaided school teachers (F (2, 507) = 1.01, p>.05). There is no significant effect of experience on techno facility of unaided school teachers (F (2, 507) = 1.01, p>.05). There is no significant effect of experience on techno facility of unaided school teachers (F (2, 507) = .102, p>.05).

Results of the one-way ANOVA of teacher autonomy and its components are given in table 39.

Table 39

<i>Results of the</i>	One-wav $A\Lambda$	IOVA of Tea	cher Autonomv	and its (Components
<i>.</i>	~		2		

Variables	Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
VariablesSource ofEstablishing School IdentityBetween GSchool IdentityWithin GrTeaching and assessmentBetween GParental InvolvementBetween GParental InvolvementWithin GrProfessional DevelopmentBetween GExtra Curricular SubjectsBetween GCurriculum developmentWithin GrCurriculum developmentBetween GMithin GrToSubjectsToCurriculum developmentBetween GTeacher AutonomyBetween GWithin GrToToToTeacher AutonomyTotal	Between Groups	55.098	2	27.549	1 41	246
	Within Groups	9930.018	507	19.586	1.41	.240
	Total	9985.116	509			
Terelineral	Between Groups	279.289	2	139.644	7 10	001
Teaching and	Within Groups	9863.662	507	19.455	/.18	.001
assessment	Total	10142.951	509			
	Between Groups	.695	2	.348	054	049
Parental	Within Groups	3272.521	507	6.455	.054	.948
Involvement	Total	3273.216	509			
	Between Groups	247.649	2	123.824	0.00	001
Professional	Within Groups	6911.726	507	13.633	9.08	.001
Development	Total	7159.375	509			
Extra	Between Groups	55.859	2	27.929	2 60	026
Curricular	Within Groups	3844.433	507	7.583	5.08	.020
Subjects	Total	3900.292	509			
	Between Groups	38.243	2	19.122	2.49	0.05
Curriculum	Within Groups	3904.431	507	7.701	2.48	.085
development	Total	3942.675	509			
	Between Groups	2752.917	2	1376.458	6.40	002
Teacher	Within Groups	107765.876	507	212.556	0.48	.002
Autonomy	Total	110518.792	509			

Table 39 reveals that there is a significant effect of experience on teacher autonomy of unaided school teachers (F (2, 507) = 6.48, p<.01). Mean scores of teacher autonomy differs significantly among novice, less experienced and experienced groups.

Table describes that there is no significant effect of experience on component establishing school identity of unaided school teachers (F (2, 507) =1.41, p>.05). There is a significant effect of experience on component Teaching and assessment of unaided school teachers (F (2, 507) = 7.18, p < .01). Mean scores of Teaching and assessment differs significantly among novice, less experienced and experienced groups. There is no significant effect of experience on component parental involvement of unaided school teachers (F (2, 507) =0.054, p>.05). There is a significant effect of experience on component professional development of unaided school teachers (F (2, 507) = 9.08, p<.01). Mean scores of professional development differs significantly among novice, less experienced and experienced groups. There is a significant effect of experience on component extracurricular subjects of unaided school teachers (F (2, 507) = 3.68, p < .05). Mean scores of extracurricular subjects differs significantly among novice, less experienced and experienced groups. There is no significant effect of experience on component Curriculum development of unaided school teachers (F (2, 507) = 0.085, p > .05).

To check the significance of difference of scores between the Groups, Post Hoc tests were carried out Teacher Autonomy and its components viz., Teaching and assessment, professional development and extracurricular subjects. Results of the Post Hoc tests are presented in table 40

Summary of Post Hoc test for Teacher Autonomy and its Components viz., Teaching and assessment, Professional Development and Extracurricular Subjects by Experience

Variable	Type of M	anagement	Mean Difference	Std. Error	Р
Teaching and	Novice (M = 34.2)	Less Experienced (M = 35.9)	1.672	0.550	.010
assessment	Novice	Experienced $(M = 36.1)$	1.941	0.523	.001
	Less Experienced	Experienced	0.269	0.442	.831
Professional Development	Novice (M = 23.1)	Less Experienced (M = 24.9)	1.815	0.460	.000
	Novice	Experienced $(M = 24.8)$	1.676	0.438	.001
	Less Experienced	Experienced	0.139	0.370	.932
Extra	Novice (M = 21.3)	Less Experienced (M = 22.3)	0.920	0.343	.028
Curricular Subjects	Novice	Experienced $(M = 22)$	0.688	0.327	.110
	Less Experienced	Experienced	0.232	0.276	.703
	Novice (M = 137.6)	Less Experienced (M = 143.6)	6.005	1.818	.005
Teacher Autonomy	Novice	Experienced $(M = 143.3)$	5.638	1.730	.005
	Less Experienced	Experienced	0.367	1.459	.969

Table 40 conveys that there is a significant difference in teacher autonomy for teachers with novice and less experience (MD = 6.005, p<0.01) at .01 level of significance. There is a significant difference in teacher autonomy for teachers with novice and experienced (MD = 5.638, p<0.01) at .01 level of significance. There is no significant difference in teacher autonomy for teachers with less experience and experienced.

Table explains that there is a significant difference in Teaching and assessment for teachers with novice and less experience (MD = 1.672, p<0.01) at .01 level of significance. There is a significant difference in Teaching and assessment for teachers with novice and experienced (MD = 1.941, p<0.01) at .01 level of significance. There is no significant difference in Teaching and assessment for significance. There is no significant difference in Teaching and assessment for

teachers with less experience and experienced. There is a significant difference in professional development for teachers with novice and less experience (MD = 1.815, p<0.01) at .01 level of significance. There is a significant difference in professional development for teachers with novice and experienced (MD = 1.676, p<0.01) at .01 level of significance. There is no significant difference in professional development for teachers with novice and experienced. There is a significant difference in extracurricular subjects for teachers with novice and less experience (MD = .920, p<0.05) at.05 level of significance. There is no significance in a significant difference in extracurricular subjects for teachers with novice & experience and less experience in extracurricular subjects for teachers with novice & experience and less experience & experience and less experience and less experience and less experience in extracurricular subjects for teachers with novice & experience and less experience in extracurricular subjects for teachers with novice & experience and less experience in extracurricular subjects for teachers with novice & experience and less experience & experience and less experience and less experience & experi

Results of the one-way ANOVA of teacher burnout and its components are given in table 41

Table 41

Variables	Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
	Between Groups	396.642	2	198.321	6.356	.002
Emotional Exhaustion	Within Groups	15819.829	507	31.203		
LAnuustion	Total	16216.471	509			
Depersonalization	Between Groups	69.535	2	34.768	1.417	.243
	Within Groups	12437.257	507	24.531		
	Total	12506.792	509			
	Between Groups	231.147	2	115.574	5.280	.005
Reduced Personal	Within Groups	11098.115	507	21.890		
recompnishment	Total	11329.263	509			
	Between Groups	1845.422	2	922.711	5.442	.005
Teacher Burnout	Within Groups	85965.942	507	169.558		
	Total	87811.365	509			

Results of the One-way ANOVA of Teacher Burnout and its Components

Table 41 discloses that there is a significant effect of experience on teacher burnout of unaided school teachers (F (2, 507) = 5.442, p<.01). Mean scores of teacher burnout differs significantly among novice, less experienced and experienced groups.

Table describes that there is a significant effect of experience on emotional exhaustion of unaided school teachers (F (2, 507) = 6.356, p<.01). Mean scores of emotional exhaustion differs significantly among novice, less experienced and experienced groups. There is no significant effect of experience on component depersonalization of unaided school teachers (F (2, 507) = 1.417, p>.05). There is a significant effect of experience on component reduced personal accomplishment of unaided school teachers (F (2, 507) = 5.280, p<.01). Mean scores of reduced personal accomplishment differs significantly among novice, less experienced and experienced groups.

To check the significance of difference of scores between the Groups, Post Hoc tests were carried out Teacher Burnout and its components viz., emotional exhaustion and reduced personal accomplishment. Results of the Post Hoc tests are presented in table 42

Table 42

Variable	Type of m	anagement	Mean difference	Std. Error	Р
Emotional	Novice (M=29.1)	Less Experienced (M= 27.9)	1.274	0.697	.189
Emotional Exhaustion	Novice	Experienced (M= 26.8)	2.322	0.663	.002
	Less Experienced	Experienced	1.049	0.559	.173
Reduced	Novice (M= 22.6)	Less Experienced (M=21.2)	1.459	0.583	.045
Personal Accomplishment	Novice	Experienced (M= 20.8)	1.783	0.555	.006
	Less Experienced	Experienced	0.323	0.468	.788
	Novice (M= 77)	Less Experienced (M=73.7)	3.365	1.624	.118
Teacher Burnout	Novice	Experienced (M= 71.9)	5.092	1.545	.005
	Less Experienced	Experienced	1.727	1.303	.417

Summary of Post Hoc Test for Teacher Burnout and its Components viz., Emotional Exhaustion and Reduced Personal Accomplishment by Experience

154 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

Table 42 conveys that there is no significant difference in teacher burnout for teachers with novice and less experience. There is a significant difference teacher burnout for teachers with novice and experienced (MD = 5.092, p<0.01) at .01 level of significance. There is no significant difference in teacher burnout for teachers with less experience and experienced.

Table explains that there is no significant difference in emotional exhaustion for teachers with novice and less experience. There is a significant difference in emotional exhaustion for teachers with novice and experienced (MD = 2.322, p<0.01) at .01 level of significance. There is no significant difference in emotional exhaustion for teachers with less experience and experienced. There a significant difference in reduced personal accomplishment for teachers with novice and less experience (MD = 1.459, p<0.01) at .01 level of significance. There is a significant difference in reduced personal accomplishment for teachers with novice and experience (MD = 1.783, p<0.01) at .01 level of significance. There is no significant difference in reduced personal accomplishment for teachers with novice and experienced (MD = 1.783, p<0.01) at .01 level of significance. There is no significant difference in reduced personal accomplishment for teachers with novice and experienced (MD = 1.783, p<0.01) at .01 level of significance. There is no

Analysis of Variance

The major objective of the study is to find out the influence of Technostress and Teacher Autonomy on Teacher Burnout of unaided school teachers. The main effect and interaction effect of two independent variables on the dependent variable, Teacher Burnout is studied for this purpose. Two-way ANOVA with 3x3 factorial design was carried out separately for the total sample and subgroups on the basis of Gender, Type of School, Experience, Locality, Subject, Computer Course and Educational Qualification. The 3 x 3 factorial design of two-way ANOVA includes three levels of Technostress and three levels of Teacher Autonomy. Technostress was classified as low level stress, moderate level stress and high level stress. Teacher Autonomy was categorized as scant autonomy, moderate autonomy and high autonomy group.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for Total Sample

Main and interaction effect of Technostress and Teacher Autonomy on the Teacher Burnout for total sample was computed. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for total sample are presented in table 43

Table 43

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Unaided School Teachers for Total Sample

Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
Technostress	12726.528	2	6363.264	50.878	.001
Teacher Autonomy	7494.138	2	3747.069	29.96	.001
Technostress - Teacher Autonomy	2350.543	4	587.636	4.70	.001
Error	62660.723	501	125.071		

Main Effects

Influence of Technostress on Teacher Burnout for Total Sample

Table 43 presents that F value for Technostress is 50.88 which is significant at .01 level with df = 2/501. Therefore, there exists significant difference in the mean scores of Teacher Burnout of unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for total sample. Scheffés test was used for the of post hoc comparison to know which group's mean score of Teacher Burnout is significantly higher. The results of Scheffés test of post hoc comparison of mean scores of Teacher Burnout of unaided school teachers for total sample among three groups of technostress are presented in Table 44

Table 44

Summary of Post Hoc Test for Teacher Burnout for Total Sample by Technostress Group

Variable	Technostress group		Mean difference	Std. Error	р
Teacher Burnout	Low Level Stress (M= 66.56)	Moderate Level Stress(M= 73.58)	7.013	1.209	.001
	Low Level Stress	High Level Stress(M= 80.12)	13.557	1.254	.001
	Moderate Level Stress	High Level Stress	6.543	1.189	.001

From the table 44 it is evident that mean difference scores of low level stress group and moderate level stress group is 7.013, which is significant at .01 level. This point outs that there is significant difference in the Teacher Burnout of unaided school teachers with low and moderate level of technostress. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low level stress group and high level stress group is 13.557, which is significant at .01 level. This describes that there is significant difference in the Teacher Burnout of unaided school teachers with low and high level of technostress. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group is 6.543, which is significant at .01 level. This illustrates that there is significant difference in the Teacher Burnout of unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout of unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout of unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for total sample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Total Sample

Table 43 explains that F value for Teacher Autonomy is 29.96 which is significant at .01 level with df= 2/501. Therefore, there exists significant difference in the mean scores of Teacher Burnout of unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for total sample.

Scheffés test was used for post hoc comparison to know which group's mean score of Teacher Burnout is significantly higher. The results of analysis are presented in Table 45.

Table 45

Summary of Post Hoc test for Teacher Burnout for Total Sample by Teacher Autonomy Group

Variable	Teacher Auto	onomy Group	Mean Difference	Std. Error	р
	Scant Autonomy (M= 78.68)	Moderate Autonomy (M = 74.51)	4.168	1.205	.003
Teacher Burnout	Scant Autonomy	High Autonomy $(M = 67.43)$	11.249	1.291	.001
	Moderate Autonomy	High Autonomy	7.081	1.186	.001

From the table 45 it is observed that mean difference scores of scant autonomy group and moderate autonomy group is 4.168, which is significant at .01 level. This displays that there is significant difference in the Teacher Burnout of unaided school teachers with scant and moderate level of autonomy. Mean Teacher

158 INFLUENCE OF TECHNOSTRESS & TEACHER AUTONOMY ON BURNOUT

Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 11.249, which is significant at .01 level. This indicates that there is significant difference in the Teacher Burnout of unaided school teachers with scant and high level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 7.081, which is significant at .01 level. This explains that there is significant difference in the Teacher Burnout of unaided school teachers with high and moderate level of autonomy. Mean Teacher Burnout score is high for moderate autonomy group is 7.081, which is significant at .01 level. This explains that there is significant difference in the Teacher Burnout of unaided school teachers with high and moderate level of autonomy. Mean Teacher Burnout score is high for moderate autonomy group than high autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for total sample of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for Total Sample

Table 43 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for total sample is 4.70 which is significant at 0.01 level. It indicates that there exists significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 501) =4.70, p<.01, for total sample at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of total sample. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for unaided school teachers. In order to know the trend of interaction between Technostress and Teacher Autonomy on Teacher Burnout for total sample, Profile Plot has been plotted and presented in Figure 13

Figure 13

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Total Sample



Figure 13 represents the mean scores of low level, moderate level and high level stress groups belonging to scant autonomy class is higher than moderate and high autonomy groups. This is an explicit indication of the reliance of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for total sample.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Male Unaided School Teachers

Influence of the Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample male was computed. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress

$160 \ \mathrm{Influence} \ \mathrm{Of} \ \mathrm{Technostress} \ \mathrm{\&} \ \mathrm{Teacher} \ \mathrm{autonomy} \ \mathrm{on} \ \mathrm{burnout}$

and Teacher Autonomy on Teacher Burnout for subsample male are displayed in table 46

Table 46

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Male Unaided School Teachers

Source of Variance	Sum of Squares	Df	Mean Square	F	Sig.
Technostress	2924.304	2	1462.152	10.106	.001
Teacher Autonomy	1570.027	2	785.013	5.426	.006
Technostress - Teacher Autonomy	2837.936	4	709.484	4.904	.001
Error	14178.367	98	144.677		

Main Effects

Influence of Technostress on Teacher Burnout for Male Subsample

Table 46, reveals that F value for Technostress is 10.106 which is significant at .01 level with df= 2/98. Therefore, there exists significant difference in the mean scores of Teacher Burnout of male unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for male teachers.

Scheffés test was used for post hoc comparison to identify which group's mean score of Teacher Burnout is significantly higher. The results are presented in Table 47

Summary of Post Hoc Test for Teacher Burnout for the Male Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 68.74)	Moderate Level Stress (M= 80.07)	11.33	3.27	.004
Teacher Burnout	Low Level Stress	High Level Stress (M= 83.02)	14.28	2.84	.001
	Moderate Level Stress	High Level Stress	2.94	2.84	.587

From the table 47, it is clear that mean difference scores of low-level and moderate level stress groups is 11.33, which is significant at .01 level.

This indicates that Teacher Burnout of unaided teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low level and high level stress groups is 14.28, which is significant at .01 level. This reveals that Teacher Burnout of male unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level and moderate level stress groups is 2.94, which is not significant at.05 level. This reveals that teacher Burnout of male unaided school teachers with high and moderate level of technostress do not differ significantly. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for male subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for male subsample

Table 46 indicates that F value for Teacher Autonomy is 5.426 which is significant at .01 level with df=2/98. Therefore, there exists significant difference in the mean scores of Teacher Burnout of unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the male teachers.

Scheffés test was used for post hoc comparison to recognize which group's mean score of Teacher Burnout is significantly higher. The results are exhibited in Table 48

Table 48

Variable	Teacher Autonomy group		Mean Difference	Std. Error	Р
	Scant Autonomy (M= 81.65)	Moderate Autonomy (M= 81.49)	.170	2.74	.998
Teacher Burnout	Scant Autonomy	High Autonomy (M= 71.23)	10.42	2.94	.003
	Moderate Autonomy	High Autonomy	10.25	2.92	.003

Summary of Post Hoc test for Teacher Burnout for Male Subsample by Teacher Autonomy Group

Table 48 displays that mean difference scores of scant and moderate autonomy groups is .170, which is not significant at .01 level. This illustrates that Teacher Burnout of male teachers with scant and moderate levels of autonomy do not differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant and high autonomy groups is 10.42, which is significant at .01 level. This indicates that Teacher Burnout of male unaided school teachers with scant and high levels of autonomy differ significantly. Mean Teacher
Burnout score is high for scant autonomy group. Mean difference scores of high and moderate autonomy groups is 10.25, which is significant at .01 level. This reveals that Teacher Burnout of male teachers with high and moderate levels of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy groups differ significantly in Teacher Burnout for male unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Male

Table 46 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample male is 4.904 which is significant at 0.01 level. It indicates that there exists significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 98) =4.904, p<.01, for subsample male at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample male. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for male unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for total sample, Profile Plot has been plotted and presented in Figure 14.

Figure 14

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Male Subsample



Figure 14 displays that the mean scores of low level stress group belonging to scant autonomy category is higher than high autonomy group but lower than moderate group. In the case of moderate level stress group, the mean score of scant autonomy is higher than moderate and high autonomy groups. The mean scores of high level stress group belonging to high autonomy is higher than moderate and slant autonomy. This is an indication of the dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for male teachers.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Female unaided school teachers.

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample female was calculated. The results of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample female are displayed in table 49.

Table 49

Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Female Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	8637.365	2	4318.683	37.946	.001
Teacher Autonomy	6333.584	2	3166.792	27.825	.001
Technostress - Teacher Autonomy	1041.651	4	260.413	2.288	.059
Error	44841.482	394	113.811		

Main Effects

Influence of Technostress on Teacher Burnout for Female Subsample

Table 49 reveals that F value for Technostress is 37.946 which is significant at .01 level with df= 2/394. Therefore, there exists significant difference in the mean scores of Teacher Burnout of female unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for female teachers.

Scheff'és test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are exhibited in Table 50

Summary of Post Hoc Test for Teacher Burnout for the Female Subsample by Technostress Group

Variable	Technostress group		Mean difference	Std. Error	Р
	Low Level Stress (M= 66.10)	Moderate Level Stress (M= 72.51)	6.41	1.26	.001
Teacher Burnout	Low Level Stress	High Level Stress (M= 78.74)	12.64	1.39	.001
	Moderate Level Stress	High Level Stress	6.22	1.31	.001

From the table 50, it is evident that mean difference scores of low-level stress group and moderate level stress group is 6.41, which is significant at .01. This indicates that Teacher Burnout of female unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 12.648, which is significant at .01 level. This displays that Teacher Burnout of female unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high level of technostress differ significantly. Mean Teacher Burnout score is high level stress group. Mean difference scores of high level stress group and high level stress group is 1.31, which is not significant at .01 level. This shows that there no significant difference in the Teacher Burnout of female unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for female subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Female Subsample

Table 49, reveals F value for Teacher Autonomy is 27.8256 which is significant at .01 level with df= 2/394. Therefore, there exists significant difference in the mean scores of Teacher Burnout of female unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the female subsample.

Scheffés test was used for post hoc comparison to find out mean score which of Teacher Burnout groups is significantly higher. The results are presented in Table 51

Table 51

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
	Scant Autonomy (M=77.63)	Moderate Autonomy (M= 72.92)	4.71	1.31	.002
Teacher Burnout	Scant Autonomy	High Autonomy (M= 66.51)	11.12	1.40	.001
	Moderate Autonomy	High Autonomy	6.41	1.26	.001

Summary of Post Hoc Test for Teacher Burnout the Female Su sample by Teacher Autonomy Group

Table 51, presents that mean difference scores of scant autonomy and moderate autonomy groups is 4.71, which is significant at .01 level. This shows that there is a significant difference in the Teacher Burnout of female unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 11.12, which is significant at .01 level. This reveals that Teacher Burnout of female unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant

autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 6.41, which is significant at .01 level. This indicates Teacher Burnout of male unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample female of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Female

Table 49 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample male is 2.288 which is not significant at 0.05 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 394) =2.288, p>05, for subsample female at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample female. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for female unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for female subsample, Profile Plot has been plotted and presented in Figure 15.

Figure 15

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Female Subsample



Figure 15 displays that the mean scores of low level stress group, moderate level stress group and high level stress group belonging to scant autonomy category is higher than moderate and high autonomy groups. This is a sign of the dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for female teachers.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Rural Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample rural teacher was calculated. The results are exhibited in table 52

Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Rural Unaided School Teachers

Source of Variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	3138.516	2	1569.258	12.914	.001
Teacher Autonomy	2940.898	2	1470.449	12.101	.001
Technostress * Teacher Autonomy	743.246	4	185.812	1.529	.195
Error	26612.192	219	121.517		

Main Effects

Influence of Technostress on Teacher Burnout for Rural Teacher Subsample

Table 52 shows F value for Technostress is 12.914 which is significant at .01 level with df= 2/219. Therefore, there exists significant difference in the mean scores of Teacher Burnout of rural unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there exists significant influence of Technostress on Teacher Burnout for rural teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout groups is significantly higher. The results are displayed in Table 53

Table 53

Summary of Post Hoc test for Teacher Burnout for the Rural Teacher Subsample by Technostress Group

Variable	Technos	Mean difference	Std. Error	Р	
	Low Level Stress (M= 67.45)	Moderate Level Stress (M= 70.91)	3.46	1.81	.163
Teacher Burnout	Low Level Stress	High Level Stress (M= 77.05)	9.60	1.83	.001
	Moderate Level Stress	High Level Stress	6.14	1.74	.002

From the table 53, it is evident that mean difference scores of low-level stress group and moderate level stress group is 1.81, which is not significant at.05 level. This indicates that Teacher Burnout of rural teachers with low and moderate level of technostress do not differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 1.83 which is significant at .01 level. This displays that Teacher Burnout of rural teachers with low and high level of technostress differ significantly. Mean Teacher Burnout of rural teachers with low and high level stress group. Mean difference scores of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group is 1.74, which is significant at .01 level. This exhibits that Teacher Burnout of rural teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is 1.74, which is significant at .01 level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for rural subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Rural Subsample

Table 52, reveals F value for Teacher Autonomy is 12.101 which is significant at .01 level with df= 2/219. Therefore, there exists significant difference in the mean scores of Teacher Burnout of rural unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the rural teacher subsample.

Scheffés test was used for post hoc comparison to identify mean scores of which Teacher Burnout group is significantly higher. The results are displayed in Table 54

Summary of Post Hoc Test for Teacher Burnout the Rural Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy group		Mean Difference	Std. Error	Р
T 1	Scant Autonomy (M= 76.19)	Moderate Autonomy (M= 73.51)	2.68	1.84	.349
Teacher Burnout Scant Autonomy Moderate Autonomy	Scant Autonomy	High Autonomy (M= 66.97)	9.21	1.91	.001
	Moderate Autonomy	High Autonomy	6.53	1.69	.001

Table 54 presents that mean difference scores of scant autonomy and moderate autonomy group is 2.68, which is not significant at .01 level. This points out that Teacher Burnout of rural teachers with scant and moderate level of autonomy do not differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 9.21, which is significant at .01 level. This indicates that Teacher Burnout of rural teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout of rural teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 6.53, which is significant at .01 level. This reveals that Teacher Burnout of rural unaided teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample rural unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Rural

Table 52 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample rural is 1.529 which is not significant at 0.05 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 219) =.1.529, p>.05, for subsample less experienced teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample rural teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for rural unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for rural teacher subsample, Profile Plot has been plotted and presented in Figure 16

Figure 16





Figure 16 shows that the mean scores of low level stress group belonging to scant autonomy category is higher than high autonomy group but lower than moderate group. Moderate level, and high level stress groupbelonging to scant autonomy is higher than moderate and high autonomy groups. This indicates the dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for rural teacher.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Urban unaided school teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample urban teacher was calculated, the results are exhibited in table 55

Table 55

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Urban Unaided School Teachers

Source of Variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	10896.272	2	5448.136	19.734	.001
Teacher Autonomy	3750.077	2	1875.039	15.166	.001
Technostress * Teacher Autonomy	2010.323	4	502.581	4.065	.003
Error	33751.123	273	123.630		

Main Effects

Influence of Technostress on Teacher Burnout for Urban Teacher Subsample

Table 55 exposes that F value for Technostress is 19.734 which is significant at .01 level with df= 2/273. Therefore, mean scores of Teacher Burnout of urban unaided school teachers for the technostress groups differ significantly i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there exists significant influence of Technostress on Teacher Burnout for urban teacher subsample.

Scheffés test was used for post hoc comparison to recognize mean score of which Teacher Burnout group is significantly higher. The results are displayed in Table 56

Table 56

Summary of Post Hoc Test for Teacher Burnout for the Urban Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 65.86)	Moderate Level Stress(M= 75.56)	9.70	1.60	.001
Teacher Burnout	Low Level Stress	High Level Stress(M= 82.91)	17.05	1.69	.001
	Moderate Level Stress	High Level Stress	7.34	1.60	.001

From the table 56, it is evident that mean difference scores of low-level stress group and moderate level stress group is 9.70, which is significant at .01 level. This indicates that the Teacher Burnout of urban teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 17.05, which is significant at .01 level. This displays that Teacher Burnout of urban unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group is 7.34, which is significant at .01 level. This shows that Teacher Burnout of urban unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for urban subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for urban subsample

Table 55 shows F value for Teacher Autonomy is 15.166 which is significant at .01 level with df= 2/273. Therefore, there exists significant difference in the mean scores of Teacher Burnout of urban unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the urban teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The outcomes of post hoc comparison of mean scores of Teacher Burnout of unaided school teachers for the subsample urban teacher among three groups of Teacher Autonomy are presented in Table 57

Table 57

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
	Scant Autonomy (M= 80.32)	Moderate Autonomy (M=75.31)	5.01	1.57	.007
Teacher Burnout	Scant Autonomy	High Autonomy (M= 67.88)	12.43	1.73	.001
	Moderate Autonomy	High Autonomy	7.42	1.63	.001

Summary of Post Hoc Test for Teacher Burnout the Urban Teacher Subsample by Teacher Autonomy Group

Table 57 displays that mean difference scores of scant autonomy and moderate autonomy groups is 5.01, which is significant at .01. This points out

Teacher Burnout of urban unaided teachers with scant and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 12.43, which is significant at .01. This presents Teacher Burnout of urban unaided teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 7.42, which is significant at .01. This presents Teacher Burnout score is high for scant autonomy group is 7.42, which is significant at .01. This presents Teacher Burnout of urban unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly.

Therefore, it is evident that scant autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample urban unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Urban

Table 55 reveals F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample urban is . 4.065 which is significant at 0.01 level. It indicates that there exists significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 273) =.4.065, p<.01, for subsample urban teacher at .01 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample urban teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for urban teacher subsample, Profile Plot has been plotted and presented in Figure 17.

Figure 17

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Urban Teacher Subsample



Figure 17 shows that the mean scores of low level stress and moderate level stress groups belonging to scant autonomy group is higher than moderate and high autonomy groups. The mean scores of high level stress group belonging to scant autonomy is higher than high autonomy group but lower than moderate group. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample urban teacher.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of CBSE Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample CBSE was calculated. The outcomes are displayed in table 58.

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of CBSE Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	7937.051	2	3968.526	30.722	.001
Teacher Autonomy	4535.358	2	2267.679	17.555	.001
Fechnostress *1344.690Feacher Autonomy1344.690		4	336.173	2.602	.036
Error	40819.217	316	129.175		

Main Effects

Influence of Technostress on Teacher Burnout for CBSE Subsample

Table 58 reveals F value for Technostress is 30.722 which is significant at .01 level with df= 2/316. Therefore, there exists significant difference in the mean scores of Teacher Burnout of CBSE unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for teachers in CBSE School.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The outcomes are exhibited in Table 59

Table 59

Summary of Post Hoc Test for Teacher Burnout for the CBSE Subsample by Technostress Group

Variable	Technostr	Mean Difference	Std. Error	Р	
T 1	Low Level Stress (M= 66.86)	Moderate Level Stress (M= 74.24)	7.38	1.50	.001
Teacher Burnout	Low Level Stress	High Level Stress (M= 80.17)	13.31	1.62	.001
	Moderate Level Stress	High Level Stress	5.93	1.55	.001

From the table 59, it is clear that mean difference scores of low-level stress group and moderate level stress group is 7.38, which is significant at .01. This displays Teacher Burnout of CBSE School teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 13.31, which is significant at .01. This Teacher Burnout of CBSE school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group is 5.93, which is significant at .01. This reveals that there is a significant difference in the Teacher Burnout of CBSE school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for CBSE subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for CBSE Subsample

Table 58 discloses F value for Teacher Autonomy is 17.555 which is significant at .01 level with df= 2/316. Therefore, there exists significant difference in the mean scores of Teacher Burnout of CBSE unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the CBSE subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The outcomes are exhibited in Table 60

Summary of Post Hoc Test for Teacher Burnout the CBSE Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std.Error	Р
- 1	Scant Autonomy (M= 79.43)	Moderate Autonomy (M= 73.67)	5.75	1.54	.001
Teacher Burnout	Scant Autonomy	High Autonomy (M= 68.05)	11.38	1.64	.001
	Moderate Autonomy	High Autonomy	5.62	1.51	.001

Table 60 displays that mean difference scores of scant autonomy group and moderate autonomy group is 5.75, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of CBSE school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy and high autonomy groups is 11.38, which is significant at .01. This illustrates Teacher Burnout of female CBSE School teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group. Mean difference scores of high evel of autonomy differ significant at .01. This states Teacher Burnout of CBSE unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group is 5.62, which is significant at .01. This states Teacher Burnout of CBSE unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant provide the school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant provide the school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample CBSE of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample CBSE

Table 58 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample

male is 2.602 which is significant at 0.05 level. It indicates that there exists significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F(4, 316) = 2.602, p<.01, for subsample CBSE at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample CBSE. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for CBSE unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for CBSE subsample, Profile Plot has been plotted and presented in Figure 18

Figure 18

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for CBSE Subsample



Figure 18 displays that the mean scores of low level stress, moderate level stress and high level stress groups belonging to scant autonomy category is higher

than moderate and high autonomy groups. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample CBSE.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of State Syllabus Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample State syllabus was calculated. The outcomes are presented in table 61.

Table 61

Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of State Syllabus Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	4992.274	2	2496.137	20.837	.001
Teacher Autonomy	3358.840	2	1679.420	14.019	.001
Technostress * Teacher Autonomy	993.561	4	248.390	2.073	.086
Error	21083.938	176	119.795		

Main Effects

Influence of Technostress on Teacher Burnout for State Syllabus Subsample

Table 61 reveals F value for Technostress is 20.837 which is significant at .01 level with df= 2/176. Therefore, there exists significant difference in the mean scores of Teacher Burnout of State syllabus unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for teachers in State syllabus school.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 62

Summary of Post Hoc test for Teacher Burnout for the State Syllabus Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 65.96)	Moderate Level Stress (M= 72.29)	6.33	2.05	.010
Teacher Burnout	Low Level Stress	High Level Stress (M= 80.06)	14.09	2.02	.001
	Moderate Level Stress	High Level Stress	7.76	1.88	.001

From the table 62, it is clear that mean difference scores of low-level stress group and moderate level stress group is 6.33, which is significant at .01. This displays Teacher Burnout of State syllabus school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 14.09, which is significant at .01. This indicates Teacher Burnout of State syllabus unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level and moderate level stress groups is 7.76, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of State syllabus unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for State syllabus subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for State Syllabus Subsample

Table 61 displays F value for Teacher Autonomy is 14.019 which is significant at .01 level with df= 2/176. Therefore, there exists significant difference in the mean scores of Teacher Burnout of State syllabus unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the State syllabus subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 63.

Table 63

Summary of Post Hoc Test for Teacher Burnout the State Syllabus Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy group		Mean Difference	Std. Error	Р
Teacher Burnout	Scant Autonomy (M= 77.35)	Moderate Autonomy (M= 75.92)	1.43	1.95	.763
	Scant Autonomy	High Autonomy (M= 66.28)	11.08	2.12	.001
	Moderate Autonomy	High Autonomy	9.64	1.93	.001

Table 63 represents mean difference scores of scant autonomy and moderate autonomy groups is 1.43, which is not significant at.05. This displays that there is no significant difference in the Teacher Burnout of State syllabus unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 11.08, which is significant at .01. This indicates Teacher Burnout of State syllabus unaided school teachers with scant and high level of autonomy differ

significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 9.64, which is significant at .01. This describes Teacher Burnout of State syllabus unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample State syllabus of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample State Syllabus

Table 61 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample male is 2.073 which is not significant at 0.05 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 176) =2.073, p>.05, for subsample State syllabus at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample State syllabus. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for State syllabus unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for State syllabus subsample, Profile Plot has been plotted and presented in Figure 19

Figure 19

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for State Syllabus Subsample



Figure 19 shows that the mean scores of low level stress and high level stress groups belonging to scant autonomy category is higher than high autonomy group but lower than moderate group. Moderate level stress group, the mean score of scant autonomy is higher than moderate and high autonomy groups. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample State syllabus.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers with Basic Qualification

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample basic qualification teacher was calculated first and their interaction effect on dependent

$188 \ {\rm Influence} \ {\rm of} \ {\rm technostress} \ {\rm a} \ {\rm teacher} \ {\rm autonomy} \ {\rm on} \ {\rm burnout}$

variable was also found out. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample basic qualification are presented in table 64

Table 64

Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of basic Qualification Unaided School Teachers

Source of Variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	5471.890	2	2735.945	22.288	.001
Teacher Autonomy	7891.089	2	3945.545	32.141	.001
Technostress * Teacher Autonomy	2049.669	4	512.417	4.174	.003
Error	35108.097	286	122.756		

Main Effects

Influence of Technostress on Teacher Burnout for basic Qualification Teacher Subsample

Table 64 presents F value for Technostress is 22.288 which is significant at .01 level with df= 2/286. Therefore, there exists significant difference in the mean scores of Teacher Burnout of basic qualification unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. So, there is significant influence of Technostress on Teacher Burnout for basic qualification teacher subsample.

Scheffés test was used for post hoc comparison. The results are presented in Table 65

Summary of Post Hoc test for Teacher Burnout for the basic Qualification Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	р
Teacher Burnout	Low Level Stress (M= 66.64)	Moderate Level Stress (M= 72)	5.37	1.63	.005
	Low Level Stress	High Level Stress (M= 79.31)	12.67	1.65	.001
	Moderate Level Stress	High Level Stress	7.29	1.50	.001

From the table 65 shows that mean difference scores of low-level stress group and moderate level stress group is 5.37, which is significant at .01. This displays Teacher Burnout of basic qualification unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 12.67, which is significant at .01. This points out Teacher Burnout of basic qualified unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress and moderate level stress groups is 7.29, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of basic qualification unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for basic qualification subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for basic Qualification Subsample

Table 64 indicates F value for Teacher Autonomy is 32.141 which is significant at .01 level with df= 2/286. Therefore, there exists significant difference in the mean scores of Teacher Burnout of basic qualification unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the basic qualification teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 66

Table 66

Summary of Post Hoc Test for Teacher Burnout the basic Qualification Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	р
	Scant Autonomy (M= 80.04)	Moderate Autonomy (M= 74.81)	5.26	1.59	.005
Teacher Burnout	Scant Autonomy	High Autonomy (M= 65.04)	15.03	1.67	.001
	Moderate Autonomy	High Autonomy	9.77	1.52	.001

Table 66 displays that mean difference scores of scant autonomy and moderate autonomy group is 2.02, which is significant at .01. This indicates Teacher Burnout of basic qualified unaided school teachers with scant and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 10.41, which is significant at .01. This presents Teacher Burnout

of basic qualified unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 8.39, which is significant at .01 level of significance. This shows that there is significant difference in the Teacher Burnout of basic qualification unaided school teachers with high and moderate level of autonomy. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample basic qualification of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample basic Qualification

Table 64 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample basic qualification is 4.174 which is significant at 0.01 level. It indicates that there exists a significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 286) =4.174, p<.01, for subsample basic qualification teacher at .01 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample basic qualification teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for basic qualification unaided school teachers. In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for basic qualification teacher subsample, Profile Plot has been plotted and presented in Figure 20

Figure 20

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for basic Qualified Teacher Subsample



Figure 20 displays that the mean scores of low-level stress, moderate level stress and of high level stress groups belonging to scant autonomy category is higher than high autonomy group and lower than moderate group. This indicates dependency of Teacher Burnout on the interaction between Technostress and

Teacher Autonomy for the subsample basic qualification.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Highly Qualified Unaided Secondary School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample highly qualified teacher was calculated first and their interaction effect on dependent variable was also found out. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample highly qualified are exhibited in table 67

Table 67

Summary of 3x3 factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Highly Qualified Unaided School Teachers

Source of variance	Sum of Squares	Df	Mean Square	F	Sig.
Technostress	7999.472	2	3999.736	34.364	.001
Teacher Autonomy	619.949	2	309.975	2.663	.072
Technostress * Teacher Autonomy	1671.193	4	417.798	3.590	.007
Error	23977.235	206	116.394		

Main Effects

Influence of Technostress on Teacher Burnout for Highly Qualified Teacher Subsample

Table 67 states F value for Technostress is 34.364 which is significant at .01 level with df = 2/206. Therefore, there exists significant difference in the mean scores of Teacher Burnout of highly qualified unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. So, there is significant influence of Technostress on Teacher Burnout for highly qualified teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 68

Summary of Post Hoc Test for Teacher Burnout for the Highly Qualified Teacher Subsample by Technostress Group

Variable	Technostress Group		Mean Difference	Std. Error	р
Teacher Burnout	Low Level Stress (M= 66.48)	Moderate Level Stress (M= 75.82)	9.33	1.73	.001
	Low Level Stress	High Level Stress (M= 81.53)	15.05	1.86	.001
	Moderate Level Stress	High Level Stress	5.71	1.85	.009

From the table 68, it is evident that mean difference scores of low-level stress group and moderate level stress group is 9.33, which is significant at .01. This displays Teacher Burnout of highly qualified unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 15.05, which is significant at .01. This Teacher Burnout of highly qualified unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout of highly qualified unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress and moderate level stress group is 5.71, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of highly qualified unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for highly qualified subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Highly Qualified Subsample

Table 67 reveals F value for Teacher Autonomy is 2.663 which is not significant at 05 level with df= 2/206. Therefore, there exists significant difference in the mean scores of Teacher Burnout of highly qualified unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the highly qualified teacher subsample.

Scheffés test was used of post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The outcomes are presented in Table 69

Table 69

Summary of Post Hoc Test for Teacher Burnout the Highly Qualified Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	р
	Scant Autonomy (M= 76.89)	Moderate Autonomy (M= 74.10)	2.78	1.76	.289
Teacher Burnout	Scant Autonomy	High Autonomy (M= 71.17)	5.72	1.94	.014
	Moderate Autonomy	High Autonomy	2.94	1.79	.263

Table 69 displays mean difference scores of scant autonomy and moderate autonomy group is 2.78, which is not significant at .01. This indicates Teacher Burnout of highly qualified unaided school teachers with scant and moderate level of autonomy do not differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 5.72, which is significant at.05 level of significance. This shows that there is significant difference in the Teacher Burnout of highly qualified

unaided school teachers with scant and high level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy and moderate autonomy groups is 2.94, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of highly qualified unaided school teachers with high and moderate level of autonomy. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample highly qualified of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Highly Qualified

Table 67 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample highly qualified is 3.590 which is significant at 0.01 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 206) =.417.798 p<.01, for subsample highly qualified teacher at .01 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample highly qualified teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for highly qualified unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for highly qualified teacher subsample, Profile Plot has been plotted and presented in Figure 21

Figure 21

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Highly Qualified Teacher Subsample



Figure 21 displays that the mean scores of low-level stress moderate level stress groups belonging to scant autonomy category is higher than moderate and high autonomy groups. The mean scores of high level stress group belonging to scant autonomy is lower than moderate and high autonomy. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for the subsample highly qualified teachers.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Language Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample language teacher was calculated. The results are exhibited in table 70

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Language Unaided School Teachers

Source of variance	Sum of Squares	Df	Mean Square	F	Sig.
Technostress	2366.254	2	1183.127	10.578	.001
Teacher Autonomy	1526.841	2	763.420	6.825	.002
Technostress * Teacher Autonomy	307.330	4	76.833	.687	.602
Error	14652.199	131	111.849		

Main Effects

Influence of Technostress on Teacher Burnout for Language Teacher Subsample

Table 70 reveals F value for Technostress is 10.578 which is significant at.05 level with df= 2/131. Therefore, there exists significant difference in the mean scores of Teacher Burnout of language unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for language teachers.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 71

Table 71

Summary of Post Hoc test for Teacher Burnout for the Language Teacher Subsample by Technostress Group

Variable	Technostress Group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 67.07)	Moderate Level Stress (M= 70.77)	3.69	2.19	.245
Teacher Burnout	Low Level Stress	High Level Stress (M= 78.28)	11.21	2.25	.001
	Moderate Level Stress	High Level Stress	7.51	2.14	.003
The table 71 shows that mean difference scores of low-level stress group and moderate level stress group is 3.69, which is not significant at .01 level of significance. This shows that there is no significant difference in the Teacher Burnout of language unaided school teachers with low and moderate level of technostress. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 11.21, which is significant at .01. This displays Teacher Burnout of language unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group. Mean difference scores of high level stress group and moderate level stress group is 7.51, which is significant at .01. This displays Teacher Burnout of language unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is 7.51, which is significant at .01. This displays Teacher Burnout of language unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is 7.51, when Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for language subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Language Subsample

Table 70 reveals F value for Teacher Autonomy is 6.825 which is significant at .01 level with df= 2/131. Therefore, there exists significant difference in the mean scores of Teacher Burnout of language unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the language teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 72.

Table 72

Summary of Post Hoc Test for Teacher Burnout the Language Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
	Scant Autonomy (M= 77.61)	Moderate Autonomy (M= 72.13)	5.48	2.24	.054
Teacher Burnout	Scant Autonomy	High Autonomy (M= 67.75)	9.86	2.27	.001
	Moderate Autonomy	High Autonomy	4.38	2.10	.119

Table 72 displays that mean difference scores of scant autonomy and moderate autonomy groups is 5.48, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of language unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 9.86, which is significant at .01. This indicates Teacher Burnout of language unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy groups is 4.38, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of language unaided school teachers with high and moderate level of autonomy. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample language unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Language

Table 70 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample

language is .687 which is not significant at 0.01 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F(4, 131) = .918, p>.01, for subsample language teacher at .01 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample language teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for language unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for language teacher subsample, Profile Plot has been plotted and presented in Figure 22

Figure 22





Figure 22 displays that the mean scores of low level stress, moderate level stress and high level stress groups belonging to scant autonomy category is higher than moderate and high autonomy groups. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample language teacher.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Social Science Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample social science teacher was calculated first and their interaction effect on dependent variable was also found out. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample social science are exhibited in table 73

Table 73

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Social Science Unaided School Teachers

Source of Variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	2853.315	2	1426.657	11.704	.001
Teacher Autonomy	1106.121	2	553.060	4.537	.013
Technostress * Teacher Autonomy	2314.961	4	578.740	4.748	.001
Error	14506.030	119	121.899		

Main Effects

Influence of Technostress on Teacher Burnout for Social Science Teacher Subsample

Table 73 points out F value for Technostress is 11.704 which is significant at 05 level with df= 2/119. Therefore, there exists significant difference in the mean

scores of Teacher Burnout of social science unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. So, there is significant influence of Technostress on Teacher Burnout for social science teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 74.

Table 74

Summary of Post Hoc Test for Teacher Burnout for the Social Science Teacher Subsample by Technostress Group

Variable	Technostress Group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 67.09)	Moderate Level Stress (M= 74.91)	7.82	2.49	.009
Teacher Burnout	Low Level Stress	High Level Stress (M= 81.28)	14.18	2.51	.001
	Moderate Level Stress	High Level Stress	6.36	2.26	.022

The table 74 shows that mean difference scores of low-level stress group and moderate level stress group is 7.82, which is significant at .01. This indicates Teacher Burnout of social science unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 14.18, which is significant at .01. This indicates Teacher Burnout of social science unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level of social science unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress groups is 6.36, which is significant at .01. This shows Teacher Burnout of social science unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout of social science unaided school teachers sproup is 14.18.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for social science subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Social Science Subsample

Table 73 displays F value for Teacher Autonomy is 4.537 which is significant at.05 level with df= 2/119. Therefore, there exists significant difference in the mean scores of Teacher Burnout of social science unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the social science teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher burnout group is significantly higher. The results are presented in Table 75

Table 75

Variable	Teacher Autonomy group		Mean Difference	Std. Error	Р
Teacher	Scant Autonomy (M= 79.51)	Moderate Autonomy (M= 76.22)	3.29	2.36	.384
Teacher Burnout	Scant Autonomy	High Autonomy (M= 69.68)	9.83	2.54	.001
	Moderate Autonomy	High Autonomy	6.54	2.34	.023

Summary of Post Hoc Test for Teacher Burnout the Social Science Teacher Subsample by Teacher Autonomy Group

Table 75 displays that mean difference scores of scant autonomy and moderate autonomy groups is 2.29, which is not significant at .01. This indicates

Teacher Burnout of social science unaided school teachers with scant and moderate level of autonomy do not differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 9.83, which is significant at .01. This displays Teacher Burnout of social science unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 6.54, which is significant at .01. This displays Teacher Burnout of social science unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group is 6.54, which is significant at .01. This displays Teacher Burnout of social science unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample unaided school social science teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Social Science

Table 73 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample social science is 4.748 which is significant at 0.01 level. It indicates that there exists a significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 119) =.4.748 p>.01, for subsample social science teacher at .01 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample social science teacher. Therefore, the Teacher

Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for social science unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for social science teacher subsample, Profile Plot has been plotted and presented in Figure 23

Figure 23

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Social Science Teacher Subsample



Estimated Marginal Means of Teacher Burnout

Figure 23 shows that the mean scores of low level stress and high level stress groups belonging to scant autonomy category is higher than high autonomy group and lower than moderate group. Moderate level stress group, the mean score of scant autonomy is higher than moderate and high autonomy groups.. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample social science.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Science Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample science teacher was calculated. The results are presented in table 76.

Table 76

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Science Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	3745.978	2	1872.989	16.031	.001
Teacher Autonomy	2092.947	2	1046.474	8.957	.001
Technostress * Teacher Autonomy	1405.282	4	351.321	3.007	.021
Error	13669.912	117	116.837		

Main Effects

Influence of Technostress on Teacher Burnout for Science Teacher Subsample

Table 76 displays F value for Technostress is 16.031 which is significant at 0.05 level with df = 2/165. Therefore, there exists significant difference in the mean scores of Teacher Burnout of science unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for science teachers.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 77.

Table 77

Summary of Post Hoc Test for Teacher Burnout for the Science Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 66.89)	Moderate Level Stress (M= 74.95)	8.07	2.27	.002
Teacher Burnout	Low Level Stress	High Level Stress (M= 82)	15.11	2.44	.001
	Moderate Level Stress	High Level Stress	7.04	2.42	.017

From the table 77, it is evident that mean difference scores of low-level stress group and moderate level stress group is 8.07, which is significant at .01. This indicates Teacher Burnout of science unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 15.11, which is significant at .01. This displays Teacher Burnout of science unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and high level stress group is 15.11, which is significant at .01. This displays Teacher Burnout of science unaided school teachers with low and high level stress group. Mean difference scores of high level stress group and moderate level stress group is 7.04, which is significant at .01. This displays Teacher Burnout of science unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is 7.04, which is significant at .01. This displays Teacher Burnout of science unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for science subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Science Subsample

Table 76 exposes F value for Teacher Autonomy is 8.957 which is significant at .01 level with df = 2/117. Therefore, there exists significant difference in the mean scores of Teacher Burnout of science unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. Thus, there is significant influence of Teacher Autonomy on Teacher Burnout for the science teachers.

Scheffés test was used for post hoc comparison to identify mean scores of which Teacher Burnout group is significantly higher. The results are presented in Table 78.

Table 78

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
	Scant Autonomy (M= 78.22)	Moderate Autonomy (M= 76.34)	1.87	2.298	.719
Teacher Burnout	Scant Autonomy	High Autonomy (M=65.73)	12.48	2.56	.001
	Moderate Autonomy	High Autonomy	10.61	2.35	.001

Summary of Post Hoc Test for Teacher Burnout the Science Teacher Subsample by Teacher Autonomy Group

Table 78 describes mean difference scores of scant autonomy and moderate autonomy groups is 1.87, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of science unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 12.48, which is significant at .01. This displays Teacher Burnout of unaided school science teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean

difference scores of high autonomy group and moderate autonomy group is 10.61, which is significant at .01. This displays Teacher Burnout of science unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample science of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Science

Table 76 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample science is 3.007 which is significant at 0.05 level. It indicates that there exists a significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 117) = 3.007, p>.01, for subsample science teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample science teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for science unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for science teacher subsample, Profile Plot has been plotted and presented in Figure 24

Figure 24

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Science Teacher Subsample



Figure 24 shows that the mean scores of low level stress group belonging to scant autonomy category is higher than high autonomy group and lower than moderate group. Moderate level stress and high level stress groups, the mean score of scant autonomy is higher than moderate and high autonomy groups... This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample science teacher.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Mathematics Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample mathematics teacher was calculated. The results are presented in table 79.

Table 79

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Mathematics Unaided School Teachers

Source of variance	Sum of Squares	Df	Mean Square	F	Sig.
Technostress	3753.776	2	1876.888	12.607	.001
Teacher Autonomy	3248.890	2	1624.445	10.912	.001
Technostress * Teacher Autonomy	611.282	4	152.820	1.027	.397
Error	15929.195	107	148.871		

Main Effects

Influence of Technostress on Teacher Burnout for Mathematics Teacher Subsample

Table 79 displays F value for Technostress is 12.607 which is significant at .01 level with df= 2/107. Therefore, there exists significant difference in the mean scores of Teacher Burnout of mathematics unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for mathematics teachers.

Scheffés test was used for post hoc comparison to identify mean scores of which Teacher Burnout group is significantly higher. The results are presented in Table 80

Table 80

Summary of Post Hoc Test for Teacher Burnout for the Mathematics Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 65)	Moderate Level Stress (M= 73.98)	8.98	2.75	.006
Teacher Burnout	Low Level Stress	High Level Stress (M= 79.14)	14.14	2.92	.001
	Moderate Level Stress	High Level Stress	5.16	2.71	.169

From the table 80 it is clear that mean difference scores of low-level stress group and moderate level stress group is 8.98, which is significant at .01. This displays Teacher Burnout of mathematics teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 14.14, which is significant at .01. This indicates Teacher Burnout of mathematics teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group is 5.16, which is not significant at.05 level of significance. This shows that there is a significant difference in the Teacher Burnout of mathematics unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for mathematics subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Mathematics Subsample

Table 79 shows F value for Teacher Autonomy is 10.912 which is significant at .01 level with df= 2/107. Therefore, there exists significant difference in the mean scores of Teacher Burnout of mathematics unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the mathematics teacher subsample.

Scheffés test was used for post hoc comparison to identify mean scores of which Teacher Burnout group is significantly higher. The results are presented in Table 81

Table 81

Summary of Post Hoc Test for Teacher Burnout the Mathematics Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
T. 1	Scant Autonomy (M= 79.51)	Moderate Autonomy (M= 73.14)	6.37	2.74	.071
Teacher Burnout	Scant Autonomy	High Autonomy (M= 66.09)	13.42	3.00	.001
	Moderate Autonomy	High Autonomy	7.05	2.73	.040

Table 81 displays that mean difference scores of scant autonomy and moderate autonomy groups is 6.37, which is not significant at.05. This indicates Teacher Burnout of mathematics unaided school teachers with scant and moderate level of autonomy do not differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 13.42, which is significant at .01 level of significance. This shows that there is significant difference in the Teacher Burnout of mathematics unaided school teachers with scant and high level of autonomy. Mean Teacher Burnout score is high for scant autonomy group is 7.05, which is significant at .01. This displays Teacher Burnout of mathematics teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group is 7.05, which is significant at .01. This displays Teacher Burnout of mathematics teachers with high and moderate autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample mathematics of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Mathematics

Table 79 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample

mathematics is 1.027 which is not significant at 0.05 level. It indicates that there is exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 107)= 1.027 p < .05, for subsample mathematics teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample less experienced teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for mathematics unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for mathematics teacher subsample, Profile Plot has been plotted and presented in Figure 25.

Figure 25

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Mathematics Teacher Subsample



Figure 25 displays that the mean scores of low level stress, moderate level stress and high level stress groups belonging to scant autonomy category is higher than moderate and high autonomy groups. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample mathematics teacher

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Novice Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample novice teacher was calculated. The results are presented in table 82

Table 82

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Novice Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	1027.177	2	513.588	4.280	.017
Teacher Autonomy	1908.935	2	954.468	7.953	.001
Technostress * Teacher Autonomy	1157.985	4	289.496	2.412	.055
Error	11160.740	93	120.008		

Main Effects

Influence of technostress on Teacher Burnout for Novice Teacher Subsample

Table 82, presents that F value for Technostress is 4.280 which is significant at 05 level with df= 2/93. Therefore, there exists significant difference in the mean scores of Teacher Burnout of novice unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. Thus, there is significant influence of Technostress on Teacher Burnout for novice teachers.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 83

Table 83

Summary of Post Hoc test for Teacher Burnout for the Novice Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 69.76)	Moderate Level Stress (M= 78.13)	8.37	2.61	.008
Teacher Burnout	Low Level Stress	High Level Stress (M= 82.86)	13.09	2.90	.001
	Moderate Level Stress	High Level Stress	4.72	2.64	.206

From the table 83 shows that mean difference scores of low-level stress group and moderate level stress group is 8.37, which is significant at .01. This displays Teacher Burnout of novice teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 13.09, which is significant at .01. This indicates Teacher Burnout of novice teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group is 4.72, which is not significant at .01 level of significance. This shows that there is no significant difference in the Teacher Burnout of novice unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for novice subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Novice Subsample

Table 69 reveals F value for Teacher Autonomy is 7.953 which is significant at .01 level with df= 2/93. Therefore, there exists significant difference in the mean scores of Teacher Burnout of novice unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the novice teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 84

Table 84

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
Teacher	Scant Autonomy (M= 83.14)	Moderate Autonomy (M= 76.60)	6.54	2.45	.033
Burnout	Scant Autonomy	High Autonomy (M= 66.18)	16.96	2.88	.001
	Moderate Autonomy	High Autonomy	10.42	2.93	.003

Summary of Post Hoc Test for Teacher Burnout the Novice Teacher Subsample by Teacher Autonomy Group

Table 84 displays that mean difference scores of scant autonomy and moderate autonomy groups is 6.54, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of novice unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 16.96, which is significant at .01. This displays Teacher Burnout of novice unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 10.42, which is significant at .01. This displays Teacher Burnout of novice unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample novice of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Novice Teachers

Table 82 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample novice is 2.412 which is not significant at 0.05 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 93) =2.412, p>.05, for subsample novice teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample novice teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for novice unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for novice teacher subsample, Profile Plot has been plotted and presented in Figure 26.

Figure 26

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Novice Teacher Subsample



Figure 26 shows that the mean scores of low level stress group belonging to scant autonomy category is higher than high autonomy group but lower than moderate group. The mean scores of moderate level stress and high level stress groups belonging to scant autonomy is higher than moderate and high autonomy. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample novice teacher.

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Less Experienced Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample less experienced teacher was calculated first and their interaction effect on dependent variable was also found out. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample less experienced are presented in table 85

Table 85

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Less Experienced Unaided School Teachers

Source of Variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	4543.219	2	2271.610	19.734	.000
Teacher Autonomy	2949.387	2	1474.694	12.811	.001
Technostress * Teacher Autonomy	422.856	4	105.714	.918	.455
Error	18993.077	165	115.110		

Main Effects

Influence of Technostress on Teacher Burnout for Less Experienced Teacher Subsample

Table 85 discloses F value for Technostress is 19.734 which is significant at.05 level with df= 2/165. Therefore, there exists significant difference in the mean scores of Teacher Burnout of less experienced unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. So, there is significant influence of Technostress on Teacher Burnout for less experienced teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 86

Table 86

Summary of Post Hoc Test for Teacher Burnout for the Less Experienced Teacher Subsample by Technostress Group

Variable	Technostress Group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 66.94)	Moderate Level Stress (M= 73.54)	6.59	1.96	.004
Teacher Burnout	Low Level Stress	High Level Stress (M= 80.72)	13.78	2.05	.001
_	Moderate Level Stress	High Level Stress	7.18	1.97	.002

From the table 86, it is clear that mean difference scores of low-level stress group and moderate level stress group is 6.59, which is significant at .01. This indicates Teacher Burnout of less experienced unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 13.78, which is significant at .01. This displays Teacher Burnout of less experienced unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress and moderate level stress group is 7.18, which is significant at .01. This shows that there is a significant difference in the Teacher Burnout of less experienced unaided school teachers with high and moderate level of technostress. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for less experienced subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Less Experienced Subsample

Table 85 reveals F value for Teacher Autonomy is 12.811which is significant at .01 level with df= 2/165. Therefore, there exists significant difference in the mean scores of Teacher Burnout of less experienced unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the less experienced teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 87

Table 87

Summary of Post Hoc test for Teacher Burnout the Less Experienced Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Autonomy Group		Mean Difference	Std. Error	Р
	Scant Autonomy (M= 78.27)	Moderate Autonomy (M= 76.25)	2.02	2.08	.625
Teacher Burnout	Scant Autonomy	High Autonomy (M= 67.86)	10.41	2.09	.001
	Moderate Autonomy	High Autonomy	8.39	1.89	.001

Table 87 expresses that mean difference scores of scant autonomy and moderate autonomy group is 2.02, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of less experienced unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 10.41, which is significant at .01. This indicates Teacher Burnout of less experienced unaided school teachers with scant and high level of scant autonomy group is 10.41, which is significant at .01. This indicates Teacher Burnout of less experienced unaided school teachers with scant and high level of

autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 8.39, which is significant at .01. This displays Teacher Burnout of less experienced unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample less experienced unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of unaided school teachers for the subsample less experienced

Table 85 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample less experienced is .918 which is not significant at 0.05 level. It indicates that there exists no significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 165) =.918, p>.05, for subsample less experienced teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample less experienced teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for less experienced unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for less experienced teacher subsample, Profile Plot has been plotted and presented in Figure 27

Figure 27

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Less Experienced Teacher Subsample



Estimated Marginal Means of Teacher Burnout

Figure 27 shows that the mean scores of low level stress and moderate level stress groups belonging to scant autonomy category is higher than moderate and high autonomy groups. The mean scores of high level stress group belonging to scant autonomy is higher than high autonomy but lower than moderate group. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample less experienced teacher

Influence of Technostress and Teacher Autonomy on Teacher Burnout of Experienced Unaided School Teachers

Influence of the independent variables, viz., Technostress and Teacher Autonomy on the dependent variable, Teacher Burnout for the subsample experienced teacher was calculated first and their interaction effect on dependent variable was also found out. The results of 3x3 factorial design ANOVA of the main effect and interaction effects of Technostress and Teacher Autonomy on Teacher Burnout for subs sample experienced are presented in table 88

Table 88

Summary of 3x3 Factorial Design ANOVA of Teacher Burnout by Technostress and Teacher Autonomy of Experienced Unaided School Teachers

Source of variance	Sum of Squares	df	Mean Square	F	Sig.
Technostress	6610.249	2	3305.124	25.137	.001
Teacher Autonomy	2277.856	2	1138.928	8.662	.001
Technostress * Teacher Autonomy	1397.520	4	349.380	2.657	.034
Error	29583.762	225	131.483		

Main Effects

Influence of Technostress on Teacher Burnout for Experienced Teacher Subsample

Table 88, presents F value for Technostress is 25.137 which is significant at 05 level with df= 2/225. Therefore, there exists significant difference in the mean scores of Teacher Burnout of experienced unaided school teachers for the technostress groups i.e., low level stress, moderate level stress and high-level stress group differ significantly. So, there is significant influence of Technostress on Teacher Burnout for experienced teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 89.

Table 89

Summary of Post Hoc Test for Teacher Burnout for the Experienced Teacher Subsample by Technostress Group

Variable	Technostress group		Mean Difference	Std. Error	Р
	Low Level Stress (M= 64.94)	Moderate Level Stress (M=71.11)	6.17	1.86	.005
Teacher Burnout	Low Level Stress	High Level Stress (M= 78.79)	13.85	1.86	.001
	Moderate Level Stress	High Level Stress	7.68	1.79	.001

Table 89 shows that mean difference scores of low-level stress group and moderate level stress group is 6.17, which is significant at .01. This displays Teacher Burnout of experienced unaided school teachers with low and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for moderate level stress group. Mean difference scores of low-level stress group and high level stress group is 13.85, which is significant at .01. This displays Teacher Burnout of experienced unaided school teachers with low and high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores is high for high level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group. Mean difference scores of high level stress group and moderate level stress group is 7.68, which is significant at .01. This Teacher Burnout of experienced unaided school teachers burnout of experienced unaided school teachers burnout of experienced unaided school teacher Burnout of experienced unaided school teachers Burnout of experienced unaided school teachers group and moderate level stress group is 7.68, which is significant at .01. This Teacher Burnout of experienced unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group is 7.68, which is significant at .01. This Teacher Burnout of experienced unaided school teachers with high and moderate level of technostress differ significantly. Mean Teacher Burnout score is high for high level stress group.

Therefore, it is evident that low level stress, moderate level stress and high level stress group differ significantly in teacher burnout for experienced subsample of unaided school teachers. Mean Teacher Burnout score of high level stress group is significantly higher than that of moderate and low stress level groups.

Influence of Teacher Autonomy on Teacher Burnout for Experienced Subsample

Table 78 points out F value for Teacher Autonomy is 8.622 which is significant at .01 level with df= 2/225. Therefore, there exists significant difference in the mean scores of Teacher Burnout of experienced unaided school teachers for the Teacher Autonomy groups i.e., scant autonomy, moderate autonomy and high autonomy group differ significantly. So, there is significant influence of Teacher Autonomy on Teacher Burnout for the experienced teacher subsample.

Scheffés test was used for post hoc comparison to identify mean score of which Teacher Burnout group is significantly higher. The results are presented in Table 90

Table 90

Summary of Post Hoc Test for Teacher Burnout the Experienced Teacher Subsample by Teacher Autonomy Group

Variable	Teacher Auto	Mean Difference	Std. Error	Р	
T 1	Scant Autonomy (M=75.81)	Moderate Autonomy (M= 72.71)	3.10	1.86	.251
Burnout	Scant Autonomy	High Autonomy (M= 67.43)	8.39	2.04	.000
	Moderate Autonomy	High Autonomy	5.28	1.77	.013

Table 90 displays mean difference scores of scant autonomy and moderate autonomy group is 1.86, which is not significant at .01. This shows that there is no significant difference in the Teacher Burnout of experienced unaided school teachers with scant and moderate level of autonomy. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of scant autonomy group and high autonomy group is 8.39, which is significant at .01. This indicates Teacher Burnout of experienced unaided school teachers with scant and high level of autonomy differ significantly. Mean Teacher Burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 5.28, which is significant at .01. This displays Teacher Burnout of experienced unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for scant burnout of experienced unaided school teachers burnout score is high for scant autonomy group. Mean difference scores of high autonomy group and moderate autonomy group is 5.28, which is significant at .01. This displays Teacher Burnout of experienced unaided school teachers with high and moderate level of autonomy differ significantly. Mean Teacher Burnout score is high for moderate autonomy group.

Therefore, it is evident that scanty autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for subsample experienced e of unaided school teachers. Mean Teacher Burnout score of scant autonomy group is significantly higher than that of moderate and high autonomy groups.

Interaction Effects

Interaction Effect of Technostress and Teacher Autonomy on Teacher Burnout of Unaided School Teachers for the Subsample Experienced

Table 88 reveals that the F value obtained for influence of interaction between Technostress and Teacher Autonomy on Teacher Burnout for subsample experienced is 2.657 which is significant at 0.05 level. It indicates that there exists a significant interaction between Technostress and Teacher Autonomy on Teacher Burnout, F (4, 225) =2.657, p<.01, for subsample experienced teacher at.05 level. This means that the mean scores of Teacher Burnout for low level stress, moderate level stress and high level stress groups not vary significantly with scant autonomy, moderate autonomy and high autonomy group of the subsample experienced teacher. Therefore, the Teacher Burnout is found to be dependent of interaction between Technostress and Teacher Autonomy for experienced unaided school teachers.

In order to know the trend of interaction between A Technostress and Teacher Autonomy on Teacher Burnout for experienced teacher subsample, Profile Plot has been plotted and presented in Figure 28.

Figure 28

Profile Plot of Interaction between Technostress and Teacher Autonomy on Teacher Burnout for Experienced Teacher Subsample



Figure 28 displays that the mean scores of low level stress and moderate level stress groups belonging to scant autonomy category is higher than moderate and high autonomy groups. The mean scores of high level stress group belonging to scant autonomy is higher than high autonomy group but lower than moderate group. This indicates dependency of Teacher Burnout on the interaction between Technostress and Teacher Autonomy for subsample experienced teacher.

Multiple Regression Analysis

Multiple correlation and regression analysis has been used to know the individual and joint contributions of Technostress, Teacher Autonomy and its components in predicting Teacher Burnout of unaided secondary school teachers.

Influence of Technostress and its components and Teacher Burnout was found by conducting Pearson's product moment correlation. Correlation coefficient for Technostress and its components and Teacher Burnout were calculated and details are presented in table 91.

Table 91

Pearson's Product Moment Coefficient of Correlation for Technostress and its Components and Teacher Burnout

Variables	Emotional Exhaustion	Depersonalization	Reduced Personal Accomplishment	Teacher Burnout
Techno Insecurity	.306**	.169**	.202**	.268**
Techno Complexity	.309**	.314**	.297**	.358**
Techno Invasion	.388**	.215**	.317**	.361**
Techno Awareness	.230**	.184**	.226**	.249**
Techno Facility	.252**	.284**	.272**	.313**
Technostress	.387**	.296**	.341**	.401**

**Significant at 0.01 level

From the table 91, it is clear that Pearson's product moment coefficient of correlation between Technostress and Teacher Burnout is 0.401 (P<0.01). It means that there is a significant positive correlation between Technostress and Teacher Burnout. There is significant increase in Technostress score for increase in Teacher Burnout. Magnitude of correlation coefficient showed that there exists moderate relationship between Technostress and Teacher Burnout. Table also shows that all the dimensions of the Technostress such as Techno Insecurity (r =.268, p<.01), Techno Complexity (r =.358, p<.01), Techno Invasion (r =.361, p<.01), Techno Awareness (r =.249, p<.01) and Techno Facility (r =.313, p<.01) are positively significantly correlated with Teacher Burnout score.

Influence of Teacher Autonomy and its components and Teacher Burnout was found by conducting Pearson's product moment correlation. Correlation coefficient for Teacher Autonomy and its components and Teacher Burnout were calculated and details are presented in table 92

Table 92

Variables	Emotional Exhaustion	Depersonalization	Reduced Personal Accomplishment	Teacher Burnout
Establishing School Identity	266**	099*	216**	229**
Teaching and assessment	367**	309**	319**	389**
Parental Involvement	094*	089*	173**	136**
Professional Development	335***	120**	278**	289**
Extra Curricular Subjects	227**	157**	249**	246**
Curriculum development	136**	.032	121**	090*
Teacher Autonomy	361**	193**	332**	347**

Pearson's Product Moment Coefficient of Correlation for Teacher Autonomy and its Components and Teacher Burnout

**Significant at 0.01 level, *Significant at 0.05 level

From the table 92, it is clear that Pearson's product moment coefficient of correlation between Teacher Autonomy and Teacher Burnout is -0.347 (P<0.01). It means that there is a significant negative correlation between Teacher Autonomy and Teacher Burnout. There is significant decrease in Teacher Burnout score for increase in Teacher Autonomy score. Magnitude of correlation coefficient showed that there exists moderate relationship between Teacher Autonomy and Teacher Burnout. Table also shows that all the dimensions of the Teacher Autonomy such as establishing school identity (r = -.229, p<.01), Teaching and assessment (r = -.389, p<.01), parental involvement (r = -.246, p<.01) and Curriculum development (r = -.090, p<.05) are negatively significantly correlated with Teacher Burnout score.

Regression Analysis

Multiple regression analysis was done for Teacher Burnout with the predictor variables Teacher Autonomy and Technostress. The model summary of multiple regression analysis is exhibited in table 93

Table 93

R	R Square	Adjusted R Square
.509	.260	.257

R, R^2 and Adjusted R^2 for Model

Pearson correlation coefficient between the scores predicted by the regression model (i.e., the predicted scores) and the actual values of the dependent variable (i.e., Teacher Burnout) is 0.509 indicate a moderate strong level of association. R^2 for the overall model was 0.260 with an adjusted R^2 0.257.which indicates that 26 percent of the variation in the Teacher Burnout can be explained by the regression model developed with predictor variables Teacher Autonomy and Technostress.

The statistical significance of the overall model (i.e., the model containing all independent variables) is presented in the table 94

Table 94

Statistical Significance of the Overall Model

	Model	Sum of Squares	Df	Mean Square	F	Sig.
	Regression	22792.694	2	11396.347	00 07	001
1	Residual	65018.671	507	128.242	00.0/	.001
	Total	87811.365	509			

From the table 94, it is clear that the predicted variables Technostress and Teacher Autonomy statistically significantly predicted Teacher Burnout, F (2, 507) = 88.87, p<0.01.

Data and details of regression coefficients and individual contribution are presented in table 95

Table 95

Details of	Regression	Coefficients	and Individual	Contribution
	0	~~~		

Predictor	Unstandardized Coefficients (b)	Standardized Coefficients (β)	t	Sig.	Percentage of Individual Contribution
(Constant)	78.75		12.525	.001	
Technostress	.269	.374	9.762	.001	14.99
Teacher Autonomy	281	316	-8.233	.001	10.96

Table 95 gives t-values in each predictor variable, which shows that the bvalues obtained differ significantly from zero. Hence the variables Technostress and Teacher Autonomy are significant predictors of Teacher Burnout.

The regression model using obtained b value can be expressed as

$$Y^1 = 78.75 + .269 X_1 + -.281 X_2$$

Where Y^1 is the predicted value of Teacher Burnout, X_1 score on Technostress and X_2 score on Teacher Autonomy.

The standardized Beta values for the predictors were used to derive the equation for predicting the standardized value of Teacher Burnout.

The equation of the model is $Z^1 = .374 Z_1 + ..316 Z_2$

Where Z1 is the predicted standard score of Teacher Burnout, Z_1 and Z_2 being the standard scores of Technostress and Teacher Autonomy respectively.

From table it is clear that b value obtained for the Technostress is .269, it means that the increase in the score of Teacher Burnout is 0.269 for increase in each unit of Technostress score. A β value of 0.374 indicates that as Technostress score increases by one standard deviation, Teacher Burnout increases by 0.374 standard deviation, provided the effects of other predictor is kept constant. The individual contribution of Technostress to the model is 14.99 % (0.374 X 0.401 X 100).

b value obtained for the Teacher Autonomy is -0.281, it means that the decrease in the score of Teacher Burnout is 0.281 for increase in each unit of Teacher Autonomy score. A β value of -0.316 indicates that as Teacher Autonomy increases by one standard deviation, Teacher Burnout decreases by 0.316 standard deviation, provided the effects of other predictor is kept constant. The individual contribution of Teacher Autonomy to the model is 10.96 % (-0.316 X -0.347 X 100).

Multiple regression analysis was done for Teacher Burnout with the predictors components of Technostress and Teacher Autonomy. The details of regression analysis are given as table

The model summary of multiple regression analysis are presented in table 96
Table 96

Model	R	R Square	Adjusted R Square
1	.560	.314	.299

R, R^2 and Adjusted R^2 for Model 1

Table 97

Model	Predictor	Unstandardized Coefficients	Standardized Coefficients	t	Sig.
		(b)	(β)		-
	(Constant)	71.409		10.44	.001
	Techno Insecurity	.133	.045	.995	.320
	Techno Complexity	.672	.194	3.40	.001
	Techno Invasion	.337	.147	2.92	.004
	Techno Awareness	238	094	1.71	.087
	Techno Facility	.574	.199	3.72	.001
1	Establishing School Identity	368	124	2.64	.009
	Teaching and assessment	489	166	3.03	.003
	Parental Involvement	.289	.056	1.36	.173
	Professional Development	724	207	3.65	.001
	Extra Curricular Subjects	071	015	.306	.760
	Curriculum development	.468	.099	2.22	.027

Details of Regression Coefficients for Model 1

In this model the unstandardized regression coefficient for the components Techno Insecurity, Techno Awareness, Parental Involvement and Extra-curricular subjects are found to be not significantly different from zero as the t value obtained is less than 1.96. Hence another model 2 was tried excluding the components, the details of the regression analysis with the other predictors is given as table 97

Table 97

Model	R	R Square	Adjusted R Square
2	.553	.305	.296

Values of R, R^2 and Adjusted R^2 of Model 2

Pearson correlation coefficient between the scores predicted by the regression model (i.e., the predicted scores) and the actual values of the dependent variable (ie., academic goal orientation) is 0.553 indicate a moderate strong level of association. R^2 for the overall model was 0.305 with an adjusted R^2 0.296. This indicates that 30.5 percent of the variation in the Teacher Burnout can be explained by the regression model developed with predictor variables Techno Complexity, Techno Invasion, Techno Facility, Establishing School Identity, Teaching and assessment, Professional Development and Curriculum development. The statistical significance of the overall model (i.e., the model containing all independent variables) is presented in the table 98

Table 98

Statistical Significance of the Overall Model

Model		Sum of Squares	df	Mean Square	F	Sig.
	Regression	26816.013	7	3830.859	31.53	.001
2	Residual	60995.352	502	121.505		
	Total	87811.365	509			

From the table 98, it is clear that the predicted variables Techno Complexity, Techno Invasion, Techno Facility, Establishing School Identity, Teaching and assessment, Professional Development and Curriculum development statistically significantly predicted Teacher Burnout, F(7, 502) = 31.53, p<0.01.

Data and details of regression coefficients and individual contribution are presented in table 99

Т	a	bl	le	99
I	a	U	e	77

Model	Predictor	Unstandardized Coefficients (b)	Standardized Coefficients (β)	t	Sig.	Percentage of Individual Contribution
	(Constant)	75.26		12.36	.001	
	Techno Complexity	.606	.175	3.33	.001	6.26
	Techno Invasion	.366	.160	3.31	.001	5.78
	Techno Facility	.404	.140	3.07	.002	4.38
2	Establishing School Identity	366	123	2.84	.005	2.82
	Teaching and assessment	470	160	3.02	.003	6.22
	Professional Development	715	204	3.68	.001	5.90
	Curriculum development	.428	.091	2.04	.041	0.82

Details of Regression Coefficients and Individual Contribution

Table 99 gives t-values in each predictor variable, which shows that the bvalues obtained differ significantly from zero. Hence the components Techno Complexity, Techno Invasion, Techno Facility, Establishing School Identity, Teaching and assessment, Professional Development and Curriculum development are significant predictors of Teacher Burnout.

The regression model using obtained b value can be expressed as

$$Y^{1} = 75.26 + .606 X_{1} + .366 X_{2} + .404 X_{3} + - .366 X_{4} + - .470 X_{5} + - .715 X_{6} + .428 X_{7}$$

Where Y^1 is the predicted value of Teacher Burnout, X_1 score on Techno Complexity, X_2 score on Techno Invasion, X_3 score on Techno Facility, X_4 score on

Establishing School Identity, X₅ score on Teaching and assessment, X₆ score on Professional Development and X₇ score on Curriculum development.

The standardized Beta values for the predictors were used to derive the equation for predicting the standardized value of Teacher Burnout.

The equation of the model is

.

$$Z^{1} = .175 Z_{1} + .160 Z_{2} + .140 Z_{3} + ..123 Z_{4} + ..160 Z_{5} + ..204 Z_{6} + .091 Z_{7}$$

Where Z^1 is the predicted standard score of Teacher Burnout, Z_1 , Z_2 , Z_3 , Z_4 , Z_5 , Z_6 and Z_7 being the standard scores of Techno Complexity, Techno Invasion, Techno Facility, Establishing School Identity, Teaching and assessment, Professional Development and Curriculum development respectively.

From table 100, it is clear that b value obtained for the Techno complexity is 0.606, it means that the increase in the score of Teacher Burnout is 0.606 for increase in each unit of techno complexity score. A β value of 0.175 indicates that as techno complexity score increases by one standard deviation, Teacher Burnout increases by 0.175 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of techno complexity to the model is 6.26 %.

b value obtained for the Techno invasion is 0.366, it means that the increase in the score of Teacher Burnout is 0.366 for increase in each unit of techno invasion score. A β value of 0.160 indicates that as techno invasion score increases by one standard deviation, Teacher Burnout increases by 0.160 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of techno invasion to the model is 5.78%.

b value obtained for the Techno facility is 0.404, it means that the increase in the score of Teacher Burnout is 0.404 for increase in each unit of techno facility score. A β value of 0.140 indicates that as techno facility score increases by one standard deviation, Teacher Burnout increases by 0.140 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of techno facility to the model is 4.38%.

b value obtained for the establishing school identity is -0.366, it means that the decrease in the score of Teacher Burnout is 0.366 for increase in each unit of establishing school identity score. A β value of -0.123 indicates that as establishing school identity increases by one standard deviation, Teacher Burnout decreases by 0.123 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of establishing school identity to the model is 2.82 %.

b value obtained for the Teaching and assessment is -0.470, it means that the decrease in the score of Teacher Burnout is 0.470 for increase in each unit of Teaching and assessment score. A β value of -0.160 indicates that as Teaching and assessment increases by one standard deviation, Teacher Burnout decreases by 0.160 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of Teaching and assessment to the model is 6.22 %.

b value obtained for the professional development is -0.715, it means that the decrease in the score of Teacher Burnout is 0.715 for increase in each unit of professional development score. A β value of -0.204 indicates that as professional development increases by one standard deviation, Teacher Burnout decreases by 0.204 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of professional development to the model is 5.90 %.

$240 \ \mathrm{influence}$ of technostress & teacher autonomy on burnout

b value obtained for the Curriculum development is 0.428, it means that the increase in the score of Teacher Burnout is 0.428 for increase in each unit of Curriculum development score. A β value of 0.091 indicates that as Curriculum development score increases by one standard deviation, Teacher Burnout increases by 0.091 standard deviation, provided the effects of other predictors are kept constant. The individual contribution of Curriculum development to the model is 0.82%.



SUMMARY OF FINDINGS & CONCLUSIONS

- ✤ Study in retrospect
- ✤ Major findings of the study
- $\boldsymbol{\boldsymbol{\star}}$ Tenability of hypothesis
- ✤ Conclusions

In this chapter an overview of the vital aspects of the executing the phases of the study, the major findings of the study and their educational significance, suggestion for improving educational practices and suggestions for further research are presented in brief. This chapter is organized under the following headings

- ➢ Study in retrospect
- Major findings of the study
- Tenability of hypothesis
- ➢ Conclusions

Study in Retrospect

The diverse aspects in the different stage of the present investigations like the Title, variables, objectives, hypotheses, methodology used are viewed retrospectively.

Restatement of the Problem

The present study is planned to find out the influence of two independent variables technostress and teacher autonomy on dependent variable teacher burnout among unaided secondary school teachers. Hence the present study entitled as "Influence of Technostress and Teacher Autonomy and Burnout of Unaided Secondary School Teachers of Kerala"

Objectives of the Study

- 1. To find out the extent of technostress, teacher autonomy and burnout among unaided secondary school teachers of Kerala
- To find out whether there exists any significant difference in technostress, teacher autonomy and burnout among unaided secondary school teachers based on relevant sub groups viz. gender, locale, type of management and subject of teaching.

- To find out the main effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant sub groups.
- To find out the first order interaction effects of technostress and teacher autonomy on burnout among unaided secondary school teachers for the total sample and relevant sub groups.
- To find out the individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.
- 6. To develop regression equation to predict teacher burnout from technostress and teacher autonomy.

Hypotheses of the Study

- 1. There exists significant gender difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- 2. There exists significant locale difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- 3. There exists significant school difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers.
- 4. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on educational qualification.
- 5. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers-based on subject of teaching.

- 6. There exists significant difference in the mean scores of technostress and teacher autonomy (total score and component wise) and burnout among unaided secondary school teachers based on teaching experience.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for total sample.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school male teachers.
- There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school female teachers.
- 10. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers in rural area.
- 11. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers in urban area.
- 12. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for CBSE unaided secondary school teachers.
- 13. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for state syllabus unaided secondary school teachers.

- 14. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers with basic qualification.
- 15. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school teachers with higher qualification.
- 16. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school language teachers.
- 17. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school social science teachers.
- 18. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school science teachers.
- 19. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school mathematics teachers.
- 20. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the

dependent variable Teacher burnout for unaided secondary school novice teachers.

- 21. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school less experience teachers.
- 22. There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for unaided secondary school experienced teachers.
- 23. There is significant individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample.

Major Findings of the Study

The findings of the study are summarized as the findings of the mean difference analysis of the independent and dependent variables based on gender, locale, type of school, discipline of teaching and educational qualification. The main effects of two independent variables technostress and teacher autonomy on dependent variable teacher burnout, the first order interaction effects of two independent variable.

Mean Difference Analysis

Mean difference analysis was done to know whether there exists any significance difference between male and female, rural and urban, discipline of subject in their technostress, teacher autonomy and teacher burnout.

Difference based on Gender

- There exists significant difference in technostress (t=2.21, P<.05) and teacher burnout (t=4.24, P<.01) of male and female unaided secondary school teachers. Male teachers have higher technostress and burnout than female teachers.
- There exists no significant difference in teacher autonomy of male and female unaided secondary school teachers.
- Techno insecurity (t=2.80, P<.01), techno complexity (t=3.19, P<.01) and techno invasion (t=2.02) of male and female teachers differ significantly. Techno awareness and techno facility of male and female teachers do not differ significantly.
- There exists significant difference in establishing school identity (t=2.55, P<.01) and extracurricular subjects (t=1.97, P<.01) of male and female teachers. Teaching and assessment, parental involvement, professional development and curriculum transaction of male and female teachers do not differ significantly.
- There exists significant difference in emotional exhaustion (t=3.79, P<.01), depersonalisation (t=2.63, P<.01) and reduced personal accomplishment (t=4.44, P<.01) of male and female unaided secondary school teachers

Difference based on Locale

- There exists significant difference in teacher autonomy (t=2.77[,], P<.01) and teacher burnout (t=2.46, P<.05) of rural and urban unaided secondary school teachers. Rural teachers have higher teacher autonomy and lower burnout than urban teachers.
- There exists no significant difference in technostress of rural and urban unaided secondary school teachers.

- There exists no significant difference in techno insecurity, techno complexity, techno invasion techno awareness and techno facility of rural and urban unaided secondary school teachers.
- There exists significant difference in establishing school identity (t=2.12, P<.05), professional development (t=2.46, P<.05), extracurricular subjects (t=3.10, P<.01) and curriculum transaction (t= 2.04, P<.05) of rural and urban unaided secondary school teachers. There exists no significant difference in teaching and assessment and parental involvement of rural and urban unaided secondary school teachers.
- There exists significant difference in emotional exhaustion (t=3.18, P<.01), and reduced personal accomplishment (t=2.78, P<.01) of rural and urban unaided secondary school teachers. There exists no significant difference depersonalisation of rural and urban unaided secondary school teachers

Difference based on Type of School

- There exists no significant difference in technostress, teacher autonomy and teacher burnout of CBSE and urban State syllabus secondary school teachers.
- There exists significant difference in techno facility (t=2.69, P<.01) of CBSE and urban State syllabus secondary School teachers.
- There exists no significant difference in techno insecurity, techno complexity, techno invasion and techno awareness of CBSE and urban State syllabus secondary school teachers.
- There exists no significant difference in establishing school identity, Teaching and assessment, parental involvement, professional development, extracurricular subjects and curriculum transaction of CBSE and urban State syllabus secondary school teachers.

• There exists no significant difference in emotional exhaustion, depersonalisation and reduced personal accomplishment of CBSE and urban State syllabus secondary school teachers.

Difference based on Qualification

- There exists no significant difference in technostress, teacher autonomy and teacher burnout of teachers with basic qualification and secondary school teachers with higher qualification.
- There exists significant difference in techno complexity (t=1.98, P<.05), techno awareness (t=2.20, , P<.05) and techno facility (t=2.69, P<.01) of teachers with basic qualification and highly qualified secondary school teachers. There exists no significant difference in techno insecurity and techno invasion of teachers with basic qualification and secondary school teachers with higher qualification.
- There exists significant difference in parental involvement (t=2.81, P<.01), professional development (t=2.08, P<.05) and curriculum transaction (t=2.73, P<.01) of teachers with basic qualification and secondary school teachers with higher qualification. There exists no significant difference in establishing school identity, teaching and assessment and extracurricular subjects of teachers with basic qualification and highly qualified secondary school teachers.
- There exists no significant difference in emotional exhaustion, depersonalisation and reduced personal accomplishment of teachers with basic qualification and highly qualified secondary school teachers.

Difference based on Subject of Teaching

• There exists no significant difference in technostress, teacher autonomy and teacher burnout based on subject of teaching for total sample and component wise.

Difference based on Teaching Experience

- There exists no significant effect of experience on technostress of unaided secondary school teachers.
- There exists no significant effect in techno insecurity, techno complexity, techno invasion, techno awareness and techno facility of unaided secondary school teachers based on teaching experience.
- There exists significant effect of experience on teacher autonomy and teacher burnout of unaided secondary school teachers.
- There exists significant effect in teaching and assessment, professional development and extracurricular subjects of unaided secondary school teachers based on experience. There exists no significant effect in establishing school identity, parental involvement and curriculum transaction of unaided secondary school teachers based on experience.

Main Effect of Select Independent Variable on Teacher Burnout

Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers was calculated and the summary of the findings is presented below:

Main Effect of Select Independent Variable on Teacher Burnout for Total Sample

Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for total sample are presented below:

The *F* value obtained indicates that influence of Technostress (*F* (2,501) =50.878, p <.01), and Teacher autonomy (*F* (2,501)=29.96, p <.01) on teacher burnout is significant for total sample. There exists significant difference in the

mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for total sample.

- The comparison of mean scores of Technostress groups indicated that that mean teacher burnout score of high-level stress group (M=80.12) is higher than that of moderate (M=73.58) and low stress level groups (M= 66.56) for total sample.
- The Scheffés Post Hoc analysis shown that low level stress, moderate level stress and high level stress group differ significantly in Teacher Burnout for total sample of unaided secondary school teachers. Those unaided school teachers group who are with high level stress (M=80.12) scores high on teacher burnout than those of moderate level stress group (M=73.58) and low level stress group (M=66.56) for total sample. Those unaided secondary school teachers group who are with moderate level stress scores high on teacher burnout than those teachers of level stress group.
- The comparison of mean scores of teacher autonomy groups indicated that that mean teacher burnout score of scant autonomy group (M=78.68) is higher than that of moderate autonomy (M=74.51) and high autonomy groups (M= 67.43) for total sample.
- The Scheffés test exposed that scant autonomy, moderate autonomy and high autonomy group differ significantly in Teacher Burnout for total sample of unaided secondary school teachers. Those unaided school teachers group who are with scant autonomy (M=78.68) scores high on teacher burnout than those of moderate autonomy group (M=74.51) and high autonomy group (M=67.43) for total sample. Those unaided secondary school teachers group who are with moderate autonomy scores high on teacher burnout than those teachers of high autonomy group.

Main Effect of Select Independent Variable on Teacher Burnout for Male Subsample

Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for male sub sample are presented below:

- The F value obtained indicates that influence of Technostress (F (2,98) =10.106, p <.01), and Teacher autonomy (F (2,98) =5.426, p <.01) on teacher burnout is significant for male sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for male sub sample.
- The comparison of mean scores of Technostress groups indicated that that mean teacher burnout score of high-level stress group (M=83.02) is higher than that of moderate (M=80.07) and low stress level groups (M= 68.74) for male sub sample.
- The Scheffés test described that low level stress, moderate level stress and high-level stress group differ significantly in Teacher Burnout for male sub sample of unaided secondary school teachers. Those unaided school teachers group who are with high level stress (M=83.02) scores high on teacher burnout than those of moderate level stress group (M=80.07) and low level stress group (M=68.74) for sub sample. Those unaided secondary school teachers group who are with moderate level stress scores high on teacher burnout than those teachers of level stress group.

Main Effect of Select Independent Variable on Teacher Burnout for Female Subsample

Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for female sub sample are presented below:

Variable	F value	Level of Significance
Technostress	37.946	0.01
Teacher autonomy	27.825	0.01

- The *F* value obtained indicates that influence of Technostress (*F* (2,394) =37.946, p<.01), and Teacher autonomy (*F* (2,394) =27.825, p<.01) on teacher burnout is significant for female sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for female sub sample.
- The post hoc analysis discloses a significant difference between low and moderate, low and high, moderate and high-level technostress groups. Also, there exist significant difference between scant and moderate, scant and high, moderate and high autonomy groups for female sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Rural Subsample

Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for rural sub sample are presented below:

Variable	F value	Level of Significance
Technostress	12.914	0.01
Teacher autonomy	12.101	0.01

- The *F* value obtained indicates that influence of Technostress (*F* (2,219) =12.914, p<.01), and Teacher autonomy (*F* (2,219) =12.101, p <.01) on teacher burnout is significant for rural sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for rural sub sample.
- The post hoc analysis exposes low and moderate technostress groups do not differ significantly. There exists significant difference between low and high,

moderate and high-level technostress groups. Also, there exist significant difference between scant and high, moderate and high autonomy groups for rural sub sample. But there exists no significant difference between scant and moderate autonomy groups.

Main Effect of Select Independent Variable on Teacher Burnout for Urban Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for urban sub sample are presented below:

Variable	F value	Level of Significance
Technostress	19.734	0.01
Teacher autonomy	15.166	0.01

- The *F* value obtained indicates that influence of Technostress (*F* ,273) =19.734, p<.01), and Teacher autonomy (*F* (2,273) =15.166, p <.01) on teacher burnout is significant for urban sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for urban sub sample.
- The post hoc analysis discloses low and moderate, low and high, moderate and high-level technostress groups differ significantly. Also, there exist significant difference between scant and moderate, scant and high, moderate and high autonomy groups for urban sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for CBSE Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for CBSE sub sample are presented below:

Variable	F value	Level of Significance
Technostress	30.722	0.01
Teacher autonomy	17.555	0.01

- The *F* value obtained indicates that influence of Technostress (*F*,316) =30.722, p<.01), and Teacher autonomy (*F* (2,316) =17.555, p<.01) on teacher burnout is significant for urban sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for CBSE sub sample.
- The post hoc analysis discloses low and moderate, low and high, moderate and high-level technostress groups differ significantly. Also, there exist significant difference between scant and moderate, scant and high, moderate and high autonomy groups for CBSE sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for State Syllabus Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for State syllabus sub sample are presented below:

Variable	F value	Level of Significance
Technostress	20.837	0.01
Teacher autonomy	14.019	0.01

• The *F* value obtained indicates that influence of Technostress (*F*, 176) =20.837, p<.01), and Teacher autonomy (*F* (2,176) =14.019, p<.01) on teacher burnout is significant for urban sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for State syllabus sub sample.

• The post hoc analysis discloses that low and moderate, low and high, moderate and high-level technostress groups differ significantly. Also, there exist significant difference between moderate and high autonomy groups for State syllabus sub sample. But there exists no significant difference between scant and moderate autonomy groups.

Main Effect of Select Independent Variable on Teacher Burnout for Basic Qualification Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for basic qualification sub sample are presented below:

Variable	F value	Level of Significance
Technostress	22.288	0.01
Teacher autonomy	32.141	0.01

- The *F* value obtained indicates that influence of Technostress (*F(* 2,286) =22.288, p<.01), and Teacher autonomy (*F* (2,286) =32.141, p <.01) on teacher burnout is significant for basic qualification sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for basic qualification sub sample.
- The post hoc analysis discloses low and moderate, low and high, moderate and high-level technostress groups differ significantly. Also, there exist significant difference between scant and moderate, scant and high, moderate and high autonomy groups for basic qualification sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Higher Qualification Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for higher qualification sub sample are presented below:

Variable	F value	Level of Significance
Technostress	34.364	0.01
Teacher autonomy	2.663	NS

- The F value obtained indicates that influence of Technostress (F (2,206) =34.364, p<.01) on teacher burnout is significant for higher qualification sample. The F value obtained indicates that influence of Teacher autonomy (F (2, 206) =2.663, p>.01) on teacher burnout is not significant for higher qualification sample There exists significant difference in the mean scores of Teacher burnout for technostress groups of unaided secondary school teachers for higher qualification sub sample.
- The post hoc analysis discloses low and moderate, low and high, moderate and high-level technostress groups differ significantly. Also, there exists significant difference between scant and high autonomy. But there exists no significant difference between scant and moderate, moderate and high autonomy groups for highly qualified teacher sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Language Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for language sub sample are presented below:

Variable	F value	Level of Significance
Technostress	10.578	0.01
Teacher autonomy	6.825	0.01

- The *F* value obtained indicates that influence of Technostress (*F*,131) =10.578, p<.01), and Teacher autonomy (*F* (2,131) =6.825, p<.01) on teacher burnout is significant for language sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for language sub sample.
- The post hoc analysis discloses that low and high, moderate and high-level technostress groups differ significantly. But that there exists no significant difference between low and moderate level technostress groups. Also, there exists significant difference between scant and high autonomy. But there exists no significant difference between scant and moderate, moderate and high autonomy groups for language teacher sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Social Science Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for social science sub sample are presented below:

Variable	F value	Level of Significance
Technostress	11.704	0.01
Teacher autonomy	4.537	0.05

• The *F* value obtained indicates that influence of Technostress (*F*, 119) =11.704, p<.01), and Teacher autonomy (*F* (2,119) =4.537, p <.05) on

teacher burnout is significant for social science sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for social science sub sample

• The post hoc analysis discloses that low and high, low and moderate, moderate and high-level technostress groups differ significantly. Also, there exists significant difference between scant and high autonomy, moderate and high autonomy groups for social science teacher sub sample. But there exists no significant difference between scant and moderate level groups for social science group.

Main Effect of Select Independent Variable on Teacher Burnout for Science Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for science sub sample are presented below:

Variable	F value	Level of Significance
Technostress	16.031	0.01
Teacher autonomy	8.957	0.01

- The *F* value obtained indicates that influence of Technostress (*F*,117) =16.031, p<.01), and Teacher autonomy (*F* (2,117) =8.957, p <.01) on teacher burnout is significant for science sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for science sub sample.
- The post hoc analysis discloses that low and high, low and moderate, moderate and high-level technostress groups differ significantly. Also, there

exists significant difference between scant and high, moderate and high autonomy groups for science teacher sub sample. But there exists no significant difference between scant and moderate level groups for science group.

Main Effect of Select Independent Variable on Teacher Burnout for Mathematics Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for mathematics sub sample are presented below:

Variable	F value	Level of Significance
Technostress	12.607	0.01
Teacher autonomy	10.912	0.01

- The *F* value obtained indicates that influence of Technostress (*F*,107) =12.607, p<.01), and Teacher autonomy (*F* (2,117) =10.912, p<.01) on teacher burnout is significant for mathematics sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for mathematics subsample
- The post hoc analysis discloses that low and high, low and moderate level technostress groups differ significantly. But there exists no significant difference between moderate and high-level technostress groups. Also, there exists significant difference between scant and high, moderate and high autonomy groups for mathematics teacher sub sample. But there exists no significant difference between scant and moderate level groups for mathematics group.

Main Effect of Select Independent Variable on Teacher Burnout for Novice Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for novice sub sample are presented below:

Variable	F value	Level of Significance
Technostress	4.280	0.05
Teacher autonomy	7.953	0.01

- The *F* value obtained indicates that influence of Technostress (*F*, 93) =4.280, p<.05), and Teacher autonomy (*F* (2, 93) =7.953, p<.01) on teacher burnout is significant for novice sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for novice sub sample
- The post hoc analysis discloses that low and high, low and moderate level technostress groups differ significantly. But there exists no significant difference between moderate and high-level technostress groups. Also, there exists significant difference between scant and high autonomy, scant and moderate, moderate and high autonomy groups for novice teacher sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Less Experienced Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for less experienced sub sample are presented below:

Variable	F value	Level of Significance
Technostress	19.734	0.01
Teacher autonomy	12.811	0.01

- The *F* value obtained indicates that influence of Technostress (*F*, 165) =19.734, p<.01), and Teacher autonomy (*F* (2,165) =12.811, p <.01) on teacher burnout is significant for less experienced sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for less experienced sub sample
- The post hoc analysis discloses that low and high, low and moderate, moderate and high-level technostress groups differ significantly. Also, there exists significant difference between scant and high, moderate and high autonomy groups for less experienced teacher sub sample. But there exists no significant difference between scant and moderate level groups for less experienced sub sample.

Main Effect of Select Independent Variable on Teacher Burnout for Experienced Subsample

• Main effect of two independent variables namely, Technostress and Teacher autonomy on dependent variable, Teacher burnout of unaided secondary school teachers for experienced sub sample are presented below:

Variable	F value	Level of Significance
Technostress	25.137	0.01
Teacher autonomy	8.662	0.01

• The *F* value obtained indicates that influence of Technostress (*F*, 225) =25.137, p<.01), and Teacher autonomy (*F* (2,225) =8.662, p <.01) on

teacher burnout is significant for experienced sample. There exists significant difference in the mean scores of Teacher burnout for technostress and teacher autonomy groups of unaided secondary school teachers for experienced sub sample

• The post hoc analysis discloses that low and high, low and moderate, moderate and high-level technostress groups differ significantly. Also, there exists significant difference between scant and high, moderate and high autonomy groups for experienced teacher sub sample. But there exists no significant difference between scant and moderate level groups for experienced sub sample.

Interaction Effect of Independent Variable

Interaction effects of independent variables technostress and teacher autonomy on the dependent variable teacher burnout were estimated for total sample and relevant sub groups based on gender, locale, type of school, educational qualification, subject of teaching and teaching experience. Summary of the first order interaction effects are presented.

Interaction Effect of Independent Variable for Total Sample

The first order Interaction effects of independent variables technostress and teacher autonomy on the dependent variable teacher burnout for total sample were estimated and presented below.

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,501, F=4.70, p<0.01) is significant for total sample

Interaction Effect of Independent Variable for Male Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,98, F=4.904, p<0.01) is significant for male sub sample

Interaction Effect of Independent Variable for Female Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,394, F=4.904, p>0.05) is not significant for female sub sample

Interaction Effect of Independent Variable for CBSE Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,316, F=2.602, p<0.05) is significant for CBSE sub sample

Interaction Effect of Independent Variable for State Syllabus Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,176, F=2.073, p>0.05) is not significant for State syllabus sub sample

Interaction Effect of Independent Variable for Novice Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,93, F=2.412, p>0.05) is not significant for novice sub sample

Interaction Effect of Independent Variable for Less Experienced Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,165, F=.918, p>0.05) is not significant for less experienced sub sample

Interaction Effect of Independent Variable for Experienced Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,225, F=2.657, p<0.05) is significant for experienced sub sample

Interaction Effect of Independent Variable for Rural Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,219, F=1.529, p>0.05) is not significant for rural sub sample

Interaction Effect of Independent Variable for Urban Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,273, F=4.065, p<0.01) is significant for urban sub sample

Interaction Effect of Independent Variable for Language Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,219, F=1.529, p>0.05) is not significant for rural sub sample

Interaction Effect of Independent Variable for Social Science Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,119, F=4.748, p<0.01) is significant for social science sub sample

Interaction Effect of Independent Variable for Science Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,117, F=3.007, p<0.05) is significant for science subsample

Interaction Effect of Independent Variable for Mathematics Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,107, F=1.027, p>0.05) is not significant for mathematics sub sample

Interaction Effect of Independent Variable for basic Qualification Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,286, F=4.174, p<0.01) is significant for basic qualification sub sample

Interaction Effect of Independent Variable for Highly Qualified Subsample

• The F value obtained for the interaction of Technostress and Teacher autonomy (4,206, F=3.590, p<0.01) is significant for highly qualified sub sample

Individual and Combined Contribution of Technostress and Teacher Autonomy on Teacher Burnout

- Multiple correlation and regression analysis has been applied to find out the individual and joint contributions of Technostress and Teacher Autonomy and its components in predicting Teacher Burnout of unaided secondary school teachers.
- Pearson's product moment coefficient of correlation between Technostress and Teacher Burnout is 0.401 (P<0.01). It means that there is a significant positive correlation between Technostress and Teacher Burnout. There is significant increase in Technostress score for increase in Teacher Burnout. Magnitude of correlation coefficient showed that there exists moderate relationship between Technostress and Teacher Burnout. Study also found that all the dimensions of the Technostress are positively significantly correlated with Teacher Burnout score.
- Pearson's product moment coefficient of correlation between Teacher Autonomy and Teacher Burnout is -0.347 (P<0.01). It means that there is a significant negative correlation between Teacher Autonomy and Teacher Burnout. There is significant decrease in Teacher Burnout score for increase in Teacher Autonomy score. Magnitude of correlation coefficient showed that there exists moderate relationship between Teacher Autonomy and Teacher Burnout
- Technostress and Teacher Autonomy statistically significantly predicted Teacher Burnout, F (2,507) = 88.87, p<0.01.

Tenability of Hypotheses

The first hypothesis states "There is significant gender difference in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers." – Partially accepted.

The findings of the study disclose that male and female teachers of unaided secondary school differ significantly in their mean scores of technostress and teacher burnout for total score.

- There exists no significant difference between male and female teachers in the mean scores of teacher autonomy for total score.
- There exists significant difference between male and female teachers of unaided secondary school in their mean scores of techno insecurity, techno complexity and techno invasion.
- There exists no significant difference between male and female teachers of unaided secondary school in their mean scores of techno awareness and techno facility.
- There exists significant difference between male and female teachers of unaided secondary school in their mean scores of establishing school identity.
- There exists no significant difference between male and female teachers of unaided secondary school in their mean scores of Teaching and assessment, parental involvement, professional development, extracurricular subjects and curriculum transaction.
- There exists significant difference between male and female teachers of unaided secondary school in their mean scores of emotional exhaustion, depersonalisation and reduced personal accomplishment.

• The second hypothesis states "There is significant locale difference in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers"- partially accepted

The findings of the study disclose that rural and urban teachers of unaided secondary school differ significantly in their mean scores of teacher autonomy and teacher burnout for total score.

- There exists no significant difference between rural and urban teachers of unaided secondary school in their mean scores of technostress for total score and component wise.
- There exists significant difference between rural and urban teachers of unaided secondary school in their mean scores of establishing school identity, professional development, extra-curricular subjects and curriculum transaction.
- There exists no significant difference between rural and urban teachers of unaided secondary school in their mean scores of Teaching and assessment and parental involvement.
- The third hypothesis states "There is significant school difference in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers". The findings of the study disclose that there exists no significant difference between CBSE and State syllabus teachers of unaided secondary school in their mean scores of technostress and teacher burnout for total score and component wise. Thus, the third hypothesis is completely rejected.
- The **fourth hypothesis** states "There is significant qualification difference in the mean scores of technostress, teacher autonomy (total score and

component wise) and teacher burnout of unaided secondary school teachers". *-partially accepted.* The findings of the study disclose that basic qualified and highly qualified teachers of unaided secondary school do not differ in their mean scores of technostress and teacher burnout for total score. There exists significant difference between basic qualified and highly qualified teachers of unaided secondary school in their mean scores of techno complexity, techno awareness and techno facility. There exists significant difference between basic qualified teachers of unaided secondary school in their mean scores of parental involvement, professional development and curriculum transaction.

- The fifth hypothesis states "There is significant qualification difference in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers". The findings of the study disclose that there exists no significant difference between the subject of teaching in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers. Thus, the fifth hypothesis is completely rejected.
- The sixth hypothesis states "There is significant experience difference in the mean scores of technostress, teacher autonomy (total score and component wise) and teacher burnout of unaided secondary school teachers". There exists significant difference between the novice, less experienced and experienced teachers in the mean scores of teacher autonomy and teacher burnout. There exists no significant difference between the novice, less experienced, less experienced and experienced teachers in the mean scores of technostress. There exists significant difference between the novices, less experienced and experienced teachers in the mean scores of technostress.

experienced teachers in the mean scores of Teaching and assessment, professional development and extra-curricular subjects. *Thus, the fifth hypothesis is completely rejected.*

- The seventh hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for total sample". The results of the study disclose that the main effect of technostress and teacher autonomy on teacher burnout is significant for total sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher significant for total sample. Thus, the seventh hypothesis is completely accepted.
- The eighth hypothesis states that "there is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for male unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for male sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is significant for male sub sample. Thus, the eight hypothesis is entirely accepted.
- The ninth hypothesis states "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for female unaided secondary school teachers". The results of the study reveal that the main
effect of technostress and teacher autonomy on teacher burnout is significant for female sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for female sub sample. Thus, the ninth hypothesis is partially accepted.

- The tenth hypothesis states "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for rural unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for rural sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for rural sub sample. Thus, the tenth hypothesis is partially accepted.
- The eleventh hypothesis states "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for urban unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for urban sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is significant for urban sub sample. Thus, the eleventh hypothesis is completely accepted.

- The twelfth hypothesis states "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for CBSE unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for CBSE sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher autonomy on teacher burnout of unaided secondary school teachers is significant for CBSE sub sample. Thus, the twelfth hypothesis is completely accepted.
- The thirteenth hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for State syllabus unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for State syllabus sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for State syllabus sub sample. Thus, the thirteenth hypothesis is partially accepted.
- The **fourteenth hypothesis** states that "*There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for basic qualification unaided secondary school teachers"*. The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for basic qualification sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher burnout of unaided

secondary school teachers is significant for basic qualification sub sample. Thus, the fourteenth hypothesis is entirely accepted.

- The **fifteenth hypothesis** states that "*There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for highly qualified unaided secondary school teachers*". The study reveals that the main effect of technostress on teacher burnout is significant for highly qualified sub sample of unaided secondary school teachers. It also reveals the main effect of teacher burnout on teacher burnout is not significant for highly qualified sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is significant for highly qualified sub sample. Thus, the fifteenth hypothesis is partially accepted.
- The sixteenth hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for language unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for language sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for language sub sample. Thus, the sixteenth hypothesis is partially accepted.
- The seventeenth hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for social science unaided secondary school teachers". The results of the study reveal

that the main effect of technostress and teacher autonomy on teacher burnout is significant for social science sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers significant for social science sub sample. Thus, the seventeenth hypothesis is entirely accepted.

- The eighteenth hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for science unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for science sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is significant for science sub sample. Thus, the eighteenth hypothesis is totally accepted.
- The nineteenth hypothesis states "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for mathematics unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for mathematics sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for mathematics sub sample. Thus, the nineteenth hypothesis is partially accepted.
- The **twentieth hypothesis** states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress

and Teacher autonomy on the dependent variable Teacher burnout for novice unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for novice sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for novice sub sample. Thus, the twentieth hypothesis is partially accepted.

- The twenty first hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for less experienced unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for less experienced sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is not significant for less experienced sub sample. Thus, the twenty first hypothesis is partially accepted.
- The twenty second hypothesis states that "There is significant main effect and interaction effect of each of the independent variable namely, Technostress and Teacher autonomy on the dependent variable Teacher burnout for experienced unaided secondary school teachers". The results of the study reveal that the main effect of technostress and teacher autonomy on teacher burnout is significant for experienced sub sample of unaided secondary school teachers. In the case of interaction effects, the interaction effect of technostress and teacher autonomy on teacher burnout of unaided secondary school teachers is significant for experienced sub sample. Thus, the twenty second hypothesis is completely accepted.

• The twenty third hypothesis states "There is significant individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary school teachers of Kerala for total sample". The result reveals that there is significant individual and combined contribution of technostress and teacher autonomy on burnout among unaided secondary teachers of Kerala.

The results of the study reveal that there is significant individual and joint contribution of Technostress, Teacher Autonomy on teacher burnout of unaided secondary school teachers.

Conclusions

The major objectives of the study were to find out the significant difference in technostress, teacher autonomy and burnout of unaided teachers, to find out the main and interaction effects of technostress, teacher autonomy on burnout among unaided teachers, and to develop regression equation to predict teacher burnout from technostress and teacher autonomy.

The results concludes that unaided secondary school teachers possess above average levels of technostress, techno insecurity, techno complexity, and techno invasion, but below average levels of techno awareness and techno facility. Additionally, unaided teachers have above average levels of teacher autonomy and above average level of autonomy in several areas, such as, autonomy for establishing school identity, autonomy for teaching and assessment, autonomy for professional development, autonomy for extra-curricular subjects, and curriculum development autonomy. However, they possess below average levels of autonomy for parental involvement. Furthermore, unaided teachers experience below average levels of teacher burnout, emotional exhaustion, depersonalization, and reduced personal accomplishment. Overall, the findings indicate that unaided teachers face

challenges in their use of technology and autonomy in certain aspect of their profession, and also experience negative effects on their well-being.

The results of this study indicate that technostress, teacher burnout, techno insecurity, techno complexity, and techno invasion differ significantly between male and female unaided secondary school teachers, with male teachers having higher levels of these challenges than female teachers. However, these findings contrast with the results of previous studies, such as Coklar et al. (2016), which found that general technostress levels did not vary based on gender and Syvanen et al. (2016) which found that male teachers were less stressed than female teachers. Furthermore, the study found that techno facility stress differs significantly between CBSE and urban State syllabus secondary school teachers, with State syllabus teachers having higher levels of techno facility stress, indicating a lack of technological facility in state syllabus schools. Additionally, the study revealed that there is a significant difference in techno complexity and techno awareness between teachers with basic qualifications and highly qualified secondary school teachers, with teachers with basic qualifications having higher levels of techno complexity and techno awareness issues. However, there was no significant effect of experience on technostress of unaided secondary school teachers, which differs from the findings of Syvanen et al. (2016) and Tarafdar (2014) Overall, the findings of this study should be interpreted with caution due to the conflicting results with previous studies and highlight the need for further research on the topic.

The results of the study indicate that there is no significant difference in technostress, teacher autonomy, and teacher burnout between CBSE and State syllabus secondary school teachers. Additionally, there is no significant difference in these factors based on the teachers' qualifications or subject of teaching. However, the study found that there is no significant effect of experience on technostress of unaided secondary school teachers, but there is a significant effect of experience on teacher autonomy and teacher burnout. The study also shows that there is a significant influence of technostress and teacher autonomy on teacher burnout, with high levels of technostress being associated with high levels of teacher burnout. This aligns with previous research, such as Tarafdar et al. (2015) which found that high levels of technostress and teacher burnout are positively correlated. Furthermore, the study found that teacher autonomy is negatively correlated with burnout in all its dimensions, such as emotional exhaustion, depersonalization, and reduced personal accomplishment. This is consistent with recent studies that have found that teacher autonomy is positively correlated with job satisfaction, empowerment, engagement, and professionalism, and negatively correlated to emotional exhaustion (Erss et al., 2016; Wermke et al., 2019; Skaalvik & Skaalvik, 2014; Wilches, 2007).

The results of the study indicate that the influence of technostress and teacher autonomy on teacher burnout is significant for the total sample, male sample, female sample, rural sample, urban sample, teachers with basic qualifications, higher qualifications, language teachers, social science teachers, science teachers, mathematics teachers, novice teachers, less experienced teachers, and experienced teachers. This suggests that the relationship between technostress, teacher autonomy, and teacher burnout is consistent across different subgroups of unaided secondary school teachers. The study supports the findings of previous research that states that high levels of technostress and low levels of teacher autonomy are associated with high levels of teacher burnout.

The results of the study indicate that the interaction effect of technostress and teacher autonomy on teacher burnout is significant for the total sample, male sample, CBSE sample, experienced sample, and urban sample. The study also found that the interaction effect is significant for social science sample, science sample and basic qualification sample and highly qualified sample. However, the interaction effect was not significant for the female sample, State syllabus sample, novice

sample, less experienced sample, rural sample, language sample, and mathematics sample. These findings suggest that the relationship between technostress, teacher autonomy, and teacher burnout is complex and may vary depending on the subgroup of teachers being considered. The study supports the findings of previous research that states that high levels of technostress and low levels of teacher autonomy are associated with high levels of teacher burnout, but this relationship may be moderated by factors such as gender, qualification, subject of teaching, and experience. The study emphasizes that reducing technostress and increasing teacher autonomy can help to reduce teacher burnout, but the specific strategies used may need to be tailored to the unique needs of different subgroups of teachers.

The results of the study indicate that there is a significant positive correlation between technostress and teacher burnout, with an increase in technostress leading to an increase in teacher burnout. Additionally, there is a significant negative correlation between teacher autonomy and teacher burnout, with an increase in teacher autonomy leading to a decrease in teacher burnout. The study found that the dimensions of technostress are positively and significantly correlated with teacher burnout. Furthermore, multiple correlation and regression analysis were applied to find out the individual and joint contributions of technostress and teacher autonomy in predicting teacher burnout of unaided secondary school teachers. The results show that technostress and teacher autonomy statistically significantly predicted teacher burnout. These findings suggest that reducing technostress and increasing teacher autonomy can help to reduce teacher burnout among unaided secondary school teachers.



SUGGESTIONS & RECOMMENDATIONS

Ð

- ✤ Recommendations
- ✤ Educational Implications
- Suggestions for Further
 Studies

Suggestions and recommendations are the vital part of every research. It is the part where the contribution of the research is summarized. Every research in the discipline of education should have some educational implications which may help to make our education system to a more competitive and productive one. The recommendations presented in this chapter are intended to serve as a guide for schools and educational leaders as they work to address the challenges of technostress, autonomy, and burnout and promote teacher well-being in the age of technology.

Recommendations

The present study has several recommendations in different dimensions of teachers and teaching. To mitigate the negative effects of technostress and promote teacher autonomy in unaided secondary schools, the following recommendations can be considered:

- Provide teachers with training and support in the use of technology: Teachers should be given training on how to use technology effectively in their teaching and provided with ongoing support as needed. This will help to reduce feelings of technostress and increase teachers' confidence in using technology.
- Encourage collaboration among teachers: Teachers should be encouraged to collaborate with one another to share ideas and strategies for using technology in their teaching. This can help to reduce feelings of isolation and increase job satisfaction.
- Foster a culture of autonomy: Schools should respect and support teachers' autonomy in decision-making related to their teaching and professional development. This can help to promote job satisfaction and reduce feelings of burnout.

- Encourage balance between technology use and face-to-face interaction: While technology can be a useful tool in teaching, it should not be overused. Teachers should be encouraged to balance the use of technology with faceto-face interaction with their students.
- Provide opportunities for self-care: Teachers should be encouraged to prioritize their own well-being and provided with opportunities for self-care.
- Monitor and review teacher's workload and expectations: It is important to ensure that teachers are not overwhelmed by unrealistic workloads or expectations, and that the technology is used to enhance their teaching and not creating additional workload.
- Provide mentorship and supervision: Mentorship and supervision can provide teachers with support and guidance as they navigate the challenges of using technology in their teaching.

By implementing these recommendations, schools can help to reduce the negative effects of technostress and promote teacher autonomy, which can in turn lead to reduced burnout among unaided secondary school teachers.

Educational Implications

The present study has numerous implications in various dimensions of teachers and teaching. The findings of the present study would support in throwing light into certain ignored fields like education, daily requirements of teachers in relation to their teaching environment and all their profession. Based on the results, it appears that technostress and a lack of teacher autonomy can contribute to burnout among unaided secondary school teachers. Technostress, which is the psychological response to the use of technology in the workplace, can lead to feelings of frustration, overload, and a lack of control. A lack of teacher autonomy may lead to feelings of powerlessness and dissatisfaction. Together, these factors can contribute

to burnout, which is characterized by emotional exhaustion, depersonalization, and a reduced sense of personal accomplishment. It is important for schools and educational organizations to address these issues in order to support the well-being and effectiveness of their teachers.

• The result of the study shows teachers possess above average level of technostress. So steps should be initiated to reduce the technostress among teachers.

Techniques to Reduce Technostress

Techniques can be categorized as general techniques and specific techniques. General techniques can be used by different types of people regardless of age, work, gender, region and education. Specific techniques are those strategies which are intended for each profession like software engineers, office workers, teachers and librarians.

Providing teachers training and professional development opportunities to help them effectively integrate technology into their instruction help them to manage technostress.

Hogge (2006) examined that "to minimize technostress, some coping strategies should be formulated and implemented. The main factor which gives rise to the problem of technostress is the improper implementation of technological innovations. To prevent and manage technostress, teachers need to be provided with adequate training, support, and resources to effectively integrate technology into their classrooms, and to be able to disconnect and balance their online and offline activities.

To reduce technostress the following strategies are suggested

• *Techno Club*. Create a platform for the teachers, which gives proper technological support programme with the collaboration of IT department.

- *KITE*. Kerala infrastructure and technology for education (KITE) is a government of Kerala initiative established to endorse and implement novice technology in the state of Kerala. Its concern only concentrated in government sector. It should be extended in the unaided sector also.
- *KOOL.* KITE's Open Online Learning (KOOL) is online training programme for teachers to improve their technical skills. It should extend to unaided teachers too.
- *Positive Technology*. Make the teachers aware about the positive technology, which is the one of the general tactics to overcome the technostress and it "is the scientific and applied approach to the use of technology for improving the quality of our experience" (Riva et al., 2012). Positive technology affects all three domains of personal experience. First one is Hedonic- It is the way of using technology for positive experience. It help reduce techno anxiety and technostress. It also helps to avoid techno overload by choosing right work device and time. Second one is Eudaimonic- it is a Greek word which means state or condition of 'good spirit 'and usually denotes as happiness or welfare. According to Aristotle the term Eudaimonic means highest human virtuous. Third one is social/interpersonal domain.
- *Digital Wellbeing*. Digital wellbeing is another strategy to overcome issue of technostress. Give appropriate training in preservice and in-service period to make the teachers a good digital wellbeing personalities. Digital wellness (digital wellbeing) is the practise of technology to ensure the physical and emotional health of a person. Although workers rely on their digital tools and the internet to do their tasks, the goal of digital wellness is to design technology in such a way that it indorses healthy use and dynamically supports the handler to keep a healthy lifestyle. A common instance of technology that can have an undesirable consequence on workers is the misuse of social media throughout the workday.

- *Digital Detox*. Train the teachers digital detox which means get available gadget free time to one's life. It is most popular technique to reduce technostress specifically issue of techno invasion. Other methods to overcome technostress are identify the signs, be purposeful, in another word keep objectivity in every action, stay away from distractions; this is also known as issue of multiscreen, Controlled use, Prepare a not to list, be elective and specific, update the changes, Become more mindful of your interactions, Switch off notifications from social media apps and other non-essential apps during working hours, Avoid using work-related tools/software during your leisure time or with family and friends, Introduce technological innovations.
- *Techno-eustress.* Inculcate the idea of techno-eustress among teachers. It is a, " positive technique to overcome technostress that causes satisfaction, joy, increases vitality, does not cause imbalances and helps facilitate people's decision making" (Tarafdar et al., 2019). Eustress, happens mainly as a results from positive interpretations of stressors (Le Fevere et al., 2003)
- *Make the Teachers as Technosavy*. Technosavy means a person who is well informed about or proficient in the use of modern technology. Make the teachers as technosavy persons which help them to overcome issue of technostress to some extent. "IT professionals reduce the technostress by understanding the functionalities of the technology and training vigorously." (Scott & Timmerman, 2005)
- *Make the teachers as Digital native*. Students of the 21st century known as digital native whereas the teachers are migrant into the digital world. So it is very important that make the teachers as digital native persons which help them to overcome issue of technostress to some extent.
- *Techno Ethics*. Make the teachers aware of techno ethics which is very essential when we deal with modern technology, which may secure us different types of technological issues.

$284\,$ influence of technostress & teacher autonomy on burnout

Other Ways of Coping with Techno Stress

There are several techniques that can help to alleviate technostress:

- *Establish Boundaries*. Set specific times during the day to check and respond to emails, messages and other digital communications. Turn off notifications that are not urgent, and avoid checking your phone or computer outside of designated times.
- *Prioritize Self-care.* Make sure to take regular breaks, get enough sleep, exercise, and eat healthy.
- *Unplug.* Take a complete break from technology at least once a week to give your mind and body a rest.
- *Practice Mindfulness*. Mindfulness techniques such as meditation and deep breathing can help to reduce stress and anxiety.
- *Keep a Balance.* Don't let technology take over your life. Make sure you have enough time for activities and people that are important to you.
- *Seek Professional Help.* If technostress is affecting your work or personal life, seek help from a mental health professional.
- *Manage Your Time.* Plan your day and set up reminders to stay focused and on task.
- *Learn to Say No.* Saying no to non-important task can help in reducing the workload which can cause technostress.
- Use Technology to Your Advantage. Use technology to organize your schedule and manage your time efficiently, so you don't feel overwhelmed by it.
- Get adequate, user friendly software
- Create better communication within the environment
- Create a level of reassurance, patience, and stability within the environment
- Maintain an ever-present system of training and education to new and old technologies

- Foster sharing of computer related knowledge within the organization.
- A responsive and easily reached help-desk can allay managers' anxiety and concerns, guide them in using and familiarizing with new applications and assure them in case of problems
- Keep teachers "involved" in the general scheme of things in the context of new computer systems. The more involved and familiar they are, the less techno stressed they would be.
- Encourage people to "experiment" and innovate in the context of technology use
- Encourage teachers to communicate, discuss, and share their knowledge about new technologies.
- To address techno insecurity, it's important to educate individuals about online security and privacy, as well as the steps they can take to protect their personal information. This can include using strong passwords, avoiding public Wi-Fi networks, and being mindful of the information shared online. Additionally, organizations can implement security measures such as encryption and multi-factor authentication to protect sensitive information and provide peace of mind for users.
- To address techno complexity, it's important to provide clear instructions, documentation and training that can help people understand how to use technology effectively.
- Teachers with basic qualifications have higher levels of techno complexity than teachers with higher qualifications. So give proper directions and motivations to gain higher qualifications.
- To address techno invasion, it's important for individuals to set boundaries and establish a healthy relationship with technology. This can include setting specific times for checking email or social media, and making sure to take breaks and engage in activities that do not involve technology. Additionally,

organizations can implement policies and practices that promote a healthy work-life balance, such as not expecting teachers to be available outside of normal working hours.

 The study indicates a lack of technological facilities in state syllabus unaided schools. So government take the initiative to make available the facilities.

Techniques to Strengthen Autonomy

Findings of the study highlight the importance of improving autonomy of teachers. Autonomy could be increased through various types of training and programmes.

Teacher autonomy refers to the degree of control teachers have over their own work. Empowering teachers with autonomy can lead to increased job satisfaction, motivation, and commitment, as well as improved student outcomes. Here are some strategies that can help to empower teacher autonomy:

- *Encourage Professional Development*. Provide teachers with opportunities for professional development and training to enhance their skills and knowledge, and to stay current with new teaching methods and technologies.
- *Provide Autonomy in the Classroom*. Give teachers the freedom to design and implement their own lesson plans, assessments, and classroom management strategies.
- *Encourage Collaboration and Teamwork*. Create opportunities for teachers to work together, share ideas and best practices, and provide constructive feedback to one another.
- *Emphasize the Importance of Teacher Input*. Encourage teachers to share their thoughts, ideas and concerns, and take them into consideration when making decisions that affect them and their students.
- *Provide Resources and Support*. Provide teachers with the necessary resources and support they need to be effective in their work, such as access to technology, instructional materials and professional development opportunities.

- *Encourage Creativity and Innovation*. Create an environment that encourages teachers to think outside the box and to be creative and innovative in their teaching methods.
- *Provide Opportunities for Leadership*. Give teachers the opportunity to take on leadership roles and responsibilities, such as mentoring new teachers or leading professional development workshops.
- *Recognize and Reward Good Work*. Recognize and reward teachers for their hard work, dedication and achievements, which will help to build their confidence and motivation.
- *Teachers' Hub*. Form the teachers hub, which lead to conduct different types of faculty development programmes to strengthen autonomy of teachers and reduce burnout.
- *Make the Teachers Enthusiastic*. "Teacher Enthusiasm, also positively associated with teachers' overall work gratification, and low levels of emotional exhaustion" (Kunter et al., 2008, 2011, 2013). In addition, Cobb and Foeller (1992) also observed positive relations between teacher enthusiasm and autonomy.
- Educational Policies. Educational policies should aid the autonomy of teachers as indicated by Li & Allen "As teacher autonomy relates positively to teacher job satisfaction and retention, this study suggests that policy makers consider a moralistic approach in creating educational policies" (Li & Allen, 2021).
- *Implement Guidelines*. The study shows the issue of excess parental involvement in school. Implement guidelines to avoid unnecessary interference of parents.

By implementing these strategies, schools and educational organizations can empower teachers with autonomy and support them in becoming more effective and satisfied in their work.

Strategies to Overcome Teacher Burnout

Here are some strategies that can help teachers overcome burnout

- *Conduct Teachers SEL Programmes*. Social Emotional Learning programmes would help to overcome the issues of burnout among teachers.
- Increasing teacher's autonomy would help to reduce the burnout of them.
- Comprehensive policy to protect teachers' right.
- Prioritize Self-care. Make sure to take regular breaks, get enough sleep, exercise, and eat healthy.
- *Establish Boundaries*. Set specific times during the day to disconnect from work and focus on your personal life and hobbies.
- Seek Support. Connect with colleagues and other teachers to share ideas and strategies for dealing with stress.
- Practice Mindfulness. Mindfulness techniques such as meditation and deep breathing can help to reduce stress and anxiety.
- *Take Time Off.* Take a leave of absence if necessary, to recharge and come back with renewed energy.
- Manage Your Time. Plan your day and set up reminders to stay focused and on task.
- *Get Professional Help*. If burnout is affecting your work or personal life, seek help from a mental health professional.
- Reflect on Your Goals and Values. Reflecting on your goals, values and why you choose teaching as a profession can help to keep things in perspective and refocus on what is important.

- Make Use of Available Resources. Make use of any programs, resources and support available through your school district or professional organizations to help with managing stress.
- Learn to Delegate. Learn to delegate tasks and responsibilities to other colleagues, students or family members.
- Practice Emotion Regulation. In observing ways to control teacher burnout, scholars have recognized emotion regulation and coping as effective means to reduce the emotional exhaustion associated with burnout (Brackett et al., 2010; Chang, 2013; Durr et al., 2014).

Earlier studies has observed that teachers use a diversity of stratagems to adjust their emotions while schooling, comprising cognitive reappraisals, which have been recognized as protective against burnout (Sutton et al., 2009; Yin et al., 2016; Taxer & Gross, 2018; Chang & Taxer, 2021). Some other suggestions are given below.

- Increasing teacher autonomy by involving teachers in decision-making processes related to technology integration in the classroom.
- Providing teachers with access to resources and support, such as technical assistance and mental health services, to help them manage the demands of technology integration and prevent burnout.
- Developing and implementing policies and practices that support teachers' well-being and work-life balance, such as reducing workload, providing flexible scheduling, and promoting self-care.
- Encouraging collaboration and community building among teachers to foster a sense of support and camaraderie.
- Encourage regular monitoring and evaluation of the impact of technology integration on teachers' well-being and make necessary adjustments.

By implementing these strategies, teachers can reduce the risk of burnout and improve their overall well-being. It's important to keep in mind that burnout is a

gradual process, so it's important to be mindful of the signs and take steps to prevent it before it becomes a more serious problem.

Suggestions for Further Studies

Succeeding the completion of the study, the investigator recognized certain areas that have need of further research, which are listed below as suggestions for upcoming research.

- Further study can be conducted on development and standardisation of digital wellbeing package for teachers.
- Replication of the study among the primary, higher secondary and college teachers.
- Replication of the study among the government and aided sectors.
- Further studies can be carried out to identify the factors affecting technostress of teachers.
- Further research can be conducted to explore the impact of technology integration on the well-being of teachers in different contexts and cultures.
- Further research can be conducted to identify the factors affecting technostress of students of different stages.
- Further study can be conducted to evaluate the moderating effects of coping strategies on the relationship between technostress and burnout among unaided secondary school teachers in Kerala.
- Conduct a qualitative study to gain a deeper understanding of the experiences and perceptions of unaided secondary school teachers in Kerala regarding technostress and burnout.
- Investigate the specific role of school culture and leadership in influencing technostress and burnout among unaided secondary school teachers in Kerala.
- Conduct a study to compare the level of technostress and burnout among unaided secondary school teachers in Kerala with those in other states or countries.

REFERENCES

- Agut, S., Grau, R., & Salanova, M. (2001). Technostress and burnout among Spanish workers: Gender differences. *Occupational Health Psychology: Europe*, 28-31.
- Al-Fudail, M., & Mellar, H. (2008). Investigating teacher stress when using technology. *Computers & Education*, 51(3), 1103-1110.
- Al-Shami, A. (2008). Coping with technology. Slideshare Presentations. http://www.slide share.net/alshami/techno-stress?src=related_normal&rel= 1312523Â. Accessed January 17, 2010.
- Anderson, L. W. (1987). The decline of teacher autonomy: Tears or cheers?. *International Review of Education*, *33*(3), 357-373.
- Antoniou, A. S., Ploumpi, A., & Ntalla, M. (2013). Occupational stress and professional burnout in teachers of primary and secondary education: The role of coping strategies. *Psychology*, 4(03), 349.
- Barfield, A., Ashwell, T., Carroll, M., Collins, K., Cowie, N., Critchley, M., Head, E., Nix, M., Obermeier, A., & Robertson, M. C. (forthcoming). Exploring and defining teacher autonomy: A collaborative discussion. In *On Developing Autonomy:* Proceedings of the 2001 CUE Conference, Shizuoka, Japan. Tokyo: JALT.
- Bédard, M. (2015). *Pedagogical autonomy and accountability: A recipe for improving academic results*. Montreal Economic Institute.
- Bell, B. R. (2016). *Technostress from the perspective of college students: A basic qualitative study* [Doctoral Dissertation, Capella University].
- Benson, P. (1996). Concepts of autonomy in language learning. In R. Pemberton, E. Li, W. Or, & H. Pierson (Eds.), Taking control. *Autonomy in Language Learning*, 27-34.
- Benson, P. (2010). Teacher education and teacher autonomy: Creating spaces for experimentation in secondary school English language teaching. *Language Teaching Research*, 14(3), 259-275.
- Bianchi, R., Schonfeld, I. S., & Laurent, E. (2014). Is burnout a depressive disorder? A reexamination with special focus on atypical depression. *International Journal of Stress Management*, 21(4), 307.
- Bondanini, G., Giorgi, G., Ariza-Montes, A., Vega-Muñoz, A., & Andreucci-Annunziata, P. (2020). Technostress dark side of technology in the workplace: A scientometric analysis. *International Journal of Environmental Research and Public Health*, 17(21), 8013.
- Booker, Q. E., Rebman Jr, C. M., & Kitchens, F. L. (2014). A model for testing technostress in the online education environment: An exploratory study. *Issues in Information Systems*, 15(2).
- Brackett, M. A., Palomera, R., Mojsa-Kaja, J., Reyes, M. R., & Salovey, P. (2010). Emotion-regulation ability, burnout, and job satisfaction among British secondary-school teachers. *Psychology in the Schools*, 47(4), 406-417.

- Brillhart, P. E. (2004). Technostress in the workplace: Managing stress in the electronic workplace. *Journal of American Academy of Business*, 5(1/2), 302-307.
- Brod, C. (1982). Managing technostress: Optimizing the use of computer technology. *Personnel Journal*, 61(10), 753-57.
- Brod, C. (1984). Technostress: The human cost of the computer revolution. Addison-Wesley
- Brouwers, A., & Tomic, W. (1999). Teacher burnout, perceived self-efficacy in classroom management, and student disruptive behaviour in secondary education. *Curriculum and Teaching*, 14(2), 7-26.
- Brouwers, A., & Tomic, W. (2000). A longitudinal study of teacher burnout and perceived self-efficacy in classroom management. *Teaching and Teacher Education*, 16(2), 239-253.
- Brunetti, G. J. (2001). Why do they teach? A study of job satisfaction among long-term high school teachers. *Teacher Education Quarterly*, 28(3), 49-74.
- Cano-García, F. J., Padilla-Muñoz, E. M., & Carrasco-Ortiz, M. Á. (2005). Personality and contextual variables in teacher burnout. *Personality and Individual Differences*, 38(4), 929-940.
- Carlgren, I., & Klette, K. (2008). Reconstructions of Nordic Teachers: Reform policies and teachers' work during the 1990s. Scandinavian Journal of Educational Research, 52(2), 117-133
- Caro, D. H., & Sethi, A. S. (1985). Strategic management of technostress. Journal of Medical Systems, 9(5), 291-304.
- Castle, K. & Aichele, D. (1994). Professional development and teacher autonomy. In D. Aichele & A. Coxford (Eds.), *Professional development for teachers of mathematics (1994 Yearbook)*. National Council of Teachers of Mathematics. p. 3.
- Champion, S. (1988). Technostress: Technology's toll. *School Library Journal*, 35(3), 48-51.
- Chang, M. L. (2009). An appraisal perspective of teacher burnout: Examining the emotional work of teachers. *Educational Psychology Review*, 21(3), 193-218.
- Chang, M. L. (2013). Toward a theoretical model to understand teacher emotions and teacher burnout in the context of student misbehavior: Appraisal, regulation and coping. *Motivation and Emotion*, 37(4), 799-817.
- Chang, M. L., & Taxer, J. (2021). Teacher emotion regulation strategies in response to classroom misbehavior. *Teachers and Teaching*, 27(5), 353-369.
- Charters Jr, W. W. (1976). Sense of teacher work autonomy: Measurement & findings. CEPM Publications.
- Cherniss, C., & Cherniss, C. (1980). *Staff burnout: Job stress in the human services*. Sage Publications.
- Christian, M., Purwanto, E., & Wibowo, S. (2020). Technostress creators on teaching performance of private universities in Jakarta during Covid-19 pandemic. *Technology Reports of Kansai University*, 62(6), 2799-2809.

- Chua, S. L., Chen, D. T., & Wong, A. F. (1999). Computer anxiety and its correlates: a meta-analysis. *Computers in Human Behavior*, 15(5), 609-623.
- Clark, K., & Kalin, S. (1996). Technostressed out? How to cope in the digital age. *Library Journal*, *121*(13), 30-32.
- Clouse, R. W. (1983). Clouse-Whitaker Career Attitude Inventory. In Stress and Burnout in the Schools (pp. A1- 10). National School Resource Network.
- Cobb, S. L., & Foeller, W. H. (1992). An organizational behavior analysis of teacher attitudes about teaching high school economics. *Theory & Research in Social Education*, 20(4), 421-439.
- Çoklar, A. N., & Sahin, Y. L. (2011). Technostress levels of social network users based on ICTs in Turkey. *European Journal of Social Sciences*, 23(2), 171-182.
- Çoklar, A. N., Efilti, E., & Sahin, L. (2017). Defining teachers' technostress levels: A scale development. Online Submission, 8(21), 28-41.
- Çoklar, A. N., Efilti, E., Sahin, Y. L., & Akçay, A. (2016). Investigation of techno-stress levels of teachers who were included in technology integration processes. *Online Submission*.
- Çoklar, A., Efilti, E., Şahin, Y., & Akçay, A. (2016). Determining the reasons of technostress experienced by teachers: A qualitative study. *Turkish Online Journal of Qualitative Inquiry*, 7(2), 71-96.
- Collie, R. J. (2021). COVID-19 and teachers' somatic burden, stress, and emotional exhaustion: Examining the role of principal leadership and workplace buoyancy. *Aera Open*, 7, 2332858420986187.
- Collie, R. J., Granziera, H., & Martin, A. J. (2018). Teachers' perceived autonomy support and adaptability: An investigation employing the job demands-resources model as relevant to workplace exhaustion, disengagement, and commitment. *Teaching and Teacher Education*, 74, 125-136.
- Cunningham, W. G. (1983). Teacher burnout-Solutions for the 1980s: A review of the literature. *The Urban Review*, 15(1), 37-51.
- Daley, M. R. (1979). 'Burnout': Smoldering problem in protective services. Social Work, 24(5), 375-379.
- Davis, F. D. (1989). Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quarterly*, 319-340.
- Davis-Millis, N. (1998, February). Technostress and the organization: A manager's guide to survival in the information age. In 67th Annual Meeting of the Music Library Association (Vol. 14).
- Deci, E. L., & Ryan, R. M. (1995). Human autonomy. In *Efficacy, agency, and self-esteem* (pp. 31-49). Springer, Boston, MA.
- Deci, E. L., Ryan, R. M., & Kernis, M. H. (1995). Efficacy, agency, and selfesteem. *Human Autonomy: the basis for true self-esteem*, 31-39.

- Diller, H., Jeffrey, S., & Fiedler, M. (2016). Searching for the silver linings of technoinvasion, *Passauer Diskussionspapiere - Betriebswirtschaftliche Reihe, No. B-22-*16, Universität Passau, Wirtschaftswissenschaftliche Fakultät, Passau.
- Dong, Y., Xu, C., Chai, C. S., & Zhai, X. (2019). Exploring the structural relationship among teachers' technostress, technological pedagogical content knowledge (TPACK), computer self-efficacy and school support. *The Asia-Pacific Education Researcher*, 1.
- Drake, M. A. (2000). Technological Innovation and Organizational Change Revisited. *Journal of Academic Librarianship*, 26(1), 53-59.
- DuFour, R. (1999). Teacher autonomy in the professional learning community. *Journal of Staff Development*, 20, 69-70.
- Durndell, A., & Haag, Z. (2002). Computer self efficacy, computer anxiety, attitudes towards the Internet and reported experience with the Internet, by gender, in an East European sample. *Computers in Human Behavior*, 18(5), 521-535.
- Elo, J., & Nygren-Landgärds, C. (2021). Teachers' perceptions of autonomy in the tensions between a subject focus and a cross-curricular school profile: A case study of a Finnish upper secondary school. *Journal of Educational Change*, 22(3), 423-445.
- Erss, M., Kalmus, V., & Autio, T. H. (2016). Walking a fine line: Teachers' perception of curricular autonomy in Estonia, Finland and Germany. *Journal of Curriculum Studies, 48,* 589–609. https://doi.org/10.1080/00220272.2016. 1167960
- Esfandiari, R., & Kamali, M. (2016). On the relationship between job satisfaction, teacher burnout, and teacher autonomy. *Iranian Journal of Applied Language Studies*, 8(2), 73-98.
- Eskridge, D. H., & Coker, D. R. (1985). Teacher stress: Symptoms, causes, and management techniques. *The Clearing House: A Journal of Educational Strategies, Issues and Ideas*, 58(9), 387-390.
- Farber, B. A. (1984). Teacher burnout: Assumptions, myths, and issues. *Teachers College Record*, 86(2), 321-338.
- Finch, A. (2002). Autonomy: Where are we? Where are we going? *JALT CUE-SIG Proceedings*, 15-42.
- Fisher, D. (1996). Technostress and the librarian: A critical discussion. *Education Libraries Journal*, *39*, 9-14.
- Freudenberger, H. J. (1974). Staff burn-out. Journal of Social Issues, 30(1), 159-165.
- Freudenberger, H. J. (1977). Burn-out: Occupational hazard of the child care worker. Child Care Quarterly, 6(2), 90–99. https://doi.org/10.1007/BF01554695
- Freudenberger, H. J. (1980). Burnout: how to beat the high cost of success. Bantam Books.
- Friedman, I. A. (1999). Teacher-perceived work autonomy: The concept and its measurement. *Educational and Psychological Measurement*, 59(1), 58-76.

- Frostenson, M. (2015). Three forms of professional autonomy: de-professionalisation of teachers in a new light. *Nordic Journal of Studies in Educational Policy*, *2*, 28464.
- Glicken, M. D. (1983). A counseling approach to employee burnout. *Personnel Journal*, 62(3), 222–228.
- Gorden, C. (2022). *Efficacy of multivitamin-mineral supplementation on measures of anxiety, depression, self-esteem, dysregulation, and perceived stress in young adults* [Doctoral dissertation, University of Arkansas].
- Grenville-Cleave, B., & Boniwell, I. (2012). Surviving or thriving? Do teachers have lower perceived control and well-being than other professions?. *Management in Education*, 26(1), 3-5.
- Gross, J. J. (1998). The emerging field of emotion regulation: An integrative review. *Review* of General Psychology, 2(3), 271-299.
- Gupta, M., & Rani, S. (2014). Burnout: A serious problem prevalent among teachers in the present times. *Journal of Education & Research*, 4(1).
- Gurganious, N. (2017). The relationship between teacher autonomy and middle school students' achievement in science [Doctoral Dissertation, Walden University].
- Hakanen, J. J., & Schaufeli, W. B. (2012). Do burnout and work engagement predict depressive symptoms and life satisfaction? A three-wave seven-year prospective study. *Journal of Affective Disorders*, 141(2-3), 415-424.
- Han, L. (2020). On the Relationship between Teacher Autonomy and Learner Autonomy. *International Education Studies*, 13(6), 153-162.
- Hanushek, E. A., Link, S., & Woessmann, L. (2013). Does school autonomy make sense everywhere? Panel estimates from PISA. *Journal of Development Economics*, 104, 212-232.
- Harahap, K., & Effiyanti, T. (2015). Technostress among educators: A revisit of social cognitive perspective. *Asia Pacific Journal of Contemporary Education and Communication Technology*, 1(1).
- Hassan, O., & Ibourk, A. (2021). Burnout, self-efficacy and job satisfaction among primary school teachers in Morocco. *Social Sciences & Humanities Open*, 4(1), 100148.
- Hill Jr, T. E. (1991). Autonomy and self-respect. Cambridge University Press.
- Huang, J. (2005). Teacher autonomy in language learning: A review of the research. *Research Studies in Education*, 3(203-18).
- Iancu, A. E., Rusu, A., Măroiu, C., Păcurar, R., & Maricuțoiu, L. P. (2018). The effectiveness of interventions aimed at reducing teacher burnout: A metaanalysis. *Educational Psychology Review*, 30(2), 373-396.
- Ibrahim, H., & Yusoff, Y. M. (2015). User characteristics as antecedents of techno stress towards EHRM: From experts' views. Procedia-Social and Behavioral Sciences, 172, 134-141

- Information Tomorrow : Reflections on Technology and the Future of Public and Academic Libraries, edited by Rachel Singer Gordon, Information Today, Inc., 2007. ProQuest Ebook Central, http://ebookcentral.proquest.com/lib/ inflibnetebooks/detail.action?docID=3316126. Created from inflibnet-ebooks on 2021-12-07 12:27:38.
- Jay, T.B. (1981). Computerphobia: What to do about it. *Educational Technology*, 21(1), 47-48.
- Jimenez, R.M., & Vieira, F. (2015). Enhancing autonomy in language education: A casebased approach to teacher and learner development. De Gruyter Mouton.
- John, K. (1992). Technostress and the reference librarian. *Reference Services Review*, 20 (2), 8–10, www.jkup.net/tstr ref.html.
- Joo, Y. J., Lim, K. Y., & Kim, N. H. (2016). The effects of secondary teachers' technostress on the intention to use technology in South Korea. *Computers & Education*, 95, 114-122.
- Kameshwara, K. K., Eryilmaz, N., Tian, M., & Sandoval-Hernandez, A. (2020). Teachers' pedagogical autonomy, professional development and students' digital skills: New evidence from Italy. *Autonomie Locali e Servizi Sociali*, 43(2), 421-439.
- Karasek Jr, R. A. (1979). Job demands, job decision latitude, and mental strain: Implications for job redesign. *Administrative Science Quarterly*, 285-308.
- Karr-Wisniewski, P., & Lu, Y. (2010). When more is too much: Operationalizing technology overload and exploring its impact on knowledge worker productivity. *Computers in Human Behavior*, 26(5), 1061-1072.
- Khmelkov, V. T. (2000). Developing professionalism: Effects of school workplace organization on novice teachers' sense of responsibility and efficacy. [Masteral Dissertation, University of Notre Dame].
- Khodabakhshzadeh, H., Hosseinnia, M., Moghadam, H. A., & Ahmadi, F. (2018). EFL teachers' creativity and their teaching's effectiveness: A structural equation modelling approach. *International Journal of Instruction*, 11(1), 227-238.
- Kim, S. S., & Malhotra, N. K. (2005). A longitudinal model of continued IS use: An integrative view of four mechanisms underlying postadoption phenomena. *Management Science*, 51(5), 741-755.
- Klecker, B., & Loadman, W. E. (1996). An analysis of the School Participant Empowerment Scale (Short and Rinehart, 1992) Based on data from 4091 teachers in 183 restructuring schools [Master's Dissertation].
- Kulavuz-Onal, D. & Tatar, S. (2017). Teacher burnout and participation in professional learning activities: Perspectives from university English language instructors in Turkey. *Journal of Language and Linguistic Studies*, 13(1), 283-303.
- Kumar, G. (2020). Vidyabhyasamenna aasoothritha kalapam. Olive Publications.
- Kunter, M., Klusmann, U., Baumert, J., Richter, D., Voss, T., & Hachfeld, A. (2013). Professional competence of teachers: Effects on instructional quality and student development. *Journal of Educational Psychology*, 105(3), 805.

- Kyriacou, C. (2001). Teacher stress: Directions for future research. *Educational Review*, 53(1), 27-35.
- Lamb, T. (2008). Learner autonomy and teacher autonomy. *Learner and Teacher Autonomy: Concepts, Realities and Responses*, 269-281.
- Landeche, P. (2009). *The correlation between creativity and burnout in public school classroom teachers*. [Master's Thesis, No. 2914]. https://digitalcommo ns.lsu.edu/gradschool theses/2914
- Law, N., Niederhauser, D. S., Christensen, R., & Shear, L. (2016). A multilevel system of quality technology-enhanced learning and teaching indicators. *Journal of Educational Technology & Society*, 19(3), 72-83.
- LeFevre, M., Matheny, J., & Kolt, G. S. (2003). Eustress, distress, and interpretation in occupational stress. *Journal of Managerial Psychology*, 18 (7), 726-744. https://doi.org/10.1108/02683940310502412
- Li, D., & Allen, A. (2021). Three-level hierarchical linear modeling analyses of the relationship between political culture and teacher autonomy. *Education and Urban Society*, 53(3), 251-278.
- Li, X., & Liu, D. (2022). The influence of technostress on cyberslacking of college students in technology-enhanced learning: mediating effects of deficient self-control and burnout. *International Journal of Environmental Research and Public Health*, 19(18), 11800.
- Little, D. (1995). Learning as dialogue: The dependence of learner autonomy on teacher autonomy. *System*, 23(2), 175-181.
- Longman, S. M. D. (2013). A comparison of the perceptions of technostress experienced by teachers versus technology used by teachers in elementary education in a Southeastern school district. Southeastern Louisiana University.
- Losos, L. W. (2000). Comparing the motivational levels of public, private and parochial *High school teachers*. Saint Louis University.
- Lucia, C. E., & Cristian, C. M. (2010). The role of local public authorities in decentralizing Romanian public education system. *Procedia-Social and Behavioral Sciences*, 2(2), 3432-3436.
- Lundström, U. (2015). Teacher autonomy in the era of new public management. *Nordic Journal of Studies in Educational Policy*, *1* (2), 28144.
- Lyle, A. M., & Peurach, D. J. (2022). Changing notions of teacher autonomy: The intersection of teacher autonomy and instructional improvement in the US. *Research in Education*, *1*. https://doi.org/10.1177/00345237211055843
- Madigan, D. J., & Kim, L. E. (2021). Does teacher burnout affect students? A systematic review of its association with academic achievement and student-reported outcomes. *International Journal of Educational Research*, 105, 101714.

- Mahalakshmi, K., & AllySornam, S. (2013, June). The impact of personality type on techno stress: a study among the library professionals of engineering colleges of Anna University of technology, Coimbatore. In 2013 International Conference on Information Science and Technology Applications (ICISTA-2013) (pp. 165-168). Atlantis Press.
- Mahapatra, M., & Pati, S. P. (2018, June). Technostress creators and burnout: A job demands-resources perspective. In *Proceedings of the 2018 ACM SIGMIS* conference on computers and people research (pp. 70-77).
- Maier, C. (2014). *Technostress: Theoretical foundation and empirical evidence* [Doctoral dissertation, Otto-Friedrich-Universität Bamberg, Fakultät Wirtschaftsinformatik und Angewandte Informatik].
- Manzano Vázquez, B. (2018). Teacher development for autonomy: An exploratory review of language teacher education for learner and teacher autonomy. *Innovation in Language Learning and Teaching*, *12*(4), 387-398.
- Maslach, C. (1976). Burned-out. Human Behavior, 5(9), 16-22.
- Maslach, C. (1982). Burnout: The cost of caring. Prentice-Hall.
- Maslach, C., & Jackson, S. E. (1981). MBI: Maslach Burnout Inventory. Palo Alto, CA, 1(2), 49-78.
- Maslach, C., & Leiter, M. P. (2016). Understanding the burnout experience: recent research and its implications for psychiatry. *World Psychiatry*, 15(2), 103-111.
- Matheny, K. B., Gfroerer, C. A., & Harris, K. (2000). Work Stress, Burnout, and Coping at the Turn of the Century: An Individual Psychology Perspective. *Journal of Individual Psychology*, 56(1).
- Mausethagen, S., & Mølstad, C. E. (2015). Shifts in curriculum control: contesting ideas of teacher autonomy. *Nordic Journal of Studies in Educational Policy*, *1* (2), 28520.
- McCarthy, C. J., Lambert, R. G., Lineback, S., Fitchett, P., & Baddouh, P. G. (2016). Assessing teacher appraisals and stress in the classroom: Review of the classroom appraisal of resources and demands. *Educational Psychology Review*, 28(3), 577-603.
- Meier, S. T. (1983). Toward a theory of burnout. Human Relations, 36(10), 899-910.
- Moller, A. C., Deci, E. L., & Ryan, R. M. (2006). Choice and ego-depletion: The moderating role of autonomy. *Personality and Social Psychology Bulletin*, 32(8), 1024-1036.
- Murray, W. C., & Rostis, A. (2007). Who's running the machine? A theoretical exploration of work stress and burnout of technologically tethered workers. *Journal of Individual Employment Rights*, 12(3), 249-263.
- Mustaffa, C. S., Yusof, R., & Saad, A. R. M. (2007, May). Technophobia and individual characteristics: Understanding computer-related technostress among Malaysian employees. In 5th International Conference on Communication and Mass Media, May (pp. 21-22).

- Nel, W. (2015). Transformative Autonomy: Mixed Notes from Teachers to Higher Education. Journal of Higher Education in Africa/Revue de l'enseignement supèrieur en Afrique, 13(1-2), 133-145.
- Nero, A. B. (1986). Intrinsic/extrinsic motivational factors and perceived need deficiencies as a function of job level in an urban school district (satisfaction, achievement).
- Oberfield, Z. W. (2016). A bargain half fulfilled: Teacher autonomy and accountability in traditional public schools and public charter schools. *American Educational Research Journal*, *53*(2), 296-323.
- Oranje, A. H. (2001). *Teacher shortages, teacher job satisfaction, and professionalism: teacher assistants in Dutch secondary schools* [Master's Dissertation].
- Pandey, M. (2016). Impact of work-life balance on job satisfaction of women doctors. Problems and Perspectives in Management, 14(2), 319.
- Paradis, A., Lutovac, S., Jokikokko, K., & Kaasila, R. (2019). Towards a relational understanding of teacher autonomy: The role of trust for Canadian and Finnish teachers. *Research in Comparative and International Education*, 14(3), 394-411.
- Pareek, U. (1983). Organizational role stress. *Pfeiffer's Classic Inventories, Questionnaires, and Surveys*, *3*, 319-329.
- Parker, G. (2015). Teachers' autonomy. Research in Education, 93(1), 19-33.
- Parker, J. S., Parris, L., Lau, M., Dobbins, A., Shatz, L., Porush, S., & Wilkins, B. (2021). Perceived teacher autonomy support and self-determination skill expression: Predictors of student engagement among African American high school students. *Journal of Black Psychology*, 47(6), 445-475.
- Paulsrud, D., & Wermke, W. (2020). Decision-making in context: Swedish and Finnish teachers' perceptions of autonomy. *Scandinavian Journal of Educational Research*, 64(5), 706-727.
- Pearson, L. C., & Moomaw, W. (2005). The relationship between teacher autonomy and stress, work satisfaction, empowerment, and professionalism. *Educational Research Quarterly*, 29(1), 38-54.
- Peral, S., & Geldenhuys, M. (2016). The effects of job crafting on subjective well-being amongst South African high school teachers. SA Journal of Industrial Psychology, 42(1), 1-13.
- Perrone, F., Player, D., & Youngs, P. (2019). Administrative climate, early career teacher burnout, and turnover. *Journal of School Leadership*, 29(3), 191-209.
- Peters, M. (2001). Education, enterprise culture and the entrepreneurial self: A Foucauldian perspective. *The Journal of Educational Enquiry*, 2(2).
- Pillay, H., Goddard, R., & Wilss, L. (2005). Well-Being, Burnout and Competence: Implications for Teachers. *Australian Journal of Teacher Education*, 30(2), n2.
- Pines, A., & Aronson, E. (1988). Career burnout: Causes and cures. Free press.
- Pithers, R. T., & Soden, R. (1998). Scottish and Australian teacher stress and strain: a comparative study. *British Journal of Educational Psychology*, 68(2), 269-279.

- Prichard, C., & Moore, J. (2016). The balance of teacher autonomy and top-down coordination in ESOL programs. *TESOL Quarterly*, 50(1), 190-201.
- Pucci, E., Cristina, S., Antonaci, F., Costa, A., Imbriani, M., & Taino, G. (2015). P034. Technostress and primary headache: psychosocial risk. *The Journal of Headache* and Pain, 16(1), 1-1.
- Pullins, E., Tarafdar, M., & Pham, P. (2020). The dark side of sales technologies: How technostress affects sales professionals. *Journal of Organizational Effectiveness: People and Performance.*
- Ragu-Nathan, T. S., Tarafdar, M., Ragu-Nathan, B. S., & Tu, Q. (2008). The consequences of technostress for end users in organizations: Conceptual development and empirical validation. *Information Systems Research*, 19(4), 417-433.
- Raya, M. J., Lamb, T., & Vieira, F. (2017). *Mapping autonomy in language education: A framework for learner and teacher development*. Peter Lang.
- Rolon, M. M. B. (2014). A quantitative study to explore the relationship between technostress symptoms and technostress among Puerto Rican university students [Doctoral dissertation, Capella University].
- Rosen, L. D., Sears, D. C., & Weil, M. M. (1987). Computerphobia. Behavior Research Methods, Instruments, & Computers, 19(2), 167-179.
- Rudolph, L. (2006). Decomposing teacher autonomy: A study investigating types of teacher autonomy and how it relates to job satisfaction. University of Pennsylvania.
- Ryan, R. M., & Deci, E. L. (2017). Self-determination theory: Basic psychological needs in motivation, development, and wellness. The Guilford Press. https://doi.org/10.1521/978.14625/28806
- Şahin, Y. L., & Çoklar, A. N. (2009). Social networking users' views on technology and the determination of technostress levels. *Procedia-Social and Behavioral Sciences*, 1(1), 1437-1442.
- Salanova, M., Llorens, S., & Cifre, E. (2013). The dark side of technologies: Technostress among users of information and communication technologies. *International Journal* of Psychology, 48(3), 422-436.
- Salanova, M., Llorens, S., Cifre, E., & Nogareda, C. (2007). El tecnoestres: Concepto, medida y prevencion. *Nota Te cnica de Prevencion, 1*, 730.
- Salazar-Concha, C., Ficapal-Cusí, P., Boada-Grau, J., & Camacho, L. J. (2021). Analyzing the evolution of technostress: A science mapping approach. *Heliyon*, 7(4), e06726.
- Salokangas, M., Wermke, W., & Harvey, G. (2020). Teachers' autonomy deconstructed: Irish and Finnish teachers' perceptions of decision-making and control. *European Educational Research Journal*, 19(4), 329-350.
- Saloviita, T., & Pakarinen, E. (2021). Teacher burnout explained: Teacher-, student-, and organisation-level variables. *Teaching and Teacher Education*, 97, 103221.

- Sareen, P. (2019). Techno stress creators-An exploratory research on teaching and non teaching staff working in colleges. *International Journal of Management and Humanities (IJMH)*, 3(9), 1-7.
- Schaufeli, W. B., & Buunk, B. P. (2003). Burnout: An overview of 25 years of research and theorizing. *The Handbook of Work and Health Psychology*, 2(1), 282-424.
- Schaufeli, W. B., & Taris, T. W. (2005). The conceptualization and measurement of burnout: Common ground and worlds apart. Work & Stress, 19(3), 256-262.
- Schaufeli, W. B., Leiter, M. P., & Maslach, C. (2009). Burnout: 35 years of research and practice. *Career Development International*, 14(3), 204-220.
- Scott, C. R., & Timmerman, C. E. (2005). Relating computer, communication, and computer-mediated communication apprehensions to new communication technology use in the workplace. *Communication Research*, 32(6), 683-725.
- Seidman, S. A., & Zager, J. (1986). The teacher burnout scale. *Educational Research Quarterly*, 11(1), 26–33.
- Sharma, D., & Sareen, P. (2015). *Physical and psychological forms of Techno-stress among Employees working in organization*.
- Shaw, J. (2008). Teachers working together. *Learner and teacher autonomy: Concepts Realities and Responses*, 187-203.
- Shen, B., McCaughtry, N., Martin, J., Garn, A., Kulik, N., & Fahlman, M. (2015). The relationship between teacher burnout and student motivation. *British Journal of Educational Psychology*, 85(4), 519-532.
- Short, P. M. (1994). Defining teacher empowerment. Education, 114(4), 488-493.
- Simola, H. (2005). The Finnish miracle of PISA: Historical and sociological remarks on teaching and teacher education. *Comparative Education*, *41*(4), 455-470.
- Skaalvik, E. M., & Skaalvik, S. (2014). Teacher self-efficacy and perceived autonomy: Relations with teacher engagement, job satisfaction, and emotional exhaustion. *Psychological Reports*, 114, 68–77.
- Sokmen, Y., & Kilic, D. (2019). The relationship between primary school teachers' selfefficacy, autonomy, job satisfaction, teacher engagement and burnout: A model development study. *International Journal of Research in Education and Science*, 5(2), 709-721.
- Subrahmanyam, K., Kraut, R. E., Greenfield, P. M., & Gross, E. F. (2000). The impact of home computer use on children's activities and development. *The Future of Children*, 1, 123-144.
- Suran, B. G., & Sheridan, E. P. (1985). Management of burnout: Training psychologists in professional life span perspectives. *Professional Psychology: Research and Practice*, 16(6), 741.
- Sutton, R. E. (2004). Emotional regulation goals and strategies of teachers. *Social Psychology of Education*, 7(4), 379-398.

- Sutton, R. E., Mudrey-Camino, R., & Knight, C. C. (2009). Teachers' emotion regulation and classroom management. *Theory Into Practice*, 48(2), 130-137.
- Syvänen, A., Mäkiniemi, J. P., Syrjä, S., Heikkilä-Tammi, K., & Viteli, J. (2016, November). When does the educational use of ICT become a source of technostress for Finnish teachers?. In *Seminar.net*, 12 (2).
- Tarafdar, M., Bolman Pullins, E., & Ragu-Nathan, T. S. (2014). Examining impacts of technostress on the professional salesperson's behavioural performance. *Journal of Personal Selling & Sales Management*, 34(1), 51-69.
- Tarafdar, M., Cooper, C. L., & Stich, J. F. (2019). The technostress trifecta-techno eustress, techno distress and design: Theoretical directions and an agenda for research. *Information Systems Journal*, 29(1), 6-42.
- Tarafdar, M., DArcy, J., Turel, O., & Gupta, A. (2015). The dark side of information technology. *MIT Sloan Management Review*, 56(2), 61.
- Tarafdar, M., Maier, C., Laumer, S., & Weitzel, T. (2020). Explaining the link between technostress and technology addiction for social networking sites: A study of distraction as a coping behavior. *Information Systems Journal*, 30(1), 96-124.
- Tarafdar, M., Tu, Q., Ragu-Nathan, B. S., & Ragu-Nathan, T. S. (2007). The impact of technostress on role stress and productivity. *Journal of Management Information Systems*, 24(1), 301-328.
- Tarafdar, M., Tu, Q., Ragu-Nathan, T. S., & Ragu-Nathan, B. S. (2011). Crossing to the dark side: Examining creators, outcomes, and inhibitors of technostress. *Communications of the ACM*, 54(9), 113-120.
- Taxer, J. L., & Gross, J. J. (2018). Emotion regulation in teachers: The "why" and "how". *Teaching and Teacher Education*, 74, 180-189.
- Thavenius, C. (1999). Teacher autonomy for learner autonomy. *Learner Autonomy in Language Learning: Defining the Field and Effecting Change*, 8, 159-163.
- Tian, Y., & Guo, Y. (2022). How Does Transformational leadership relieve teacher burnout: The role of self-efficacy and emotional intelligence. *Psychological Reports*, 1. https://doi.org/10.1177/00332941221125773
- Van-Dierendonck, D., Schaufeli, W. B., & Buunk, B. P. (1998). The evaluation of an individual burnout intervention program: The role of inequity and social support. *Journal of Applied Psychology*, 83(3), 392.
- Varanasi, R. A., Vashistha, A., Kizilcec, R. F., & Dell, N. (2021). Investigating technostress among teachers in low-income Indian schools. *Proceedings of the ACM on Human-Computer Interaction*, 5, 1-29.
- Vieira, F. (2020). Pedagogy of experience in teacher education for learner and teacher autonomy. Profile Issues in Teachers Professional Development, 22(1), 143-158.
- Vieira, F., Barbosa, I., Paiva, M., & Fernandes, I. S. (2008). Teacher education towards teacher (and learner) autonomy. *Learner and Teacher Autonomy: Concepts, Realities, and Responses, 217*(1), 236-246.

- Wang, K., Shu, Q., & Tu, Q. (2008). Technostress under different organizational environments: An empirical investigation. *Computers in Human Behavior*, 24(6), 3002-3013.
- Wang, Q., & Zhang, H. (2014). Promoting teacher autonomy through university–school collaborative action research. *Language Teaching Research*, 18(2), 222-241.
- Wang, X., & Li, B. (2019). Technostress among university teachers in higher education: A study using multidimensional person-environment misfit theory. *Frontiers in Psychology*, 10, 1791.
- Weil, M. M., & Rosen, L. D. (1997). Technostress: Coping with technology@ work@ home@ play (Vol. 13, p. 240). J. Wiley.
- Wermke, W., & Höstfält, G. (2014). Contextualizing teacher autonomy in time and space: A model for comparing various forms of governing the teaching profession. *Journal of Curriculum Studies*, 46(1), 58-80.
- Wermke, W., & Salokangas, M. (2015). Autonomy in education: Theoretical and empirical approaches to a contested concept. *Nordic Journal of Studies in Educational Policy*, 2, 1–6. https://doi.org/10.3402/nstep.v1.28841.
- Wermke, W., Olason Rick, S., & Salokangas, M. (2019). Decision making and control: Perceived autonomy of teachers in Germany and Sweden. *Journal of Curriculum Studies*, 51, 306–325. https://doi.org/10.1080/00220272.2018. 1482960
- West, J. (2007). Technophobia, technostress. Information Tomorrow: Reflections on Technology and the Future of Public and Academic Libraries, 203.
- Wharton, A. S. (1993). The affective consequences of service work: Managing emotions on the job. *Work and Occupations*, 20(2), 205-232.
- White, P. A. (1992). Teacher empowerment under "ideal" school-site autonomy. *Educational Evaluation and Policy Analysis*, 14(1), 69-82.
- Wilches, J. U. (2007). Teacher autonomy: A critical review of the research and concept beyond applied linguistics. *Íkala, Revista de Lenguaje y Cultura, 12*(18), 245-275.
- Willner, R. G. (1990). *Images of the Future Now: Autonomy, Professionalism, and Efficacy*. Fordham University.
- Xu, H. (2015). The development of teacher autonomy in collaborative lesson preparation: A multiple-case study of EFL teachers in China. System, 52, 139-148.
- Yin, C., Lim, K. H., Sun, Y., & Zhao, D. (2012). A technology-individual contingency perspective of mobile technostress: The moderating role of personality. In 18th Americas Conference on Information Systems 2012, AMCIS 2012.
- Yin, H., Huang, S., & Wang, W. (2016). Work environment characteristics and teacher well-being: The mediation of emotion regulation strategies. *International Journal of Environmental Research and Public Health*, 13(9), 907.
- Zhao, G., Wang, Q., Wu, L., & Dong, Y. (2022). Exploring the structural relationship between university support, students' technostress, and burnout in technologyenhanced learning. *The Asia-Pacific Education Researcher*, 31(4), 463-473.
APPENDICES

Appendix I

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TECHNO STRESS SCALE

MALAYALAM (DRAFT)

Dr. Hassan Koya M.P Associate Professor

Shafeek P Senior Research Fellow

Personal Information						
¥Ç¢Þɵæa çÉøí	:					
(ÈßVÌ t Îß^)						
ÕMcÞÜÏJ NæaçÉøi	:					
148^ (ØiµâZ)	:					
صåZ §È¢	: ØßÌ ß ®ØiØl⁄ കേരള സ്റ്റേറ്റ് സിലബസ്					
صáZ ÄÜ¢	: æædÉĨùß/æØA1 ùß/ÙÏVæØA1 ùß					
Ül¢to¢	: ¦ Y/æÉY					
¥Ç¢ÞÉÈÉø%°Ϊ¢	: 5 ÕV·×J ßW ÄÞæÝ/10 ÕV·×J ßW ÄÞæÝ/10 ÕV·×J ßW µå;áÄW					
dÉçÆÖ¢	: dupf ¢ /È u@¢					
Õß×ï ¢	: أÞ-× / ØÞĴ âÙc ÖÞØtÄÖ / أَلَّكُ / أَلَّهُ أَطْڰ / عَدَّا اللَّهُ الْعَالَ (اللَّهُ المَّانِ اللَّ					
ϛΪ ϷϭϲÄμΖ	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET					
µOcâGV çµÞÝíØáµZ	: PGDCA/DCA/Other/ No Courses					

ÈN¢ÆÖBZ

ÄÞæÝ æµÞ¿áJ BøBA áK ²ÞçøÞ dÉØÄÞÕÈA ᢠçÈæø 1. ÉåVHÌÞÏ á¢,çÏ Þ¼B A áK á 2.çÏ Þ¼&A áK á 3. ¥Í&ÉÞÏ Ĵ&^ 4.Õ&ÇÏ Þ¼&A áK á 5.ÉåVHĴÞÏ á¢ Õ&ÇÏ Þ¼&A á Ká®KBæÈ¥F ÍdÉÄBµøÃBZæµÞ¿áJ BøBAáKá.

 2 pçøþ dÉØiÄþÕÈÏ ál ÈlBæ{ Ølì -t ß, ß¿çJ þ{l ®dÄÍþdÄl ÖøßÏ þæÃ -Kí Äðøál ÞÈBA áµ. ¦ dÉÄBµøÃ ¢ dÉØÄÞÕÈÏ áæ; ÈOùBÈí çÈæøÏ áU çµÞ{-J ßW (✓) ° ßÙIÈÎßGí çø¶æM¿áJ áµ.

SI. No.	(ÉØÄÞÕÈ	EåVHIÞI át çi þýðaáKá	çÏ Þ¼łłAáKá	¥10161131	ÕlçÏ Þ¼lA áK á	EavhTpi at Ölçi þylbAáKá
1						
2	ØÞçCÄlµÕlkt©ÉçÏÞolk, áU¥ÇcÞÉÈøàÄlA í ÇÞøÞ{¢Ä‡Þæù¿áMáµZ¦ÕÖcÎÞÃí.					
3	ÉáÄßI ØÞçCÄßµ ÕÆc Øbàµø&A áçOÞZ ¥ÇcÞÉ ÈøàÄßµZ ÎÞxÞX È®VÌ t &A æMzáK ál í					
4	ÉáÄßI ØÞçCÄ&µ Õ&ci áÎÞI ßÄÞ&Þ∨ c¢ dÉÞÉ&A á KÄ&ÈÞI ߥÇcÞÉÈÄ‡Þæù;áMáµ{&WÎÞxBZÕøá JÞXÈ®VÌ t &ÄÈÞIß					
5	<code>Æav- cè@cæ;µcèp{%s@tµ@ÃBZ@tcipobAaccondenterpreserve} copZÖpøaøbµlaißacaµ{alpôpùali}</code>					
6	ÉáÄß ØÞçCÄBµ ÕÆc ²ÞVNÖµÄB µáùA áK á.					
7	μΦΦῦἀἶ ἀμ{Β₩ ΦϷҫϹӒβμÕιϞϲ ©ÉҫΪ ϷϭͽΑ ἀΚ ӒίÕß ℰϲϷϒϳ βμ{ΒΨ ΦϷϛϹӒβμÕιϞϲΪ åæ¿ ÆåøåÉҫΪ ϷϭͼÕV- ϳ ϐΑϷΧ μϷͽÃ ἶϷμάΚά.					
8	² ÞY ææÜX Ø¢ÕKÇÞÈB {KÜåæ; ØbµÞøcĵZÈ×í ;æM;áKá ®KÄí¦ÖCÏáIÞAáKá.					
9	ÙÞA \$B íçÉÞæÜÏ áU ææØÌ V¦ dµĨà Bæ{A á ùß, í ÍàÄßİ ál í					
10	çØÞ×cW ÎàÁßİ, ÎxᦠMáµZ Äá;Ðßİ Õ®Éçİ Þ oß, íÕcµÆßÉøÎÞÏ Õ®ØøÐZ°à×â氇æM;á Kál í					
11	ÕF &A æM;áçÎÞ ®KíÍÏ Kí¥ù®ÕáµZ ²ÞY ææÜ È®W ÉCíæÕA Þùß^.					
12	ÉảÄŀJ ØÞçCÄŀµ ÕŀEc µÞøÃ ¢ ° áøåÐŀJ ØĤ J ŀ ĚáU ſW µå¿áÄW ç¼ÞÜŀæ° ‡ÞX ØÞÇŀA Þùál í					
13	æÈxÕVA (ÙÞVÁ í æÕӴ V, çØÞË kí æÕӴ V Äá¿ B Å ÕӴ áĨ ÞӴ ßÌ t æMG dÉÖIÈ B Z µå¿áÄW ØĨĬÈ ×ý¢ ©ł ÞA Þùál í					
14	ÉÀÈ ØÞÙº øcJ &ëÜÎÞxƁæ{Aáù&, íµāÄcĨÞÏ ÇÞøÃ AíçÕI & ¥ÕÇM&ÕØƁZÄc¾&A1 kÕ- øáKá.					
15	μΦØĨÙåĨßW æ¿μIÇÈÞ{¼ß©ÉÇÏ Þ∽ßA áçOÞZ ØĨĬ Ì t ßÄĨÞÏ ßÉÞÀÍÞ∽¢ÉåVJ ßÏ ÞAÞX ØÞÇßAá Kß^.					

Sl. No.	d ÉØÄÞÕÈ	EåVHIÞI át çi þýðaáKá	çÏ Þ¼ ßA áKá	¥11000 ¥1000	ÕIÇİ Þ¼BA áK á	Eavhīpī at Õrcī þyrða áKá
16	ÉảÄŀĨ ØÞçCÄŀµ ÕŀÆc µÞøÃ ¢ ÕcµiÄŀ¼àÕŀÄ¢ dµ1 øÙŀÄ1 ÞÏ ŀ.					
17	濵çÈÞ{¼ß صßWØiÕßµØßMßA ÞX ¦ ÕÖcÎÞÏ øàÄßÏ ßÜáU §XØVÕàØíææd¿ÈßBíÜÍßAáKß^.					
18	æ® ¿₿ÉÀçÈÞɵøÃ Ѣ Z ØCàVĦŐáť ÉøØÉøť ©- ÉçÏ ÞのЉАÞX ØÞÇЉАÞJ ÄáÌÞÏ ₿¥ÈáÍÕæM;áKá.					
19	¥ÇcÞÉÈJ ßW ²øá ØiÅßø¢ Íà×Ãßİ ÞÌßæ¿µíçÈÞ{- ¼ßĨÞùáKá.					
20						
21	ÉáÄŀŸ ØÞçCÄŀµ ÕŀÆc µÞøÃ ¢ µáùE ØĨŸ ¢ ĨÞŀÄçĨ µá¿á¢Ì ŐáĴÞŸ ŀ8æ° ÜÕÝŀA ÞX µÝŀŸ áK áU á.					
22	ÉáÄÄ ØÞçCÄßµÕßci áæ; ©Éçi Þo¢ ÉáØÄµÕÞ ï ÈÖàÜæJ ì ÞÇß, ßál í					
23	ÉáÄİİ ØÞçCÄIµÕN£c ææµµÞøc¢æ°‡áKÄIW µáGI µZAí¥ÇcÞɵçøAÞZ¥Õ o ÞÙÎáæI KçÄÞK WdÉÖIÈ¢Øå×įjBAÞùáI í					
24	ØÞçCÄ&µÕÆcÏ &Ü¥ù®ÕíÕ{æø µáùÕÞÃí					
25	ÉÀBMBA áK ÕB×ÏÕáÌÞÏßÌt æMG¦Máµ{á¢ çØÞËxáæÕÏùáµ{ᢩÉçÏÞഗBA áKÄBÜáUÉøB° Ï¢µáùÕÞÃí					
26	ÕMECÞÍ CÞØ ø¢oj í ØÞçCÄßµ ÕMECİ áæ; ÉáÄßİ ØÞÇcĵæ{ µáùß, í ¥çÈb×ßA Þùál í					
27	®æa Øì 14µí;áî ÞÏ ßì t æMG ¦ Máµ{ᢠçØÞЁíxá æÕÏ ùáµ{áæ; Ï á¢ ÜÍcÄ µáùÕÞà í					
28	\emptyset ÞçCÄ\$µÕ\$£c [©] ÉçÏÞvs}, áU ¥ÇcÞÉÈ øàÄ\$µÞØá Îáù\$µ{8W tÉçÏÞvs&PÈáU¥ÕØø¢ÜÍ\$APù\$^.					
29	æÎÞææÌ WÇËÞY, ÜÞÉÇ¿ÞMíÄá¿BŔIŐÏáæ; ©É çÏÞovîµÞøÃ¢ØĨåÙŐáĨÞÏáUÌt¢µáùÏáKá.					
30	ŞaVæÈxWÈlKíµKGáKÕlõøBZ¦ÇlµÞølµÎ^ ®KçÄÞKW©Ií					
31	®ÁáçA-×ÃW ¦ MáµZ, ²ÞY ææÜX µÞØáµZÄá ¿Bĺİ Õ µá¿áÄÜÞİ ßÕKçÄÞæ; µÞØûᢠÉÀÈ¢ ÈßÜ- ÕÞøÄµV, çÈø‰áKál í					

SI. No.	(ÉØÄÞÕÈ	EåVHIÞI át çi þýðaáKá	çÏ Þ¼&A áK á	¥1100000000000000000000000000000000000	ÕIÇİ Þ¼BA áK á	Eavhīpī at Õrcī þyaba áKá
32	ÉøOøÞonÄ øàÄßİ IW ÉÀIMIA áK ÄBÈÞà í dÉÞÇcÞÈc¢ æµÞ; áA ÞùáU Äí					
33	µáG8µ{áæ; ÉÀÈÈBÜÕÞø¢ ØÞçCÄ8µÕ84cï 8Üâæ; ÕV- i 8M8A ÞX µÝßİ á¢					
34	ÉáÄŀÍ ØÞçCÄŀµÕŀÆc ÉÀŀA ÞÈát ¥Äi ¥ÇcÞÉÈ- J ſW çÕł ÕŀÇt dÉçÏ ÞoßA ÞÈát ÄÞÜŀÉøct µÞà ŀ A Þùál í					
35	æ° ÜÕí µáùÆ øàÄÅ ßW ØÞçCÄßµ ÕßEc dÉç'İ Þ¼È- æMzáJ Þùál í					
36	ØϷϛϹӒϐμÕϜϾ©ÉϛΪϷϭϐ, áU ¥ÇϲϷÉÈøàÄߥÇϲϷ Éμæα θμϐΪϷν μἂӕΪ θΈӒβμåÜἶϷΪβÌϷÇϐΑáΚá.					
37	μÞÞØáĨáùßµ{ſW æżµģÈÞ{¼ß©ÉçÏÞഗßAáçOÞZ ¥ÇcÞɵëÕßEcÞVjßÌt¢È×íjæMzáKá.					
38	ÕcÄcØÄÎÞæÏ Þøá ÉÀÈÞÈáÍÕ¢ ØÞçCÄ&µÕÆcÏ 8Üå æ; ØÞÇcĨÞµáKá.					
39	ØÞÇCÄBµÕKEc©ÉÇÏÞ06, áU ÉÀÈ¢¼àÕBÄ05tß ÏÞÏ ÕKECÞÍCÞØÎ^®KÇÞØÃÏáIí					
40	ÉáÄßİ Õß4cÞVj ßµZ濵çÈÞ{¼ßçµdwàµãÄĨÞİ ÉÀ- ÈØÞÙºøcBç{Þ;í®{áMJ ßWæÉÞøáJ æM;áKá.					
41	μΦØίùåÎßW æ¿μ¢ÈϷ{¼ß©ÉçΪϷտ&ΑáçΟϷΖμΦØί ùå¢ÎϷçȼæÎaiÈßİdLÃÕßçÇÏĴ^ϷӒϷμáΚá.					
42	ØĨĴ₽VGíµÞPØíùåĴáµZçÕłÕßÇJßW®ÉçÏÞ∽ßAÞ ùálí					
43	ØÎÎ ÞVGi ù lî Î & Ü µÞØáµç { Þ¿i Ő M:cÞVj ßµZA í ÄÞÜí Éøc¢ µå¿áÄÜÞà í					
44	μÞØíùáĨ®V ¥ÇcÞɵæa ØíÅÞÈ¢ææµ‡¿AÞX ØÞçCÄ&µÕ&cAíµÝ8Ïá¢.					
45	ÉáÄßİ ØÞçCÄßµ Õ&c µÞøÃ¢°áøáBßİ ØÎİJ ß ÈáUßW µâ;áÄW ç¼ÞÜ&c°‡ÞX ÈßVÌ t ßAæM;á Kálí					
46	濵íçÈÞ{¼ßØíµßWØíÕßµØßMßAÞX¦ÕÖcÎÞÏ ØÎÏ¢µæIJÞXµÝßÏáKß^.					
47	ÉáÄßİ ØÞçCÄßµ Õ&c µÞØùáÌßW ËÜdÉÆÌÞÏß ©ÉçÏÞ∽ßAÞX µÝßÏáçĨÞ®K ¦ÖC©I ÞÕÞùál í					

Appendix II

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TECHNO STRESS SCALE

MALAYALAM (FINAL)

1.1.6

4.

п

Dr. Hassan Koya M.P Associate Professor Shafeek P Senior Research Fellow

	Personal Information
¥Ç¢ÞɵæaçÉøi	:
(È₨Ì t Ĩß^)	
ÕMcÞÜÏ J NæaçÉøí	:
148^ (ØiµâZ)	:
صåZ §È¢	: ØłÌ ł ®ØiØł/ കേരള സ്റ്റേറ്റ് സിലബസ്
صáZ ÄÜ¢	: æædÉÎùß/ æØA‡ ùß /ÙÏ VæØA‡ ùß
¢0¢	: ¦ Y/æÉY
¥Ç¢ÞÉÈÉøl⁰Ї¢	: 5 ÕV-×J № ÄÞæÝ/10 ÕV-×J № ÄÞæÝ/10 ÕV-×J № µå;áÄW
dÉçÆÖ¢	: dub] ¢/Èvø¢
Õß×ä ¢	: أÞ-× / ØÅأ هُلُ c ÖÞØtlÄ¢/ ØÏ XØ//سà ÅÅ¢/ æ® الأج
ϛΪϷϭϭÄμΖ	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET
µOcâGV çµÞÝíØáµZ	: PGDCA/DCA/Other/ No Courses

<u>ÈNÇÆÖBZ</u>

ÄÞæÝ æµÞ¿áJ BØBA áK ²ÞçØÞ dÉØÄÞÕÈA át ç龯 1. ÉåVHĴÞÏ át.çÏ Þ¼B A áK á 2.çÏ Þ¼BA áK á 3.¥ÍBIÉÞÏ ÍB^ 4.ÕBçÏ Þ¼BA áK á 5.ÉåVHĴÞÏ át ÕBçÏ Þ¼BA á K á \mathbb{R} K BB æÈ ¥F í dÉÄBµØÃ BZ æµÞ¿áJ BØBA áK á.

 $\label{eq:production} ^{2} \mbox{pc} d {\Bar} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \$

SI. No.	(ÉØÄÞÕÈ	EåVHIÞI át çi þýðsáKá	çÏ Þ¼łłAáKá	¥{MÉH Î}^	Õlçİ Þ¼lbA áK á	ÉåVHTÞT át Ölçi þylld áK
1						
2	ÉáÄßI ØÞçCÄ&µÕ&c Øbàµø&A áçOÞZ ¥ÇcÞÉ ÈøàÄ&µZÎÞxÞXÈ®VÌ t &A æMzáK ál í					
3	ÉáÄÖ ØÞçCÄBµ ÕÆci áÍÞI BÄÞÆÞ∨c¢dÉÞÉBA á KÄBÈÞI B¥ÇcÞÉÈÄ‡Þæù;áMáµ{BWÎÞxBZŐøá JÞXÈRVÌ tBÄÈÞIB					
4	ÆàV-, çÈø¢æ¿µçÈÞ{¼ß©ÉµøÃ BZ©ÉçÏÞ∽‰Aá çOÞZ ÖÞøàø&µÌái ßìáGáµ{ái ÞÕÞùáI í					
5	ÉáÄß ØÞçCÄ&µÕ&c ²ÞVNÖµiÄßµáùAáKá.					
6	μΦΦῦἀἶ ἀμ{ϐ₩ ΦϷҫϹӒϐμÕιϜϲ ©ÉҫΪ ϷϭϐΑ ἀΚ ӒίÕϐ ℰϲϷϒϳ ϐμ{ϐ₩ ΦϷϛϹӒϐμÕιϜϲΪ åæ¿ ϜάøάÉҫΪ ϷϭϐÕV- ϳ ϐΑϷΧ μϷͽÃἶϷμάΚά.					
7	² ÞY ææÜX Ø¢Õ&ÇÞÈB {8Üåæ; ØbµÞøcĵZ È×í ;æM;áKá ®KÄí¦ÖCÏál ÞAáKá.					
8	ÙÞA \$B í çÉÞæÜÏ áU ææØÌ V¦ ¢µĨà Bæ{A á ùß, í Í àÄßÏ ál í					
9	çØÞ×cWÎàÁŔĬ,ÎxᦠMáµZÄá;BŔĬÕ©ÉçÏÞ on&,íÕcµÄ&ÉøĨÞÏ Õ®Õø⊕Z°à×â氇æM;á Kálí					
10	ÕF &A æM;áçÎÞ ®KíÍÏ Kí¥ù®ÕáµZ ²ÞY ææÜ È®W ÉCíæÕA Þùß^.					
11	ÉáÄŔĬ ØÞçCÄßµ ÕŀÆc µÞøÃ¢°áøáÐŔİ ØĨĬJ ß ÈáU ſW µâźáÄW ç¼ÞÜlæ°‡ÞX ØÞÇ&A ÞùáI í					
12	æÈxÕVA (ÙÞVÁ í æÕ¦ V, çØÞË kí æÕ¦ V Äá¿ B ßÌ - Õ¦ ál Þ¦ ß Ì -t æMG dÉÖÈ B Z µá;áÄW Ø1 ¦ È- ×í;¢ ©I ÞA Þùál í					
13	ÉÀÈ ØÞÙº øcJ &ëÜÎÞxBæ{Aáùß, íµāÄcĨÞÏ ÇÞøÃ AíçÕI ߥÕÇM£ÕØBZÄc¼&çA1 kÕ- øáKá.					
14	μΦΦίὺåἶßW æ¿μιςἘϷ{¼ß ©ÉçΪ ϷϭϧΑάςΟϷΖ ΦἶΫ· Ì-t βӒἶϷΫßĖϷλίϷϭ¢ÉåVJßΪϷΑϷΧØϷζϐΑά Κβ^.					
15	ÉáÄŀÍ ØÞçCÄŀµ ÕŀÆc µÞøÃ ¢ ÕcµiÄŀ¼àÕŀÄ¢ dµì- øÙŀÄÌ ÞÏ ß.					

SI. No.	d ÉØÄÞÕÈ	EåVHIÞI át çi þýðsáKá	çÏ Þ¼łłAáKá	¥{触É时〕}	ÕIÇİ Þ¼lA áKá	ÉåVHĨÞĨ át Õlçï þ¼lþA áKá
16	濵çÈÞ{¼ß∅µl₩∅íÕßµØßM&ÞX¦ÕÖcÎÞÏ øàÄßÏ 8ÜåU§XØVÕàØíæætjÈBBíÜÍ&AáKß^.					
17	æ® ¿₿ÉÀçÈÞɵøÃ BZØCàVHÕå¢ÉøØÉø¢© ÉçÏ Þ∽ЉАÞХØÞÇЉАÞJÄåÎҎÏ₿¥ÈåÍÕæM;åKå.					
18	¥ÇcÞÉÈJ ßW ²øá ØíÅßø¢ Íð×Ãßİ ÞÍ ßæ¿µíçÈÞ{· ¼8 Ĩ ÞùáKá.					
19	\emptyset PçCÄBµÕB£c [©] ÉçÏ Þv3BA áK áæl CBÜát ¥ÄiØBÜ- Ì ØBÈi¥ÈáçÏ Þ½cĨ ÞŐÞùB^.					
20	ÉáÄßI ØÞçCÄßµ Õ&c µÞøÃ ¢ µáùE Ø1`I ¢ ÍÞdÄçÎ µázá¢Ì Őá1Þ'I ßæ°ÜÕÝ&A ÞX µÝßI áKáU å.					
21	ÉảÄĨ ØÞçCÄBµÕMc ææµµÞøct æ°‡áKÄW µáGb µZAí¥ÇcÞɵçøAÞZ¥Õ∽ÞÙĴæI KçÄÞK WdÉÖÆt Øð×itBAÞùáI í					
22	ØÞçCÄ&µÕ&cÏ &Ü¥ù®ÕíÕ{æø µáùÕÞÃí					
23	ÉÀBMBA áK ÕB×ÏÕáÌÞÏBÌt æMG¦Máµ{á¢ çØÞËxáæÕÏùáµ{ᢩÉçÏÞഗBA áKÄBÜáUÉøB° Ï¢µáùÕÞÃí					
24	®æa Øì 14µí;áÎ ÞÏ ßì ・t æMG ¦ Máµ{ᢠçØÞЁíxá æÕÏ ùáµ{áæ;Ï á¢ ÜÍcÄ µáùÕÞà í					
25						
26	æÎÞææÌ WÇËÞY, ÜÞÉſÇ¿ÞMí Äá; Ðßİ ÕÏ áæ; ©É- ÇÏ Þo¢ µÞØÃ¢ØÎâÙÕáĨÞÏ áUÌ t¢µáùÏ áKá.					
27	ŚaVæÈx℻È℻íμೠ℅KÕ℻℗℞Z¦Ç&µÞø&µĴ^ ®KçÄÞKW©Ií					
28	®ÁáçA-×ÃW ¦ MáµZ, ²ÞY ææÜX µÞØáµZÄá ¿B ßİ Õ µá¿áÄÜÞÏ ß ÕK çÄÞæ¿ µÞØûᢠÉÀÈ¢ ÈßÜ· ÕÞøÄµV, çÈø‰áKál í					
29	ÉøOøÞonÄ øàÄßİ IW ÉÀIMIA áK ÄBÈÞà í dÉÞÇcÞÈc¢ æµÞ; áA ÞùáU Äí					
30	µáG8µ{áæ; ÉÀÈÈBÜÕÞø¢ ØÞçCÄ8µÕ84cï 8Üâæ; ÕV- i 8M8A ÞX µÝ8ï á¢					
31	æ° ÜÕí µáùÆ øàÄßİ ßW ØÞçCÄßµ ÕßEc dÉçİ Þ¼È- æM;áJ Þùál í					

Sl. No.	d ÉØÄÞÕÈ	EåVH I ÞI át çi þýðð á Ká	çÏ Þ¼ ßA áKá	¥1100000000000000000000000000000000000	Õlçİ Þ¼lA áK á	Eavhīpī ac Õrcī þyrðaáká
32	\emptyset PçCÄ&µ ÕÆc ©ÉçÏ Þo‰, áU ¥Ç¢ÞÉÈøàÄ&¥Ç¢Þ ɵæa dµ& Þ∨ µÄæï dÉÄ&µàÜÎÞÏ &Ì ÞÇ&A áKá.					
33	μÞØáĨáùßµ{ſW æ;µíçÈÞ{¼ß ©ÉçÏ ÞơßA áçOÞZ ¥ÇcÞɵëÕßcÞVj ßÌt ¢È×íjæM;jáKá.					
34	ÕcÄcØíÄÎÞæÏ Þøá ÉÀÈÞÈáÍÕ¢ ØÞçCÄßµÕÆcÏ 8Üâ æ; ØÞÇcĨÞµáKá.					
35	ØÞçCÄ&µÕÆc©ÉçÏÞvn&, áU ÉÀÈ¢¼àÕ&Ävst& ÏÞÏ ÕÆcÞÍcÞØĴ^®KÇÞøÃÏáI í					
36	μΦØίùåÎßW æ¿μ¢ÈÞ{¼ß©ÉçÏ Þഗ&A áçOÞZ μΦØi ùå¢ÎÞçȼæÎ aí ÈßÏ dL à ÕßçÇÏ Ĵ^ÞÄÞµáKá.					
37	ØÎÎ ÞVGi µÞPØi ùâĨ áµZçÕłÕßÇJßW©ÉçÏ ÞoßAÞ ùáIí					
38	μÞØíùáĨW ¥ÇcÞɵæa ØíÅÞÈ¢ææµ‡¿AÞX ØÞçCÄ&µÕÆcAíµÝßİá¢.					
39	ÉáÄBÍ ØÞÇCÄBµ ÕM£c µÞøÃ¢°áøáBBÍ ØĨĬJB ÈáUBW µà¿áÄW ç¼ÞÜBæ°‡ÞXÈBVÌ tBAæM¿á Kálí					
40	濵íçÈÞ{¼ßØíµßWØíÕßµØßMßAÞX¦ÕÖcÎÞÏ ØĨÏ¢µæIJÞXµÝßÏáKß^.					
41	ÉáÄßİ ØÞçCÄßµ Õ&c µÞØûâÎßW ËÜdÉÆÎÞÏß ®ÉçÏÞ0ßAÞX µÝßÏáçĨÞ®K ¦ ÖC ©I ÞÕÞùál í					

Appendix III

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TECHNO STRESS SCALE

ENGLISH (FINAL)

Dr. Hassan Koya M.P			Shafeek P
Associate Professor			Senior Research Fellow
		<u>Personal Information</u>	
Name of the Teacher	:		

Name of the Teacher	
(Optional)	
Name of the School	:
District (School)	:
Type of School	: CBSE/ Kerala State Syllabus
Level of School	: Primary/Secondary/Higher Secondary
Sex	: Male/Female
Teaching Experience	: Below 5 years/ Below 10 years/ Above 20 years
Locale	: Rural/Urban
Subject	: Language/ Social Science/ Science/ Mathematics/ IT
Educational Qualification	ns: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET
Computer Courses	: PGDCA/DCA/Other/ No Courses

Instructions

Five responses are given to each of the following Statements. 1. Strongly Agree, 2. Agree, 3. No Opinion, 4. Disagree and 5. Strongly Disagree.

Place a \checkmark in the appropriate column to indicate how true each statement is for you.

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
1	I think technology based teaching is better than any other method of teaching					
2	My teaching methods are forced to change as new technology is adapted.					
3	I forced to make changes in teaching preparation to adapt new technology.					
4	Physical difficulties can occur when using technology for a long time.					
5	I think new technology reduces memory.					
6	In my opinion the use of technology in classrooms leads to increased misuse of technology among students.					
7	I have lot of concerns about the loss of privacy through online systems					
8	I am afraid of cyber-attacks like hacking.					
9	My personal information is being exploited through social media and other apps.					
10	I didn't share my knowledge through online for fear of being cheated.					
11	New technology helps me to do more work in less time.					
12	Problems related to network, hardware and software cost me more time.					
13	I have to sacrifice my holidays for a better understanding of the changes in the learning environment.					
14	When technology is used in the classroom, I am unable to complete the lessons on time.					
15	My personal life has become chaotic due to new technology.					

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
16	Not getting the in-service training required to develop technology skills.					
17	I feel that IT learning tools are complex and difficult to use.					
18	I think technology is becoming a constant threat in teaching.					
19	I use the technology in teaching but it does not fit the syllabus.					
20	Due to new technology, I am able to spend less time with my family.					
21	Teachers feel that students are savvier than teachers in dealing with new technology.					
22	My knowledge of technology is limited.					
23	I have little experience in using apps and software related to the subject being taught.					
24	The availability of apps and software related to my subject is low.					
25	I didn't get adequate facility in classrooms for practicing technology -based teaching methods.					
26	My communication with the community is decreasing due to the use of devices such as mobile phones and laptops.					
27	I think that the information received through the internet is not reliable.					
28	I feel classroom learning is declining with the advent of educational apps and online classes.					
29	I emphasize teaching in traditional style.					

Sl. No.	Statements	ongly Agree	ree	Idecided	sagree	rongly sagree
		Stı	Ag	Un	Dis	Str Dis
30	I think the level of learning of children can be enhanced through technology.					
31	I try to use technology in a cost effective manner.					
32	I think that the teaching method using technology adversely affects the creativity of the teacher.					
33	I think the teacher-student relationship decreases when technology is used in the classroom.					
34	A different learning experience is possible through technology.					
35	There is a perception that learning with technology is not a life-affirming education.					
36	Classroom management becomes uncontrollable when technology is used in the classroom.					
37	I use smart classrooms effectively.					
38	I think technology can take the place of the teacher in the classroom.					
39	Due to new technology, people are forced to work more in less time.					
40	I can't find enough time to develop technology skills.					
41	I have concerns about whether new technology can be used effectively in the classroom.					

Appendix IV

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TEACHER AUTONOLY SCALE

MALAYALAM (DRAFT)

Dr. Hassan Koya M.P Associate Professor Shafeek P Senior Research Fellow

	Personal Information						
¥Ç¢ÞɵæaçÉøi	:						
(ÈℕÌ tÎß^)							
ÕMcÞÜÏ J NæaçÉøí	:						
148^ (ØiµâZ)	:						
صåZ §È¢	: Øß Ì ീ ®ØiØl/ കേരള സ്റ്റേറ്റ് സിലബസ്						
صáZ ÄÜ¢	: æædÉÎùß/æØA1 ùß/ÙÏVæØA1 ùß						
Ül¢ ஸ	: ¦ Y/æÉY						
¥Ç¢ÞÉÈÉøl⁰Ï¢	: 5 ÕV-×J ® ÄÞæÝ/10 ÕV-×J ® ÄÞæÝ/10 ÕV-×J ® µå¿åÄW						
dÉçÆÖ¢	: INPI ¢/Èvø¢						
Õß×ä ¢	: أك× / @كأ أَلُك OPØtlÄb/ @ï XØl/ مَكْ اللهُ المَعْ @ اللهُ المَعْ						
çΪ ϷϭϭÄμΖ	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET						
µOcâGV çµÞÝíØáµZ	: PGDCA/DCA/Other/ No Courses						

<u>ÈNÇÆÖBZ</u>

ÄÞæÝ æµÞ¿áJ BØBA áK ²ÞçØÞ dÉØÄÞÕÈA át ç龯 1. ÉåVHĴÞÏ át.çÏ Þ¼B A áK á 2.çÏ Þ¼BA áK á 3.¥ÍBIÉÞÏ ÍB^ 4.ÕBçÏ Þ¼BA áK á 5.ÉåVHĴÞÏ át ÕBçÏ Þ¼BA á K á \mathbb{R} K BB æÈ ¥F í dÉÄBµØÃ BZ æµÞ¿áJ BØBA áK á.

 $\label{eq:production} ^{2} \mbox{pc} d {\Bar} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \$

Sl. No.	(ÉØÄÞÕÈ	EåVHIPI å¢ çi þýðða áKá	çï Þ4 6A áKá	¥{MÉM`]}^	Õlçi Þidlið á Ká	EåVHTÞÍ át Ölçi þýlbAáKá
1	ÈÕàÈÕᢠdµßİ Þ∨ µÕáÎÞİ ¥ÇcÞÉÈ øàÄßµZ ØbàµøßA ÞX ØbÞÄdL cÕᢠçdÉÞr ÞÙÈÕáĨáI í					
2	øf ßÄÞA {áæ; ¥ÎßÄ §;æÉ;WÌ ái ßĨáGáµZ Øã×i;ßAáKál í					
3	\emptyset kÞËíÎàxßBáµ{kæÜ¥ÍkdÉÞÏ BZAí¥VÙ ÎÞÏ ÉøßoÃÈÜÍßAÞùál í					
4	ØÞĨ âÙßµ çØÕÈ dÉÕVJ ÈB {®VÏ çÅ×į;¢ÍÞ∽ ÎÞÕÞX ØÞÇßA áK á.					
5	ÕŀÆcÞVj ßµ{áæ; ÖßF Ã ÍÞ∽ĨÞÏß²øá È;É;ß ®;áAÞÈá©¥ÇcÞɵÈíØbÞÄdL cÎß^.					
6	µ₱Ø8ÜᢠÉáùJ áĨ ÞÏ ßÏ çÅ×í¿¢ ÉÀÈdÉÕVJ È BZÈ¿J ÞX ØÞÇ8A Þùál í					
7	dÉÇÞÈÞÇ¢Þɵæa ÈløLøÈløàf âµÞØlæa ØbÞÍÞÕlµÄæÏÌÞÇlAÞùálí					
8	ŐŀÆcÞVj ßµ{áæ; ËàWÁíd;ßMáµ{ᢠØíxÁß;âùáµ {ᢠÉâVà ÎÞÏ á¢ ¥ÇcÞɵøáæ; ÈßÏ dLÃJ ßÜÞÃí È;AáKÄí					
9	ÉáÄßİ çİ ÞocĵZ çÈźßdÉçÎÞ×X çÈżÞX ¥çȵ¢ ¥ÕØøB {ál í					
10	¥ÇcÞɵøáæ; ¥ÕÇßÕß×ÏJßW¥ÈÞçøÞưcµ- øÎÞÏÈßÏdLÃB{álí					
11	Øiµå{&a çÌ ÞÇÈÖÞØtlÄ¢ ÄàøáÎÞÈ&A áK Äí ¥ÇcÞɵøÞà í					
12	ÕŒcÞVj ßµ{áæ¿¥ÁíÎß×XÎÞÈÆm¢ÄàøáĨÞÈß AáKÄßW¥ÇcÞɵVAá¢ÉCálí					
13	µÞØíùáĨ ßW ÕÆcÞVj ßµ{áæ; ØßxßB í Ø¢ÕßÇÞÈ¢ ÄàøáĨ ÞÈßA áK Äí ¥ ÇcÞɵøÞà í					
14	صå{&a Èß Î¢, Èß dL âÄá;Bß µÞøcB ZAíøåÉ¢æµÞ;áAáKÄí¥ÇcÞɵøÞÃí					
15	صá{læa µølA áÜV Üf cB {ᢠÎáXoÃÈ dµÎB {ᢠÄàøáÎÞÈlA áK Äí¥ÇcÞɵøÞà í					
16	ÕM£cÞÍcÞØÕáÌÞÏßÌ tæMGÕcÄcØÄÌÞÙc ÈÏøåÉàµøÃ "g¼XØßµ{áÎÞÏßÌ tæM¿ÞX ¥ÇcÞɵVÎáXææµ®¿áAáKá.					

SI. No.	(ÉØÄÞÕÈ	EåVHIPI å¢ çi þýåðaáKá	çï þ %lA áKá	¥11帖的孔	Õlçi ÞikliðáKá	ÉåVHTÞÍ át Ölçi þýals áKá
17	ÕM£cÞVj ßµ{áæ; æÉøáÎÞx °GB {ᢠÈßÏ ÎB {á¢ øâÉàµøßA áK Äí¥ÇcÞɵøÞà í					
18	¥ÇcÞɵøáæ; æÉøáÎÞx ÈÏ BZÄàøáÎÞÈßA áKÄí ¥ÇcÞɵøáæ; I á¢ÎÞçȼæÎaßaI ᢵåGÞI °V, I ßÜåæ; I Þà í					
19	ØØæÉX×X, Éløl, áÕl;WÄi;BlİµÞøcBZAí ÏÞæÄÞøi ÎÞÈÆm B {i¢ÉÞÜlAiKI^.					
20	ŐM£cÞVj ßµ{áæ; ÖßF ÞÈÏ BZøâÉàµøßAáKÄí ¥ÇcÞɵøÞÃí					
21	µáG8µ{áæ; ØÎdwÎÞÏ ÎâÜc¢ÈRVÃÏJ 8W ²ÞçøÞ ¥ÇcÞɵVAᢥVÙĨÞÏ ÉCí¥ÈáÕÆ8, á ÈWµá Kál í					
22	ÉáÄßİ Äá©ÉµÞødÉÆŐáÎÞİ çÌ ÞÇçÈÞɵøÃ BZ©ÉçİÞ∽ßAáKÄíÈßøár ÞÙæM¿áJ áµİá© Ä¿TæM¿áJ áµIá©æ°‡áKá.					
23	øf μVÄð ØÙμøÃ ÈΪ ΒΖ ÄðdðÎÞÈßA áK Äí ¥Ç¢ÞÉμøÞà í					
24	§X æØVÕàØíædįΪ βÈι&ωí¥ÇcÞɵçøÞ¿íçÕΙ dÄ µå¿βΪ ÞçÜÞ°ÈçΪ Þæ;Ϊ ^È;J áK Äί					
25	µø&AáܢȮNÞÃJ®W¥icÞɵVAíÉCíµáù ÕÞÃí					
26	µø&A áÜJ &W dÉÞçÆÖ&µ ÕcÄ&ø&µÄÄA ÈáØø&, á U æÄøæE ;áMáµZAí¥ÕØøÎ&^.					
27	¥ÇcÞɵøÞÃíÉÞçÀcÄødÉÕVJÈBZÎáçKÞGá æÕAáKÄí					
28	ÕMECÞVj ßµ{áæ; ÉáçøÞ ơÄßÕßÜÏ BøáJ ÞÈáU dÉÞçÏ Þoßµ ÎÞVPBZÄàøáĨÞÈßAáKÄí¥i cÞÉ µøÞÃí					
29	صá{læa Íì Älµ ØÞÙ°øc Ø¢ÕlçÞÈJ lW (Physical Environment) ¥çcÞɵVAíÏ ÞæÄÞøá ÉCáĨlî					
30	μÞØíùᢠdÉÕVJÈBZÈ¿MBÜÞA áKÄBW¥Ç¢Þ ɵVA í ÉøBÎBÄBµ{álí					

SI. No.	d ÉØÄÞÕÈ	EåVHIPI át çi þyða áKá	çÏ Þ¼ ŀA áKá	¥{10℃时〕{}^	ÕIÇÏ Þ¼IA áKá	ÉåVHTÞÍ át Ölçi þýlbAáKá
31	µÞØí ùác dÉÕVJ ÈøàÄß ÄàøáĨ ÞÈßA áK Äí ¥ ÇcÞÉ µøÞÃ í					
32	ÈMÌ t BÄØBÜÌ ØBÈſÉáùæÎ°BÜ dÉçÄcµÕB× ÏBZAſµÞØſÈWµÞX¥ÇcÞɵVAſØbÞÄdLc Ĵál í					
33	dÉßXØßMÜßæa ØNÄÎß^ÞæÄ Õ&cÞVj ßµZAí ØNÞÈ¢ÈWµÞX ¥ÇcÞɵVAíØbÞÄdLcĨál í					
34	ØiµåZØÞOJ &µÈÏ¢øåÉæM;áJ áKÄW¥- Ç¢ÞɵVAíÉCáI í					
35	ælÉÞË×ÃWÕßµØÈJßÈíçÕIßİáUæÉÞÄá ÎÞÈÆm BZćÞùÞAáKÄí¥Ç¢ÞɵøÞÃí					
36	ÉåVà ΠÞÏ á¢ ÉáÄßİ µøßA áÜ¢ ÕßµØßMßA ÞX ¥Ç¢ÞɵVA í ¥ÕØøÎ ál í					
37	ŐŀÆcÞVj ßµZAíÉáÄßÏ ÉÀçÈÞɵøÃ BZÉøß° ÏæM¿áJ ÞX ¥ÇcÞɵVAí¥ÈáÕÞÆĨáI í					
38	ŐŀÆcÞVj ßµZAí¦ ÕÖcÎÞÏ µcÞOáµZØ¢, الله M&AÞX ¥ÇcÞɵVAíØbÞÄ&LcÍál í					
39	¥ÇcÞɵçÖ×8ÕVÇ8M8AÞÈÞÕ8ÖcÎÞÏædÉÞ¼- µí;áµZæ°‡ÞX¥ÇcÞɵVAíØbÞÄdLcĨálí					
40	ĨåÜcÈMHÏJ M ÕcÄcØÅĨÞÏ øàÄ&μZ®ÉçÏÞ ω&Ανμ¥ÇcÞÉμVAíØbÞÄ&LcĨáI í					
41	Éø&ÙÞøçÌÞÇÈ&BZ¦Ő&Ö¢ÈåØøÃ¢È;JÞ ÈåUØ&ÞÄ&Lc¢¥Ç¢ÞɵVAálí					
42	¥ÇcÞɵøáæ; çÈGBæ{ÎÞçȼæÎaá¢ØixÞËá¢ çÕłÕ&ÇJ®V¥ÈáçĨÞÆ&AÞùálí					
43	Øiµi{læaÈli ΢,Èli dLâÄi¿Bli µÞøcB{l æÜløf läÞA{iæ; S;æÉ;ÜiµZdÉÖÈÌÞÕÞùil í					
44	µá6kµ{áæ; ÕkµØÈJ kÈMáù¢øf kÄÞA{áæ; Ø¢ ÄåÉÄkI ÞÃíÎÞçȼæîakæaÜf cæîKíçÄÞKÞ ùál í					

Appendix V

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TEACHER AUTONOLY SCALE

MALAYALAM (FINAL)

1.1.6

4.

n

Dr. Hassan Koya M.P Associate Professor Shafeek P Senior Research Fellow

	Personal Information
¥Ç¢ÞɵæaçÉøi	:
(È№Ì tĺß^)	
ÕMcÞÜÏ J NæaçÉøí	:
148^ (ØiµâZ)	:
صåZ §È¢	: Øß Ì ീ ®ØíØl⁄ കേരള സ്റ്റേറ്റ് സിലബസ്
صáZ ÄÜ¢	: æædÉĨùß/æØA1 ùß/ÙÏVæØA1 ùß
ிழில்	: ¦ Y/æÉY
¥Ç¢ÞÉÈÉøl⁰Ï¢	: 5 ÕV-×J ſW ÄÞæÝ/10 ÕV-×J ſW ÄÞæÝ/10 ÕV-×J ſW µå¿áÄW
dÉçÆÖ¢	: INPI ¢/Èvø¢
Õß×ä ¢	: أÞ-× / @ÞĴ هُلُ C ÖÞØtlÄ¢/ ØÏ XØ//سà ßÄ¢/ æ® إ
çΪ ϷϭϭÄμΖ	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET
µOcâGV çµÞÝíØáµZ	: PGDCA/DCA/Other/ No Courses

<u>ÈNÇÆÖBZ</u>

ÄÞæÝ æµÞ¿áJ BØBA áK ²ÞçØÞ dÉØÄÞÕÈA át ç龯 1. ÉåVHĴÞÏ át.çÏ Þ¼B A áK á 2.çÏ Þ¼BA áK á 3.¥ÍBIÉÞÏ ÍB^ 4.ÕBçÏ Þ¼BA áK á 5.ÉåVHĴÞÏ át ÕBçÏ Þ¼BA á K á \mathbb{R} K BB æÈ ¥F í dÉÄBµØÃ BZ æµÞ¿áJ BØBA áK á.

 $\label{eq:production} ^{2} \mbox{pc} d {\Bar} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \mbox{d} \$

Sl. No.	d ÉØÄÞÕÈ	EåVHIPI å¢ çi þ‰ðAáKá	çİ Þ¥BA áK á	¥{MÉH]}^	Õlçİ Þ¼lAáKá	EåVHTPT át Ölçi þýlka áKá
1	ÈÕàÈÕᢠdµßİ Þ∨ µÕáÎÞİ ¥ÇcÞÉÈ øàÄßµZ ØbàµøßA ÞX ØbÞÄdL cõᢠçdÉÞr ÞÙÈÕáĨáI í					
2	øf ßÄÞA {áæ; ¥ÎßÄ §;æÉ;WÌ ái ßĨáGáµZ Øã×i;ßAáKál í					
3	\emptyset kþËíÎàxßBáµ{&Ü¥Í&ÉÞÏ BZAí¥VÙ ÎÞÏ ÉøßoÃÈÜÍßAÞùál í					
4	ØÞĨ âÙßµ çØÕÈ dÉÕVJ ÈB {®VÏ çÅ×į;¢ÍÞ∽ ÎÞÕÞX ØÞÇßA áK á.					
5	ÕŀÆcÞVj ßµ{áæ; ÖßF Ã ÍÞ∽ĨÞÏß²øá È;É;ß ®;áAÞÈá©¥ÇcÞɵÈíØbÞÄdL cÎß^.					
6	µ₱Ø8ÜᢠÉáùJ áĨ ÞÏ ßÏ çÅ×í¿¢ ÉÀÈdÉÕVJ È BZÈ¿J ÞX ØÞÇ8A Þùál í					
7	dÉÇÞÈÞÇ¢Þɵæa ÈløLøÈløàf âµÞØlæa ØbÞÍÞÕlµÄæÏÌÞÇlAÞùálí					
8	ÕÆcÞVj ßµ{áæ; 莒àWÁíd;ßMáµ{ᢠØíxÁß;àùáµ {ᢠÉåVà ÎÞÏ á¢ ¥ÇcÞɵøáæ; ÈßÏ dL à J ßÜÞÃí È;AáK Äí					
9	ÉáÄßİ çİ ÞocĵZ çÈźßdÉçÎÞ×X çÈżÞX ¥çȵ¢ ¥ÕØøB {ál í					
10	¥ÇcÞɵøáæ; ¥ÕÇßÕß×ÏJßW¥ÈÞçøÞưcµ- øÎÞÏÈßÏdLÃB{álí					
11	Øiµå{&a çÌ ÞÇÈÖÞØtlÄ¢ ÄàøáÎÞÈ&A áK Äí ¥ÇcÞɵøÞà í					
12	ÕŒcÞVj ßµ{áæ¿¥ÁíÎß×XÎÞÈÆm¢ÄàøáĨÞÈß AáKÄßW¥ÇcÞɵVAá¢ÉCálí					
13	µÞØíùáĨ ßW ÕÆcÞVj ßµ{áæ; ØßxßB í Ø¢ÕßÇÞÈ¢ ÄàøáĨ ÞÈßA áK Äí ¥ ÇcÞɵøÞà í					
14	صå{&aÈßİ΢,ÈßİdLâÄá¿BßİµÞøcB ZAíøåÉ¢æµÞ¿áAáKÄí¥ÇcÞɵøÞÃí					
15	صá{læa µølA áÜV Üf cB {ᢠÎáXoÃÈ dµÎB {ᢠÄàøáÎÞÈlA áK Äí¥ÇcÞɵøÞÃí					
16	ÕŀÆcÞĺcÞØÕáĨÞÏßÌtæMGÕcÄcØÄÌÞÙc ÈÏøåÉàµøÃ "g¼XØßµ{áĨÞÏßÌtæM¿ÞX ¥ÇcÞɵVÎáXææµ®¿áAáKá.					

SI. No.	d ÉØÄÞÕÈ	EåVHIPI å¢ çi þýåbAáKá	çÏ Þ¼ ßA áKá	¥1000000000000000000000000000000000000	ÕIÇİ Þ¼IA áKá	ÉåVHTÞÍ át Ölçi Þíald áKá
17	ÕŒcÞVj ßµ{áæ¿æÉøáÎÞx °GB{á¢ÈßİÎB{á¢ øåÉàµøßAáKÄí¥ÇcÞɵøÞÃí					
18	¥ÇcÞɵøáæ; æÉøáÎÞx ÈÏ BZÄàøáĨÞÈ&A áKÄí ¥ÇcÞɵøáæ;ïá¢ÎÞçȼæĨa&aïᢵåGÞï°V, ïßÜåæ;ïÞÃí					
19	$\emptyset \emptyset$ æÉX×X, Éløl, áÕljWÄjjBli µÞøcBZAí ÏÞæÄÞøjÎÞÈÆm B{í¢ÉÞÜlAáKl^.					
20	ÕM£cÞVj ßµ{áæ; ÖßF ÞÈÏ BZøåÉàµøßAáKÄí ¥ÇcÞɵøÞÃí					
21	μáG8μ{áæ; ØÎdωĨÞΪ ĨáÜc¢ Ě8VÃÏJ 8W ²ÞçøÞ ¥ÇcÞɵVAᢠ¥VÙĨÞΪ ÉCí¥ÈáÕÆ8, á ÈWµá Kál í					
22	øf μVÄā ØÙμøÃ ÈΪ ΒΖ ÄàøáÎ ÞÈßA áK Äí ¥Ç¢ÞÉμøÞà í					
23	§ΧæØVÕàØíælįΪβÈ۩ơí¥ÇαϷÉμçøÞ¿íçÕΙ dÄ μå¿βΪÞçÜÞ°ÈçΪÞæ¿Ϊ ^È¿J áΚÄί					
24	µø&AáܢȮVNÞÃJ®W¥icÞɵVAíÉCíµáù ÕÞÃí					
25	μ øßA áÜJ ßW dÉÞçÆÖßµ ÕcÄßøßµíÄÄA ÈáØøß, á U æÄøæE ¿áMáµZAí ¥ÕØøÎß^.					
26	¥ÇcÞɵøÞÃíÉÞçÀcÄødÉÕVJÈBZÎáçKÞGá æÕAáKÄi					
27	$ ilde{O}$ MECPVj Bµ{áæ; ÉáçøÞorÄBÕBÜÏBøáJÞÈáU dÉÞçÏÞoßµÎÞVPBZÄàøáÎÞÈBAáKÄí¥iCÞÉ µøÞÃí					
28	صå{&a ĺì Ä&µ ØÞÙ°øc Ø¢Õ&ÇÞÈJ &W (Physical Environment) ¥ÇcÞɵVAíÏ ÞæÄÞøá ÉCáĨ§^					
29	µ₱Øíùå¢ dÉÕVJÈBZÈ¿M8ÜÞA áKÄ&W¥ÇcÞ ɵVA í Éø8Ĵ ßÄ&µ{álí					
30	μΦØí ùᢠdÉÕVJ ÈøàÄß ÄàøáÎ ÞÈßA áK Äí ¥ Ç¢ÞÉ μøÞÃ í					
31	ÈMÌ t BÄØBÜÌ ØBÈſÉáùæÎ°BÜdÉçÄcµÕB× ÏBZAſµÞØſÈWµÞX¥ÇcÞɵVAſØbÞÄdLc Ĵál í					

Sl. No.	d ÉØÄÞÕÈ	EåVHIPI át çi þýððáKá	çİ Þ¥BA áK á	¥10164 18^	ÕIÇÏ Þ¼BAáKá	ÉåVHĨÞĨ át Ößj þýßAáKá
32	dÉßXØßMÜßæa ØNÄÎß^ÞæÄ Õ&cÞVj ßµZAí ØNÞÈ¢ÈWµÞX ¥ÇcÞɵVAíØbÞÄdLcĨál í					
33	ØiµâZØÞOJ ßµÈÏ¢øâÉæM;áJ áKÄ®W¥- Ç¢ÞɵVA1ÉCáI í					
34	ælÉÞË×ÃWÕßµØÈJßÈíçÕIßİáUæÉÞÄá ÎÞÈÆm BZćÞùÞAáKÄí¥Ç¢ÞɵøÞÃí					
35	ÉåVà ΠÞ'Í á¢ ÉáÄßI µøßA áÜ¢ ÕßµØßMßA ÞX ¥ÇcÞɵVA í ¥ÕØøĨ ál í					
36	ŐŀÆcÞVj BµZAÍÉáÄßÏ ÉÀçÈÞɵØÃ BZÉØB° ÏæM;áJ ÞX ¥ÇcÞɵVAÍ¥ÈáÕÞÆÍáI í					
37	ŐM£cÞVj &µZAí¦ ÕÖcÎÞÏ µcÞOáµZØ¢, الله M&AÞX ¥ÇcÞɵVAíØbÞÄ&LcÍál í					
38	¥ÇcÞɵçÖ×IŐVÇIMBA ÞÈÞŐBÖcÎ ÞÏ ædÉÞ¼- µí;áµZæ°‡ÞX ¥ÇcÞɵVA í ØbÞÄdL cí ál í					
39	ÎâÜcÈMHÏJ MÕcÄcØÅÎÞÏ øàÄ&µZ®ÉçÏÞ ω&AÞX¥ÇcÞɵVAíØbÞÄ&LcĨáI í					
40	ÉøBÙÞøçÌÞÇÈÐZ¦ŐbÖcÞÈáØøÃ¢È¿JÞ ÈáUØbÞÄ&Lc¢¥ÇcÞɵVAálí					
41	¥ÇcÞɵøáæ; çÈGBæ{ÎÞçȼæÎaá¢ØkvÞËá¢ çÕłÕkÇJMV¥ÈáçĨÞÆBA Þùálí					
42	Øiµi{læaÈki ΢,Èki dLâÄi¿Bki µÞøcB{k æÜøf lkÄÞA{iæ; S;æÉ;ÜiµZdÉÖÈÌÞÕÞùil í					
43	µáGkµ{áæ; ÕkµØÈJ kÈMáù¢øf kÄÞA{áæ; Ø¢ ÄðÉÄKI ÞÃíÍÞçȼæíakæaÜf cæíKíçÄÞKÞ ùál í					

Appendix VI

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TEACHER AUTONOMY SCALE

ENGLISH (FINAL)

Dr. Hassan Koya M.P	Shafeek P
Associate Professor	Senior Research Fellow

Personal Information

Name of the Teacher	:
(Optional)	
Name of the School	:
District (School)	:
Type of School	: CBSE/Kerala State Syllabus
Level of School	: Primary/Secondary/Higher Secondary
Sex	: Male/Female
Teaching Experience	: Below 5 years/ Below 10 years/ Above 20 years
Locale	: Rural/Urban
Subject	: Language/ Social Science/ Science/ Mathematics/ IT
Educational Qualification	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET
Computer Courses	: PGDCA/DCA/Other/ No Courses

Instructions

Five responses are given to each of the following Statements. 1. Strongly Agree, 2. Agree, 3. No Opinion, 4. Disagree and 5. Strongly Disagree.

Place a \checkmark in the appropriate column to indicate how true each statement is for you.

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
1	There is freedom and encouragement to adopt innovative and creative teaching methods.					
2	Excessive parental involvement creates difficulties.					
3	I think opinions in staff meetings receive due consideration.					
4	I can be a part of community service activities as much as.					
5	The teacher is not free to take any action as part of disciplining students.					
6	It is possible to conduct learning activities in and out of the classroom.					
7	The constant observation of the headmaster affects the naturalness of my class.					
8	Student field trips and study tours are conducted under the complete control of the teachers.					
9	There are many opportunities for teachers to get new qualifications and get promotion.					
10	There are unhealthy restrictions on the subject of teachers' leave.					
11	The pedagogy of the school is decided by the teachers.					
12	Teachers also play a role in determining the admission criteria of students.					
13	Teachers decide the seating arrangement of students in the classroom.					
14	Teachers form rules and regulations of the school.					
15	Teachers decide the curriculum goals and priorities of the school					
16	Teachers take the initiative to liaise with various external policy-making agencies related to education.					

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
17	Teachers frame rules and rules of conduct for students.					
18	Teachers' behavioral policies are decided through a joint discussion between teachers and management.					
19	There are no criteria for suspension or dismissal of teachers.					
20	Teachers formulate punishment policies for students.					
21	Each teacher has great role in determining the comprehensive evaluation of the student.					
22	Teachers decide parents - Co-operation Policies.					
23	In-service training is conducted without adequate discussion with the teachers.					
24	The role of teachers in curriculum development is less.					
25	There is no provision for local discrimination in the curriculum.					
26	Teachers suggest and guide extracurricular activities.					
27	Teachers decide practical ways to assess students' progress.					
28	Teachers have no role to play in the physical environment of the school.					
29	Teachers have limitations in carrying out classroom activities.					
30	Classroom functioning is decided by the teachers.					
31	Teachers are free to give classes on certain subjects in addition to the compulsory syllabus.					
32	Teachers are free to give gifts to students without the consent of the principal.					

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
33	Teachers have a role to play in shaping school economic policy.					
34	Teachers set general criteria for professional development.					
35	Teachers have the opportunity to develop a completely new curriculum.					
36	Teachers are allowed to introduce new learning materials to students.					
37	Teachers are free to organize camps for students.					
38	Teachers are free to carry out projects required to enhance their teaching skills.					
39	Teachers are free to use different methods in evaluation.					
40	Teachers have the freedom to make solutions as they see fit.					
41	Teachers' achievements are adequately commended by management and staff.					
42	I think parental involvement in school rules and regulations is problematic.					
43	Parents' satisfaction seems to be the goal of management rather than children's development.					

Appendix VII

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TEACHER BURNOUT INVENTORY

MALAYALAM

Dr. Hassan Koya M.P Associate Professor

Shafeek P Senior Research Fellow

Personal Information						
¥Ç¢Þɵæa çÉøi	:					
(ÈßVÌ t Îß^)						
ÕMcÞÜÏJ NæaçÉøi	:					
148^ (ØíµâZ)	:					
صåZ §È¢	: -ØlÌ l ®ØlØl/ കേരള സ്റ്റേറ്റ് സിലബസ്					
صáZ ÄÜ¢	: æædÉĴùß/æØA1 ùß/ÙÏ VæØA1 ùß					
ീഡ%U	: ¦ Y/æÉY					
¥Ç¢ÞÉÈÉøl⁰Ї⊄	: 5 ÕV-×J 16W ÄÞæÝ/10 ÕV-×J 16W ÄÞæÝ/10 ÕV-×J 16W µå¿áÄW					
dÉçÆÖ¢	: dupfi (/Èuac					
Õß×ï ¢	: أÞ-× / ØÞĴ åÙc ÖÞØtÄč/ ØÏ XØt/oà ßÄ¢/ æ® الأؤ					
çΪϷϭϭÄμΖ	: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET					
µOcâGV çµÞÝíØáµZ	: PGDCA/DCA/Other/ No Courses					

ÈN¢ÆÖBZ

ÄÞæÝ æµÞ¿áJ BøBA áK ²ÞçøÞ dÉØÄÞÕÈA ᢠçÈæø 1. ÉåVHÌÞÏ á¢,çÏ Þ¼B A áK á 2.çÏ Þ¼BA áK á 3.¥ÍBÉÞÏ ÎB^ 4.ÕBçÏ Þ¼BA áK á 5.ÉâVHÎÞÏ á¢ ÕBçÏ Þ¼BA á Ká®KBæÈ¥F ÍdÉÄBµøÃBZæµÞ¿áJ BøBAáKá.

 2 pçøþ dÉØiÄþÕÈÏ ál ÈlBæ{ Ølì -t ß, ß¿çJ þ{l ®dÄÍþdÄl ÖøßÏ þæÃ -Kí Äðøál ÞÈBA áµ. ¦ dÉÄBµøÃ ¢ dÉØÄÞÕÈÏ áæ; ÈOùBÈí çÈæøÏ áU çµÞ{-J ßW (✓) ° ßÙIÈÎßGí çø¶æM¿áJ áµ.

Sl. No.	(ÉØÄÞÕÈ	EåVHIPI åt çi þyabaáKá	çÏ Þ¼ ßA áKá	¥{\\Éhǐ }\^	ÕIÇÏ Þ¼BAáKá	ÉåVHTÞÍ át Ölçi þýllA áKá
1	¥ÇcÞÉÈ© ²øá ç¼ÞÜߦÏߨbàµøß, ÄßW ¾ÞX ØçLÞ×ÕÞÈÞÃí					
2	\mathbb{B} æa ç¼ÞÜBA í ¥ Èảçỉ Þ¼cĨ Þỉ cÕÄÈ¢ ® ÈBA í Ü- \mathbb{I} BA åK I^					
3	®È&Aíç¼ÞÜÖİ áÎÞÏ ßÌ tæMGíÈ^ ÎÞÈØ&µØ- NVg¢©I ÞÕÞùáI í					
4	ç¼ÞÜߨNVg¢µÞøÃ¢®È&AíÖÞøàø&µ¥ÈÞçøÞ oc ¢©I ÞÕÞùál í					
5	®æa ØÙŧÉÕVJµVçtÉÞrÞÙßM&AáKÄ&çÈ AÞZÕßVÖ&AÞùÞÃíÉÄ®Őí					
6	¥ÇcÞÉÈĨ^ÞJ ĨæxÞøá ç¼ÞÜ߿ĸæE ÏßøáKáæÕKí¾ÞX¦çÜÞ°ßA ÞùáIí					
7	¥ÇcÞɵçøÞ¿áU ŐÆcÞVj ßµ{áæ; Ì ÙáÌÞÈ J ßV µáùÕíÕøáKáæI Kí®È&AíçÄÞKÞùáI í					
8	2 ÞçøÞ ÆÕØÕát µÝŔľáçL Þùát ¥ÇcÞÉÈçJ Þjá U ®æa ÄÞÜÆ áct µáùE áÕááKá.					
9	¾ÞX ©J øÕÞÆÐ Ð{®V ÈßK ᢠ²ÝÆ áÎ ÞùÞX S·×íjæM;jáKá.					
10	µå¿áÄW ØĨĬ¢₃µÞŁÈÞĬß§øßAÞÈÞÃí¾ÞX Ś×jċæM;áKÄi					
11	$ \begin{array}{llllllllllllllllllllllllllllllllllll$					
12	ØÙdÉÕVJ μçøÞzíç¼ÞÜߨ¢Ì tÎÞÏ °V, μΖ ÈzJ áKÄí²ÝßÕÞAÞX ¾ÞX §-×ízæMzáKá.					
13	¥Ç¢ÞÉÈçÏÞ00;ÄÉøðFµZ¥Ç¢ÞÉÈÕðJ &Aí ©°BÄÎÞæÃKí®È&AíçÄÞKÞùálí					
14	¾ÞX ÉÀBMBA áK ÄíµáGBµZA ÍÎÈT BÜÞŐáK B^ ®Kí®ÈBA íçÄÞK Þùál í					
15	®ÈBA (È^ øàÄÄ KW ÉÀBMBA ÞX ØÞÇBA áK B^ ®K í ®ÈBA íçÄÞK Þùál í					
16	Î ÞùlðøáK Éøàf ØlOÆÞÏ B {á¢ØlÜÌ Øá¢ µÞøÃ¢¾ÞX ¥ØbØlÅÈÞÕÞùál í					

SI. No.	(ÉØÄÞÕÈ	EåVHIPI åt çi þyððáKá	çÏ Þ¼ ßA áKá	¥{触的孔	ÕlçT Þ¼lA áKá	ÉåVHTÞÍ át Ölçi þýlbA áK á
17	¥Y ®Ï 1ÁÁ1¥Ç¢Þɵøáæ; 1ÉÖIÈBæ{µáùB, 1 ¦øá¢çÕI 1ÄçÌÞÇÕÞXĨÞø^®K1®È&A1çÄÞ KÞùál 1					
18	çî ÞÖÎ ÞÏ ßæÉøáÎ ÞùáK ŐMEcÞVj ßµæ{ ¾ÞX Î È- ÉáVŒ¢ Î ÞxBÈßV:J Þùál í					
19	SA ÞÜJ í ÕÆcÞVj βμ{ ίæ; dÉÖÈ B { fW μιἰ; ίÄ· ÜÞÏ l S; æÉ; ÞÄlkolA ÜÞà í È^Äí ® K í ® ĚlA í çÄÞK Þùál					
20	¥Y ®ÏÍÁÁͥǢÞɵVAͲøáÕKÇJ KÜáU æÄÞÝKW Øáøf KÄÄbÕá¢ÜÍKAáKI^®KÍ®ÈKAÍ çÄÞKÞùáI í					
21	ÉáÄßI ÄÜŦáùT &Ü µáGßµæ{ĨÈŦßÜÞAÞX®Èß AíµÝßI áKß^.					
22	¥ÇcÞɵŐM£cÞVjßÌt¢ŞçMÞÝᢩì-×Íł{ĺÞ ÏßÈBÜÈBWAáKáIí®Kí®ÈBAíçÄÞKáKá.					
23	ÕMECPÍCÞØJ Næa ÕÞà N 4 cÕĵøÃ¢µÞøÃ¢ \mathbb{R} -ÈßAí¥ÇcÞÉÈç 4 ÞÜNçÏÞ¿í¥Ø¢ÄãÉÄßçÄÞKáKá.					
24	² ÞçøÞ ¥ÇcÏ ÈÕV×¢ Äáz BáçOÞÝᢠØÍµáZ ÎÞçÈ- ¼æÎ aí ¥ÇcÞɵVA í ¥ĨßÄç¼ÞÜßÍÞø¢ ÈWµáK Äí ÕÜßİ dÉÖÈĨÞÏ ß®ÈßA íçÄÞK Þùál í					
25	¥ÇcÞÉÈÕãJ KÍ KeÜØNVgÐZ®æaµá;á¢Ì ¼àÕKÄæJ ØÞøĨÞÏßÌÞÇKA áK álí					
26	ÕMECÞVj NçµnWaµaÄÖMECÞÍCÞØÐØ10ÆÞÍJNW ¥ÇcÞÉÈ¢ÎÈ:ØNVg¢©Uç¼ÞÜNİÞÃí®ÈBAí çÄÞKÞùálí					
27	¥ÇcÞÉȼàÕBÄ¢®ÈBAídÉÄàfß, ÄBçÈAÞÞZ Ø¢ÄãÉÄBÈWµáKálí					
28	¥ÇcÞÉÈÕãJ ßİßçÜAíÕøÞX¦doùbAáKÕV- Aí¦ÕÖcÎÞÏ ÈNVçÆÖBZÈWµß¾ÞXçdÉÞrÞ- ÙBMBAÞùál í					

Appendix VIII

FAROOK TRAINING COLLEGE

Affiliated to University of Calicut

TEACHER BURNOUT INVENTORY

ENGLISH

Dr. Hassan Koya M.P	Shafeek P
Associate Professor	Senior Research Fellow

Personal Information

Name of the Teacher	:
(Optional)	
Name of the School	:
District (School)	:
Type of School	: CBSE/Kerala State Syllabus
Level of School	: Primary/Secondary/Higher Secondary
Sex	: Male/Female
Teaching Experience	: Below 5 years/ Below 10 years/ Above 20 years
Locale	: Rural/Urban
Subject	: Language/ Social Science/ Science/ Mathematics/ IT
Educational Qualification	s: D.Ed. /B.Ed./Degree/ M.A/MS.C/M.COM/M.ED/ Ph.D./NET
Computer Courses	: PGDCA/DCA/Other/ No Courses

Instructions

Five responses are given to each of the following Statements. 1. Strongly Agree, 2. Agree, 3. No Opinion, 4. Disagree and 5. Strongly Disagree.

Place a \checkmark in the appropriate column to indicate how true each statement is for you.

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
1	I am happy to have preferred teaching as a career.					
2	I didn't get any appropriate salary for my job					
3	I feel stressed related to my profession					
4	I felt some physical illness due to job-related matters.					
5	My colleagues constantly criticize me than encouraging					
6	I think I could have chosen any other job other than teaching					
7	I think there is a devaluation in the behavior of students toward teachers					
8	I lose my composure every day with teaching.					
9	I like to avoid responsibilities					
10	I like to be alone most of the time					
11	I couldn't address the students' problems					
12	I try to avoid work-related discussions with co- teachers					
13	I think Teacher Eligibility Tests are suitable for teaching.					
14	I can't understand the taste of new-generation students					
15	I think I couldn't teach well					
16	I feel irritated due to changing syllabus and examination system					
17	I think no one is bothered about the issues of unaided teachers					
18	I try to keep away from misbehaving students					

Sl. No.	Statements	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
19	I think teachers do not get involved in student issues					
20	I think unaided teachers do not get any job security					
21	I think that students didn't get what I have taught					
22	I think the teacher-student relationship is still warm					
23	I feel dissatisfied with the teaching profession due to the commercialization of education					
24	I feel overwork of teachers given by school management at the beginning of each academic year is a big problem					
25	Stress in the teaching profession significantly affects my family life					
26	I feel that teaching in student-cantered education is a stressful job					
27	Teaching life is more satisfying than I expected					
28	I always encourage those who want to enter the teaching profession by giving the necessary instruction					